#### STATE OF VERMONT CONTRACT AMENDMENT

It is hereby agreed by and between the State of Vermont, Department of Buildings and General Services (the "State") and LimeLite Restoration Services, Inc. with a principal place of business in Irasburg, VT (the "Contractor") that the contract between them originally dated as of March 1, 2019, Contract #37787, as amended to date, (the "Contract") is hereby amended as follows:

- I. <u>Maximum Amount</u>. The maximum amount payable under the Contract, wherever such reference appears in the Contract, shall be changed from \$300,000.00 to \$400,000.00, representing an increase of \$100,000.00.
- II. <u>Contract Term</u>. The Contract end date, wherever such reference appears in the Contract, shall be changed from February 28, 2022 to February 28, 2023.

<u>Taxes Due to the State</u>. Contractor certifies under the pains and penalties of perjury that, as of the date this contract amendment is signed, the Contractor is in good standing with respect to, or in full compliance with a plan to pay, any and all taxes due the State of Vermont.

<u>Child Support (Applicable to natural persons only; not applicable to corporations, partnerships or LLCs)</u>. Contractor is under no obligation to pay child support or is in good standing with respect to or in full compliance with a plan to pay any and all child support payable under a support order as of the date of this amendment.

<u>Certification Regarding Suspension or Debarment</u>. Contractor certifies under the pains and penalties of perjury that, as of the date this contract amendment is signed, neither Contractor nor Contractor's principals (officers, directors, owners, or partners) are presently debarred, suspended, proposed for debarment, declared ineligible or excluded from participation in federal programs, or programs supported in whole or in part by federal funds.

Contractor further certifies under pains and penalties of perjury that, as of the date this contract amendment is signed, Contractor is not presently debarred, suspended, nor named on the State's debarment list at: http://bgs.vermont.gov/purchasing-contracting/debarment

This document consists of 2 pages. Except as modified by this Amendment No. 4, all provisions of the Contract remain in full force and effect.

[Remainder of Page Intentionally Left Blank]

Revision Date: 08/05/2021

The signatures of the undersigned indicate that each has read and agrees to be bound by this Amendment to the Contract.

STATE OF VERMONT	LIMELITE RESTORATION SERVICES, INC	
By:	By:	
Name: Jennifer M.V. Fitch	Name:	
Title: BGS Commissioner	Title:	
Date:	Date:	

#### STATE OF VERMONT CONTRACT AMENDMENT

It is hereby agreed by and between the State of Vermont, Department of Buildings and General Services (the "State") and LimeLite Restoration Services, Inc. with a principal place of business in Irasburg, VT (the "Contractor") that the contract between them originally dated as of March 1, 2019, Contract #37787, as amended to date, (the "Contract") is hereby amended as follows:

- I. Contract Term. The Contract end date, wherever such reference appears in the Contract, shall be changed from February 28, 2021 to February 28, 2022.
- II. <u>Attachment D, General Provisions for Construction Contracts</u>. Attachment D is hereby deleted in its entirety and replaced by the Attachment D, May 22, 2019 attached to this Amendment.

<u>Taxes Due to the State</u>. Contractor certifies under the pains and penalties of perjury that, as of the date this contract amendment is signed, the Contractor is in good standing with respect to, or in full compliance with a plan to pay, any and all taxes due the State of Vermont.

Child Support (Applicable to natural persons only; not applicable to corporations, partnerships or <u>LLCs</u>). Contractor is under no obligation to pay child support or is in good standing with respect to or in full compliance with a plan to pay any and all child support payable under a support order as of the date of this amendment.

<u>Certification Regarding Suspension or Debarment</u>. Contractor certifies under the pains and penalties of perjury that, as of the date this contract amendment is signed, neither Contractor nor Contractor's principals (officers, directors, owners, or partners) are presently debarred, suspended, proposed for debarment, declared ineligible or excluded from participation in federal programs, or programs supported in whole or in part by federal funds.

Contractor further certifies under pains and penalties of perjury that, as of the date this contract amendment is signed, Contractor is not presently debarred, suspended, nor named on the State's debarment list at: http://bgs.vermont.gov/purchasing-contracting/debarment

This document consists of 2 pages. Except as modified by this Amendment No. 3, all provisions of the Contract remain in full force and effect.

[Remainder of Page Intentionally Left Blank]

The signatures of the undersigned indicate that each has read and agrees to be bound by this Amendment to the Contract.

STATE OF VERMONT	LIMELITE RESTORATION SERVICES, INC	
By:	Ву:	
Name:	Name:	
Title:	Title:	
Date:	Date:	

# STATE OF VERMONT STANDARD FORM

## GENERAL CONDITIONS FOR CONSTRUCTION CONTRACTS (May 22, 2019)

The following general conditions are for use with all construction contracts with the State of Vermont, administered through the Department of Buildings and General Services.

These general conditions will be attached to, and become part of, the Contract Documents.

If there is a conflict between the terms and conditions stated in these general conditions and the terms and conditions contained in the State-Contractor Agreement; the terms of the State-Contractor Agreement control.

#### **ARTICLE 1**

#### CONTRACT DOCUMENTS

#### 1.1 **DEFINITIONS**

The term "State-Contractor Agreement" refers to the specific agreement between the State of Vermont and a particular contractor whereby the State and the contractor identify, explain and agree to the specific contractual obligations of each party and identify the nature and scope of a particular project. The "State-Contractor Agreement" will utilize one of the following construction delivery methods: (i) A general contractor hired by the State to provide all services necessary for the construction of the particular project. The general contractor is responsible for completing the particular project for a fixed price; (ii) a construction manager hired by the State to provide professional services and act as a consultant to the State in the design development and construction phases and to deliver the particular project for a Guaranteed Maximum Price; or (iii) a design-build contractor hired by the State to provide both design and construction services, under a single contract, for a particular project.

#### 1.1.1 THE CONTRACT DOCUMENTS

The Contract Documents consist of the State-Contractor Agreement, the Conditions of the Contract (General, Supplementary and other Conditions), the Drawings, the Specifications, and all Addenda issued prior to bid opening and any Change Orders after execution of the Contract.-

#### 1.1.2 THE CONTRACT

The Contract Documents form the contract for construction. This contract represents the entire and integrated agreement between the parties and supersedes all prior negotiations, representations, or agreements, either written or oral. The contract may only be amended by Change Order. The Contract Documents shall not be construed to create any contractual relationship of any kind between the Architect and the Contractor, but the Architect shall be entitled to performance of obligations intended for his benefit, and to enforcement thereof. Nothing contained in the Contract Documents shall create any contractual relationship between the State or the Architect and any Subcontractor or Sub-subcontractor.

#### **1.1.3 THE WORK**

The Work comprises the completed construction required by the Contract Documents and includes all labor necessary to produce such construction, and all materials and equipment incorporated or to be incorporated in such construction.

#### 1.1.4 THE PROJECT

The Project is the total construction of which the Work performed under the Contract Documents may be the whole or a part.

#### 1.1.5 PROJECT MANAGER

The term "Project Manager" refers to the employee of the State who has been assigned responsibility for overseeing and managing the proper and timely implementation of the project.

#### 1.2 EXECUTION & INTENT

- **1.2.1** Upon request, the State-Contractor Agreement will be signed in duplicate by the State and Contractor.
- **1.2.2** By executing the Contract, the Contractor represents that he or she has visited and examined the site; is fully informed as to the extent and character of the Project; familiarized himself or herself with the local conditions under which the Work is to be performed, and correlated his / her observations with the requirements of the Contract Documents and understands that no consideration will be given for any alleged misunderstanding of same.
- **1.2.3** The intent of the Contract Documents is to include all items necessary for the proper execution and completion of the Work. The Contract Documents are complementary, and what is required by any one shall be as binding as, if required by all. Work not covered in the Contract Documents will not be required unless it is consistent with the Contract Documents and is reasonably inferable from the Contract Documents as being necessary to produce the intended results. Words and abbreviations which have well-known technical or trade meanings are used in the Contract Documents in accordance with such recognized meanings.
- 1.2.4 The organization of the Specifications into divisions, sections and articles, and the arrangement of Drawings shall not control the Contractor in dividing the Work among

Subcontractors or in establishing the extent of Work to be performed by any trade.

#### 1.3 OWNERSHIP AND USE OF DOCUMENTS

**1.3.1** All drawings, specifications, estimates, and all other documents, including shop drawings and calculations, prepared at any time in connection with the Project, shall, upon payment for services in connection therewith, become the sole property of the State.

#### **ARTICLE 2**

#### **ARCHITECT**

#### 2.1 **DEFINITION**

The term Architect, as used in this document, means a person, or entity, lawfully licensed to practice architecture or a person or entity lawfully licensed to provide professional engineering services and has been identified as such in the State-Contractor Agreement. The term Architect means the Architect or his / her authorized representative. These General Conditions are applicable whether or not an Architect is utilized for a particular project. In the event that an Architect has not been utilized for the project, the Project Manager will be responsible for administration of the contract as specifically enumerated in paragraph 2.2. All other functions, duties, or obligations of the Architect, under these General Conditions, in the absence of an Architect, will be performed by the Project Manager.

#### 2.2 ADMINISTRATION OF THE CONTRACT

- **2.2.1** The Architect will provide administration of the Contract as hereinafter described.
- **2.2.2** The Architect will be the State's representative during construction and until final payment is due. The Architect will advise and consult with the State. The State's instructions to the Contractor shall be forwarded through the Architect. The Architect will have authority to act on behalf of the State only to the extent provided in the Contract Documents, unless otherwise modified by written instrument in accordance with Subparagraph 2.2.10.
- **2.2.3** The Architect will visit the site at intervals appropriate to the stage of construction to familiarize himself or herself generally with the progress and quality of the Work and to determine in general if the Work is proceeding in accordance with the Contract Documents. However, the Architect will not be required to make exhaustive or continuous on-site inspections to check the quality or quantity of the Work. On the basis of his or her on-site observations as an architect, he or she will keep the State informed of the progress of the Work, and will endeavor to guard the State against defects and deficiencies in the Work of the Contractor.
- **2.2.4** The Architect will not be responsible for and will not have control or charge of construction means, methods, techniques, sequences or procedures, or for safety precautions and programs in connection with the Work, and he or she will not be responsible for the Contractor's failure to carry out the Work in accordance with the Contract Documents. The Architect will not be responsible for or have control or charge over the acts or omissions of the Contractor,

Subcontractors, or any of their agents or employees, or any other persons performing any of the Work.

- **2.2.5** The Architect shall at all times have access to the Work wherever it is in preparation and progress. The Contractor shall provide facilities for such access so the Architect may perform his or her functions under the Contract Documents.
- **2.2.6** Based on the Architect's observations of the Work and an evaluation of the Contractor's Applications for Payment, the Architect will determine the amounts owing to the Contractor and will issue Certificates for Payment in such amounts, as provided in Paragraph 9.4 of these General Conditions.
- **2.2.7** The Architect will review and approve or take other appropriate action upon Contractor's submittals such as Shop Drawings, Product Data and Samples, but only for conformance with the design concept of the Work and with the information given in the Contract Documents. Such action shall be taken with reasonable promptness so as to cause no delay.
- **2.2.8** The Architect will prepare Change Orders in accordance with Article 12. The Architect has authority to order minor changes in the Work with the consent of the owner.
- **2.2.9** The Architect will conduct inspections to determine the dates of Substantial Completion and final completion, will receive and forward to the State for the State's review written warranties and related documents required by the Contract and assembled by the Contractor, and will issue a final Certificate for Payment upon compliance with the requirements of Paragraph 9.9.
- **2.2.10** The duties, responsibilities and limitations of authority of the Architect as the State's representative during construction as set forth in the Contract Documents will not be modified or extended without written consent of the State and the Architect. In the event that the duties, responsibilities and limitations of authority of the Architect as State's representative during construction as set forth in the Contract Documents is modified or extended, the State will immediately notify the Contractor, in writing, of the modification or extension and the date that the modified or extended authority is effective. If the change of duties, responsibilities, or limitations of authority of the Architect, causes the Contractor to incur new, additional, and unexpected expenses completing the Work, then the State and the Contractor shall agree to an equitable adjustment to be implemented by Change Order to the State-Contractor Agreement.
- **2.2.11** In case of the termination of the employment of Architect, the State may appoint a subsequent architect in which case, the subsequent architect will have the same status under the Contract Documents as that of the former architect. In the event of termination of the Architect, the State will immediately notify the Contractor, in writing, of the termination and the appointment of a replacement architect, if any.
- 2.2.12 The Architect will be the interpreter of the requirements of the Contract Documents and will make recommendations to the State regarding the progress and quality of the Work and whether the Work is proceeding in accordance with the Contract Documents.
- 2.2.13 The Architect will render interpretations necessary for the proper execution and progress of the Work, with reasonable promptness and in accordance with any time limit agreed

upon, in writing, by the Architect, Contractor, and the State.

2.2.14 Claims, disputes and other matters in question between the Contractor and the State relating to the execution or progress of the Work or the interpretation of the Contract Documents shall be referred initially to the Architect for review. The Architect will review the nature of the claim, dispute or other matter, along with any documentation provided by the Contractor or the State, and within a reasonable time, the Architect will provide a written recommendation to the State.

#### **ARTICLE 3**

#### **STATE**

#### 3.1 **DEFINITION**

The State of Vermont, by and through the Department of Buildings and General Services is the entity identified as "State" in the State-Contractor Agreement and is referred to throughout the Contract Documents as the "State".

#### 3.2 RESPONSIBILITIES OF THE STATE

- 3.2.1 The State will identify a Project Manager for the Project. The Project Manager will be the sole point of contact between the Contractor and the State. The Project Manager will be the State's representative authorized to communicate the State's position and directions related to all contract work and to coordinate all change orders with the Commissioner of Buildings and General Services as deemed necessary.
- **3.2.2** If applicable, the State shall furnish all surveys describing the physical characteristics, legal limitations and utility locations for the site of the Project, and a legal description of the site.
- **3.2.3** Except as provided in Subparagraph 4.7.1, the State shall secure and pay for necessary approvals, easements, assessments and charges required for the construction, use or occupancy of permanent structures or for permanent changes in existing facilities.
- **3.2.4** Information or services under the State's control shall be furnished by the State with reasonable promptness to avoid delay in the orderly progress of the Work.
- **3.2.5** Unless otherwise provided in the Contract Documents, the Contractor will be furnished, free of charge, all copies of Drawings and Specifications reasonably necessary for the execution of the Work.
- **3.2.6** The State shall forward instructions to the Contractor through the Architect.
- **3.2.7** The foregoing are in addition to any other duties and responsibilities of the State enumerated in the Contract Documents including those responsibilities described in Articles 6, 9 and 11 of this agreement.

**3.2.8** The State may utilize a Clerk-of-the-Works to provide inspection and monitoring work on behalf of the State. If a Clerk-of-the-Works is utilized, it is the responsibility of the Architect and the Contractor to review and understand the Clerk's duties, responsibilities and limitations of authority. Upon request, the State shall provide a copy of the Clerk-of-the-Works contract to Architect and Contractor.

#### 3.3 STATE'S RIGHT TO STOP THE WORK

**3.3.1** If the Contractor fails to correct defective Work as required by Paragraph 13.2 or persistently fails to carry out the Work in accordance with the Contract Documents, the State, by a written order signed personally or by an agent specifically so empowered by the State in writing, may order the Contractor to stop the Work, or any portion thereof, until the cause for such order has been eliminated; however, this right of the State to the stop the Work shall not give rise to any duty on the part the State to exercise this right for the benefit of the Contractor or any other person or entity, except to the extent required by Subparagraph 6.1.3.

#### 3.4 STATE'S RIGHT TO CARRY OUT THE WORK

**3.4.1** If the Contractor defaults or neglects to carry out the Work in accordance with the Contract Documents and fails within seven days after receipt of written notice from the State to commence and continue correction of such default or neglect with diligence and promptness, the State may, after seven days following receipt by the Contractor of an additional written notice and without prejudice to any other remedy he may have, make good such deficiencies. In such case an appropriate Change Order shall be issued deducting from the payments then or thereafter due the Contractor the cost of correcting such deficiencies, including compensation for the Architect's and the State's additional services made necessary by such default, neglect or failure. Such action by the State and the amount charged to the Contractor are both subject to review by the Architect under Paragraph 2.2.14 of these General Conditions. If the payments then or thereafter due the Contractor are not sufficient to cover the amounts paid by the State to make good such deficiencies, then the Contractor shall pay the difference to the Owner.

#### **ARTICLE 4**

#### **CONTRACTOR**

#### 4.1 **DEFINITION**

The Contractor is the person or entity identified as such in the State Contractor Agreement and is referred to throughout the Contract Documents as Contractor. The term Contractor means the Contractor or his authorized representative.

#### 4.2 REVIEW OF CONTRACT DOCUMENTS

**4.2.1** The Contractor shall carefully study and compare the Contract Documents and shall immediately report in writing to the Architect and the State any error, inconsistency or omission he may discover. The Contractor shall not be liable to the State or the Architect for any damage resulting from any such errors, inconsistencies or omissions in the Contract Documents. The

Contractor shall perform no portion of the Work at any time without Contract Documents or, where required, approved Shop Drawings, Product Data or Samples for such portion of the Work.

#### 4.3 SUPERVISION AND CONSTRUCTION PROCEDURES

- **4.3.1** The Contractor shall supervise and direct the Work, using his / her best skill and attention. The Contractor shall be solely responsible for all construction means, methods, techniques, sequences and procedures and for coordinating all portions of the Work under the Contract. All aspects of the Project shall be subject to the inspection and approval of the State. Contractor guarantees to repair, replace, re-execute or otherwise correct any defect in workmanship, materials, or the like that fails to conform to the requirements of this Contract or that appears during the progress of the Work or within one year of final acceptance by the State.
- **4.3.2** The Contractor shall be responsible to the State for the acts and omissions of his / her employees, Subcontractors and their agents and employees, and other persons performing any of the Work under a contract with the Contractor.
- **4.3.3** The Contractor shall not be relieved from his / her obligations to perform the Work in accordance with the Contract Documents either by the activities or duties of the Architect in his / her administration of the Contract, or by inspections, tests or approvals required or performed under Paragraph 7.7 by persons other than the Contractor.
- 4.3.4 The Contractor shall acquaint himself (herself/itself) with the limits of the property or right-of-way of the State and shall not trespass on other property. The Contractor shall adequately protect the project, adjacent property and the public, and shall be responsible for any damage or injury due to the Contractor's act or neglect, and shall save the State harmless in respect thereto.
- 4.3.5 All work shall be done in such a manner as not to interfere with the State's operating functions. Contractor and his employees shall familiarize themselves and comply with all rules and regulations applicable to the project.
- 4.3.6 The Contractor shall keep the premises free from liens arising out of or from the Project. Contractor shall obtain and submit waivers of liens with a request for a progress or final payment.

#### 4.4 LABOR AND MATERIALS

**4.4.1** Unless otherwise provided in the Contract Documents, the Contractor shall provide and pay for all labor, materials, equipment, tools, construction equipment and machinery, water, heat, utilities, transportation, and other facilities and services necessary for the proper execution and completion of the Work, whether temporary or permanent and whether or not incorporated or to be incorporated in the Work. Contractor guarantees that materials shall be of the best quality, that work shall be completed in a neat and workmanlike manner, that equipment will be installed in a first class manner, and that all aspects of the project will be delivered in good working order complete and perfect in every respect and that all systems and materials necessary to make the project a complete operating utility as contemplated by the above description of the project is included in the Contract price.

**4.4.2** The Contractor shall a t all times enforce strict discipline and good order among his employees and shall not employ on the Work any unfit person or anyone not skilled in the task assigned to him.

#### 4.5 WARRANTY

**4.5.1** The Contractor warrants to the State and the Architect that all materials and equipment furnished under this Contract will be new unless otherwise specified, and that all Work will be of good quality, free from faults and defects and in conformance with the Contract Documents. All Work not conforming to these requirements, including substitutions not properly approved and authorized, may be considered defective. If requested by the Architect or the State, the Contractor shall furnish satisfactory evidence as to the kind and quality of materials and equipment. This warranty is not limited by the provisions of Paragraph 13.2 of these General Conditions.

#### 4.6 TAXES

The State is exempt from all sales and federal excise taxes. Contractors will be responsible for the payment of any sales, consumer, use and other similar taxes for the Work or portions thereof provided by the Contractor which are legally enacted at the time bids are received, whether or not yet effective.

#### 4.7 PERMITS, FEES AND NOTICES

- **4.7.1** Unless otherwise provided in the Contract Documents, the Contractor shall secure and pay for all permits and governmental fees, licenses and inspections necessary for the proper execution and completion of the Work which are customarily secured after execution of the Contract and which are legally required at the time the bids are received. Contractor shall comply with the regulations and requirements of any insurance company which issues a policy on any part of the work or site.
- **4.7.2** The Contractor shall give all notices and comply with all laws, ordinances, rules, regulations and lawful orders of any public authority bearing on the performance of the Work.
- **4.7.3** It is not the responsibility of the Contractor to make certain that the Contract Documents are in accordance with applicable laws, statutes, building codes and regulations. However, if the Contractor observes that any of the Contract Documents are at variance therewith in any respect, he / she shall promptly notify the Architect in writing, and any necessary changes shall be accomplished by appropriate modification.
- **4.7.4** If the Contractor performs any Work knowing it to be contrary to such laws, ordinances, rules and regulations, and without such notice to the Architect, he / she shall assume full responsibility therefor and shall bear all costs attributable thereto.
- 4.7.5 The Contractor agrees to comply with all the requirements of Title 21 V.S.A., Chapter 5, Subchapter 6 relating to fair employment practices and agrees further to include a similar provision in any and all subcontracts. A link to 21 V.S.A. Chapter 5, Subchapter 6 is provided: http://legislature.vermont.gov/statutes/section/21/005/00495

#### 4.8 ALLOWANCES

**4.8.1** The Contractor shall include in the Contract Sum all allowances stated in the Contract Documents. Items covered by these allowances shall be supplied for such amounts and by such persons as dictated by the process contained in the Contract Documents.

#### **4.8.2** Unless otherwise provided in the Contract Documents:

- (a) these allowances shall cover the cost to the Contractor, less any applicable trade discount, of the materials and equipment required by the allowance delivered at the site, and all applicable taxes;
- (b) the Contractor's costs for unloading an d handling on the site, labor, installation costs, overhead, prof it and other expenses contemplated for the original allowance shall be included in the Contract Sum and not in the allowance:
- (c) whenever the cost is more than or less than the allowance, the Contract Sum shall be adjusted accordingly by Change Or der, the a mount of which will recognize changes, if any, in handling costs on site, labor, installation costs, overhead, profit and other expenses.

#### 4.9 SUPERINTENDENT

The Contractor shall employ a competent superintendent and necessary assistants who shall be in attendance at the Project site during the progress of the Work. The superintendent shall represent the Contractor and all communications given to the superintendent shall be as binding as if given to the Contractor. All Communications shall be in writing when made or shall be confirmed in writing, by the Contractor, within twenty-four (24) hours of the communication.

#### 4.10 PROGRESS SCHEDULE

The Contractor, within ten (10) working days of receiving notice of the award of the contract, shall prepare and submit for the State's and Architect's information an estimated progress schedule for the Work. The progress schedule shall be related to the entire Project to the extent required by the Contract Documents, and shall provide for expeditious and practicable execution of the Work.

#### 4.11 DOCUMENTS AND SAMPLES AT THE SITE

**4.11.1** The Contractor shall maintain at the site for the State one record copy of all Drawings, Specifications, Addenda, Change Orders and other modifications, in good order and marked currently to record all changes made during construction, and approved Shop Drawings, Product Data and Samples. These shall be available to the Architect and the State for monthly review prior to approval of Contractor's monthly application for payment. Prior to substantial completion of the Work, the Contractor shall ensure that one record copy of all Drawings, Specifications, Addenda, Change Orders and other Modifications, in good order and marked currently to record all changes made during construction, and approved Shop Drawings, Product Data and Samples are delivered to the Architect for the State and a duplicate copy shall be delivered to the State for the State's use.

#### 4.12 SHOP DRAWINGS, PRODUCT DATA AND SAMPLES

- **4.12.1** Shop Drawings are drawings, diagrams, schedules and other data specially prepared for the Work by the Contractor or any Subcontractor, manufacturer, supplier or distributor to illustrate some portion of the Work.
- **4.12.2** Product Data are illustrations, standard schedules, performance charts, instructions, brochures, diagrams and other information furnished by the Contractor to illustrate a material, product or system for some portion of the Work.
- **4.12.3** Samples are physical examples which illustrate materials, equipment or workmanship and establish standards by which the Work will be judged.
- **4.12.4** The Contractor shall review, approve and submit, with reasonable promptness and in such sequence as to cause no delay in the Work or in the work of the State or any separate contractor, all Shop Drawings, Product Data and Samples required by the Contract Documents.
- **4.12.5** By approving and submitting Shop Drawings, Product Data and Samples, the Contractor represents that he / she has determined and verified all materials, field measurements, and field construction criteria related thereto, or will do so, and that he / she has checked and coordinated the information contained within such submittals with the requirements of the Work and of the Contract Documents.
- **4.12.6** The Contractor shall not be relieved of responsibility for any deviation from the requirements of the Contract Documents by the Architect's approval of Shop Drawings, Product Data or Samples under Subparagraph 2.2.7 of these General Conditions unless the Contractor has specifically informed the Architect and the State in writing of such deviation at the time of submission and the Architect and the State has given written approval to the specific deviation. The Contractor shall not be relieved from responsibility for errors or omissions in the Shop Drawings, Product Data or Samples by the Architect's approval thereof.
- **4.12.7** The Contractor shall direct specific attention, in writing or on resubmitted Shop Drawings, Product Data or Samples, to revisions other than those requested by the Architect on previous submittals.
- **4.12.8** No portion of the Work requiring submission of a Shop Drawing, Product Data or Sample shall be commenced until the submittal has been approved by the Architect as provided in Subparagraph 2.2.7 of these General Conditions. All such portions of the Work shall be in accordance with approved submittals.

#### 4.13 USE OF SITE

**4.13.1** The Contractor shall confine operations at the site to areas permitted by law, including areas designated for operations or prohibited from operations by any applicable permit and or regulation, ordinances, permits and the Contract Documents and shall not unreasonably encumber the site with any materials or equipment.

#### 4.14 CUTTING AND PATCHING OF WORK

- **4.14.1** The Contractor shall be responsible for all cutting, fitting or patching that may be required to complete the Work or to make its several parts fit together properly.
- **4.14.2** The Contractor shall not damage or endanger any portion of the Work or the work of the State or any separate contractors by cutting, patching or otherwise altering any work, or by excavation. The Contractor shall not cut or otherwise alter the work of the State or any separate contractor except with the written consent of the State and of such separate contractor. The Contractor shall not unreasonably withhold from the State or any separate contractor his consent to cutting or otherwise altering the Work.

#### 4.15 CLEANING UP

- **4.15.1** Before commencement of any on-site activities, Contractor is required to prepare and submit to the Project Manager a Construction Site Waste Reduction Plan. A sample plan is available at: <a href="http://www.anr.state.vt.us/dec/wastediv/recycling/CandD.htm">http://www.anr.state.vt.us/dec/wastediv/recycling/CandD.htm</a>. Failure to comply with this provision or a failure to comply with the plan itself will result in withholding of general conditions' money from the contractor's monthly requisition until Contractor has rectified the situation and is in full compliance with these provisions.
- **4.15.2** The Contractor at all times shall keep the premises free from accumulation of waste materials or rubbish. At the completion of the Work he shall remove all his waste materials and rubbish from and about the Project as well as all his tools, construction equipment, machinery and surplus materials and shall leave the premises in a clean and satisfactory condition.
- **4.15.3** If the Contractor fails to clean up at the completion of the Work, the State may do so as provided in Paragraph 3.4 and the cost thereof shall be charged to the Contractor.

#### 4.16 COMMUNICATIONS

**4.16.1** The Contractor shall forward all communications to the State through the Architect.

#### 4.17 ROYALTIES AND PATENTS

The Contractor shall pay all royalties and license fees. The Contractor shall defend all suits or claims for infringement of any patent rights and shall save the State harmless from loss on account thereof, except that the State shall be responsible for all such loss when a particular design, process or the product of a particular manufacturer or manufacturers is specified, but if the Contractor has reason to believe that the design, process or product specified is an infringement of a patent, The Contractor shall be responsible for such loss unless he promptly gives such information to the Architect.

#### 4.18 INDEMNIFICATION

The Contractor hereby agrees and consents to indemnify and hold harmless the State under the terms and conditions specified in "Attachment C".

#### **SUBCONTRACTORS**

#### 5.1 **DEFINITION**

- **5.1.1** A Subcontractor is a person or entity who has a direct contract with the Contractor to perform any of the Work at the site. The term Subcontractor means a Subcontractor or his / her authorized representative. The term Subcontractor does not include any separate contractor or his / her subcontractors.
- **5.1.2** A Sub-subcontractor is a person or entity who has a direct or indirect contract with a Subcontractor to perform any of the Work at the site. The term Sub-subcontractor means a Sub-subcontractor or an authorized representative thereof.

### 5.2 AWARD OF SUBCONTRACTS AND OTHER CONTRACTS FOR PORTIONS OFTHE WORK

Unless other procedures are specified or required by the Contract Documents or the Bidding Documents, then the following provisions are applicable:

- **5.2.1** The Parties agree that it is in the best interests of both the Contractor and the State to determine and select subcontractors as soon as possible after the Contractor has been awarded the Project. Therefore, the Contractor agrees, to furnish to the State and the Architect in writing the names of the persons or entities (including those who are to furnish materials or equipment fabricated to a special design) proposed for each of the principal portions of the Work within Ten (10) working days of award of the Project to Contractor. The State will reply within Ten (10) working days to the Contractor in writing stating whether or not the State, after due investigation, has reasonable objection to any such proposed person or entity.
- **5.2.2** The Contractor shall not contract with any such proposed person or entity to whom the State has made reasonable objection under the provisions of Subparagraph 5.2.1 of these General Conditions. The Contractor shall not be required to contract with anyone to whom he has a reasonable objection.
- **5.2.3** If the State has reasonable objection to any such proposed person or entity, the Contractor shall submit a substitute to whom the State has no reasonable objection, and the Contract Sum shall be increased or decreased by the difference in cost occasioned by such substitution and an appropriate Change Order shall be issued; however, no increase in the Contract Sum shall be allowed for any such substitution unless the Contractor has acted promptly and responsively in submitting names as required by Subparagraph 5.2.1. of these General Conditions.
- **5.2.4** The Contractor shall make no substitution for any Subcontractor, person or entity previously selected if the State makes a reasonable objection to such substitution.

#### 5.3 SUBCONTRACTUAL RELATIONS

**5.3.1** By written agreement, the Contractor shall require each Subcontractor, to the extent of the Work to be performed by the Subcontractor, to be bound to the Contractor by the terms of the Contract Documents, and to assume toward the Contractor all the obligations and responsibilities

which the Contractor, by these Documents, assumes toward the State and the Architect. Said agreement shall preserve and protect the rights of the State and the Architect under the Contract Documents with respect to the Work to be performed by the Subcontractor so that the subcontracting thereof will not prejudice such rights, and shall allow to the Subcontractor, unless specifically provided otherwise in the Contractor Subcontractor agreement, the benefit of all rights, remedies and redress against the Contractor that the Contractor, by these Documents, has against the State. The Contractor shall require each Subcontractor to enter into similar agreements with his Sub-subcontractors. The Contractor shall make available to each proposed Subcontractor, prior to the execution of the Subcontract, copies of the Contract Documents to which the Subcontractor will be bound by this Paragraph 5.3, and identify to the Subcontractor any terms and conditions of the proposed Subcontract which may be at variance with the Contract Documents. Each Subcontractor shall similarly make copies of such Documents available to his / her Sub-subcontractors.

#### **ARTICLE 6**

#### WORK BY STATE OR BY SEPARATE CONTRACTORS

## 6.1 STATE'S RIGHT TO PERFORM WORK AND TO AWARD SEPARATE CONTRACTS

- **6.1.1** The State reserves the right to perform work related to the Project with its own forces, and to award separate contracts in connection with other portions of the Project or other work on the site under these or similar Conditions of the Contract. If the Contractor claims that delay or additional cost is involved because of such action by the State, he / she shall make such claim as provided elsewhere in the Contract Documents.
- **6.1.2** When separate Contracts are awarded for different portions of the Project or other work on the site, the term Contractor in the Contract Documents in each case shall mean the Contractor who executes each separate State Contractor Agreement.
- **6.1.3** The State will provide for the coordination of the work of its own forces and of each separate contractor with the Work of the Contractor, who shall cooperate therewith as provided in Paragraph 6.2.

#### **6.2 MUTUAL RESPONSIBILITY**

- **6.2.1** The Contractor shall afford the State and separate contractors reasonable opportunity for the introduction and storage of their materials and equipment and the execution of their work, and shall connect and coordinate his / her Work with the work of the State and separate contractors as required by the Contract Documents.
- **6.2.2** If any part of the Contractor's Work depends for proper execution or results upon the work of the State or any separate contractor, the Contractor shall, prior to proceeding with the Work, promptly report to the Architect any apparent discrepancies or defects in such other work that render it unsuitable for such proper execution and results. Failure of the Contractor so to report shall constitute an acceptance of the State's or separate contractors work as fit and proper to receive

his Work, except as to defects which may subsequently become apparent in such work by others.

- **6.2.3** Any costs caused by defective or ill-timed work shall be borne by the party responsible therefor.
- **6.2.4** Should the Contractor wrongfully cause damage to the work or property of the State, or to other work on the site, the Contractor shall promptly remedy such damage as provided in Subparagraph 10.2.5 of these General Conditions.

#### 6.3 STATE'S RIGHT TO CLEAN UP

**6.3.1** If a dispute arises between the Contractor and separate contractors as to their responsibility for cleaning up as required by Paragraph 4.15 of these General Conditions, the State may clean up and charge the cost thereof to the Contractor.

#### **ARTICLE 7**

#### MISCELLANEOUS PROVISIONS

#### 7.1 GOVERNING LAW

**7.1.1** The Contract shall be governed by the laws of the State of Vermont.

#### 7.2 SUCCESSORS AND ASSIGNS

**7.2.1** The State and the Contractor each bind themselves, their partners, successors, assigns and legal representatives to the other party hereto and to the partners, successors, assigns and legal representatives of such other party in respect to all covenants, agreements and obligations contained in the Contract Documents. Neither party to the Contract shall assign the Contract or sublet it as a whole without the written consent of the other, nor shall the Contractor assign any moneys due or to become due to him or her hereunder, without the previous written consent of the State.

#### 7.3 WRITTEN NOTICE

**7.3.1** Written notice shall be deemed to have been duly served if delivered in person to the individual or member of the firm or entity or to an officer of the corporation for whom it was intended, or if delivered at or sent by registered or certified mail to the last business address known to him who gives the notice.

#### 7.4 CLAIMS FOR DAMAGES

**7.4.1** Should either party to the Contract suffer injury or damage to person or property because of any act or omission of the other party or of any of his / her employees, agents or others for whose acts he / she is legally liable, claim shall be made in writing to such other party within a reasonable time after the first observance of such injury or damage.

#### 7.5 BOND REQUIREMENTS

The Contractor hereby agrees to comply with the State's bonding requirements as identified in the Instructions to Bidders which are included in the RFP and are attached and incorporated into the Contract Documents.

#### 7.6 RIGHTS AND REMEDIES

- **7.6.1** The duties and obligations imposed by the Contract Documents and the rights and remedies available there under shall be in addition to and not a limitation of any duties, obligations, rights and remedies otherwise imposed or available by law.
- **7.6.2** No action or failure to act by the State, Architect or Contractor shall constitute a waiver of any right or duty afforded any of them under the Contract, nor shall any such action or failure to act constitute an approval of or acquiescence in any breach thereunder, except as may be specifically agreed in writing.

#### 7.7 TESTS

- **7.7.1** If the Contract Documents, laws, ordinances, rules, regulations or orders of any public authority having jurisdiction require any portion of the Work to be inspected, tested or approved, the Contractor shall give the Architect timely notice of its readiness so the Architect may observe such inspection, testing or approval. The Contractor shall bear all costs of such inspections, tests or approvals conducted by public authorities.
- **7.7.2** If the Architect determines that any Work requires special inspection, testing, or approval which Subparagraph 7.7.1 does not include, he or she will, upon written authorization from the State, instruct the Contractor to order such special inspection, testing or approval, and the Contractor shall give notice as provided in Subparagraph 7.7.1. If such special inspection or testing reveals a failure of the Work to comply with the requirements of the Contract Documents, the Contractor shall bear all costs thereof, including compensation for the Architect's additional services made necessary by such failure; otherwise the State shall bear such costs, and an appropriate Change Order shall be issued.
- **7.7.3** Required certificates of inspection, testing or approval shall be secured by the Contractor and promptly delivered by him to the Architect.
- **7.7.4** If the Architect is to observe inspections, tests or approvals required by the Contract Documents, he will do so promptly.

#### 7.8 INTEREST

Payments due and unpaid under the Contract Documents shall bear interest from the date payment is due at a rate of twelve percent (12 %) per annum. A payment is NOT due and payable unless and until the State has received an invoice filled out correctly and completely with all required supporting documentation.

#### 7.9 DISPUTE RESOLUTION PROCESS

Any claim, dispute or other matter in question not resolved by the process identified in Paragraph

2.2.14 of these General Conditions shall be subject to the following dispute resolution process: Contractor may seek review by the Commissioner of Buildings and General Services of the Architect's recommendation. Contractor must seek review by the Commissioner within ten (10) days of the Architect's written recommendation otherwise the recommendation shall be deemed accepted by the Contractor. Failing resolution by the Commissioner, the Contractor then may request that the parties submit the claim to non-binding mediation with a mutually agreed upon mediator utilizing the most current Construction Industry Mediation Rules of the American Arbitration Association.

#### **ARTICLE 8**

#### **TIME**

#### 8.1 **DEFINITIONS**

- **8.1.1** Unless otherwise provided, the Contract Time is the period of time allotted in the Contract Documents for Substantial Completion of the Work as defined in Subparagraph 8.1.3 of these General Conditions, including authorized adjustments thereto.
- **8.1.2** The date of commencement of the Work is the date established in a notice to proceed. If there is no notice to proceed, it shall be the date indicated in State Contractor Agreement. In the absence of a notice to proceed and an express commencement date in the State Contractor Agreement, then the date of execution of the State Contractor Agreement shall be the commencement date.
- **8.1.3** The Date of Substantial Completion of the Work or designated portion thereof is the Date certified by the Architect when construction is sufficiently complete, in accordance with the Contract Documents, so the State can occupy or utilize the Work or designated portion thereof for the use for which it is intended. The Work shall not be considered substantially complete prior to the receipt of approved O & M manuals and record drawings as required by paragraph 4.11 and as required elsewhere in the Contract Documents.
  - **8.1.3.1** Contract completion date shall be defined as: The contract work shall not be considered complete prior to the receipt of approved O & M manuals and record drawings as required by paragraph 4.11 and as required elsewhere in the contract documents.
- **8.1.4** The term day as used in the Contract Documents shall mean calendar day unless otherwise specifically designated.

#### 8.2 PROGRESS AND COMPLETION

- **8.2.1** All time limits stated in the Contract Documents are material terms and time is the essence of the Contract. A failure by Contractor to do what is required by the time specified in the Contract Documents is a breach of the contract.
- 8.2.2 The Contractor shall begin the Work on the date of commencement as defined in

Subparagraph 8.1.2 of these General Conditions. He or She shall carry the Work forward expeditiously with adequate forces and shall achieve Substantial Completion within the Contract Time.

8.2.3 If in the sole opinion of the State, the Contractor fails to commence work on the project or to complete the work of said project within the time specified above, or to prosecute the work in such a manner that it appears that the completion date can be assured, the State shall have the right to notify the Contractor by Certified Mail that the terms of the Contract have been violated, and that effective immediately the Contract is terminated and the State has the right to and in fact is taking over and attending to completion of the project without prejudice to the State's remedies for any losses sustained

#### 8.3 DELAYS AND EXTENSIONS OF TIME

- **8.3.1** If the Contractor is delayed at any time in the progress of the Work by any act or failure to act by the State or the Architect, or by any employee of either, or by any separate contractor employed by the State, or by changes ordered in the Work, fire, unusual delay in transportation, adverse weather conditions not reasonably anticipatable, unavoidable casualties, or any causes beyond the Contractor's control, or by delay authorized by the State pending alternative dispute resolution proceedings, or by any other cause which may justify the delay, then the Contract Time shall be extended by Change Order for such reasonable time as recommended by the Architect and approved by the State.
- **8.3.2** Any claim for extension of time shall be made in writing to the Architect not more than twenty days after the commencement of the delay; otherwise it shall be waived. In the case of a continuing delay only one claim is necessary. The Contractor shall provide an estimate of the probable effect of such delay on the progress of the Work.
- **8.3.3** This Paragraph 8.3 does not exclude the recovery of damages for delay by either party under other provisions of the Contract Documents.

#### **ARTICLE 9**

#### PAYMENTS AND COMPLETION

#### 9.1 CONTRACT SUM

**9.1.1** The Contract Sum is stated in the State-Contractor Agreement and, including authorized adjustments thereto, is the total amount payable by the State to the Contractor for the performance of the Work under the Contract Documents.

#### 9.2 SCHEDULE OF VALUES

**9.2.1** Before the first Application for Payment, the Contractor shall submit to the Architect a schedule of values allocated to the various portions of the Work, prepared in such form and supported by such data to substantiate its accuracy as the Architect may require. This schedule, unless objected to by the Architect, shall be used only as a basis for the Contractor's Applications for Payment.

#### 9.3 APPLICATIONS FOR PAYMENT

- **9.3.1** At least ten days before the date for each progress payment established in the State-Contractor Agreement, the Contractor shall submit to the Architect an itemized Application for Payment, notarized if required, supported by such data substantiating the Contractor's right to payment as the State or the Architect may require. The application for payment must, at a minimum, reflect retainage and the required waivers of lien and any other support documentation enumerated elsewhere in the Contract Documents.
- **9.3.2** Unless otherwise provided in the Contract Documents, payments will be made on account of materials or equipment not incorporated in the Work but delivered and suitably stored at the site and, if approved in advance by the State, payments may similarly be made for materials or equipment suitably stored at some other location agreed upon in writing. Payments for materials or equipment stored on or off the site shall be conditioned upon submission by the Contractor of bills of sale or such other procedures satisfactory to the State to establish the State's title to such materials or equipment or otherwise protect the State's interest, including applicable insurance and transportation to the site for those materials and equipment stored off the site.
- **9.3.3** The Contractor warrants that title to all Work, materials and equipment covered by an Application for Payment will pass to the State either by incorporation in the construction or upon the receipt of payment by the Contractor, whichever occurs first, free and clear of all liens, claims, security interests or encumbrances, hereinafter referred to in this Article 9 as "liens"; and that no Work, materials or equipment covered by an Application for Payment will have been acquired by the Contractor, or by any other person performing Work at the site or furnishing materials and equipment for the Project, subject to an agreement under which an interest therein or an encumbrance thereon is retained by the seller or otherwise imposed by the Contractor or such other person.

#### 9.4 CERTIFICATES FOR PAYMENT

- **9.4.1** The Architect will, within seven days after the receipt of the Contractor's completed Application for Payment, either issue a Certificate for Payment to the State, with a copy to the Contractor, for such amount as the Architect determines is properly due, or notify the Contractor in writing his or her reasons for withholding a Certificate as provided in Subparagraph 9.6.1 of these General Conditions. In case of conflict between any time requirements in this paragraph and any time requirements for action by the Architect on Contractor's Application for Payment contained in the State-Contractor Agreement, the time requirements contained in the State-Contractor Agreement shall control.
- **9.4.2** The issuance of a Certificate for Payment will constitute a representation by the Architect to the State, based on his observations at the site as provided in Subparagraph 2.2.3 of these General Conditions and the data comprising the Application for Payment, that the Work has progressed to the point indicated; that, to the best of his or her knowledge, information and belief, the quality of the Work is in accordance with the Contract Documents (subject to an evaluation of the Work for conformance with the Contract Documents upon Substantial Completion, to the results of any subsequent tests required by or performed under the Contract Documents, to minor deviations from the Contract Documents correctable prior to completion, and to any specific

qualifications stated in his Certificate); and that the Contractor is entitled to payment in the amount certified.

#### 9.5 PROGRESS PAYMENTS

- **9.5.1** After the Architect has issued a Certificate for Payment, the State shall make payment in the manner and within the time provided in the Contract Documents.
- **9.5.2** The Contractor shall promptly pay each Subcontractor, upon receipt of payment from the State, out of the amount paid to the Contractor on account of such Subcontractor's Work, the amount to which said Subcontractor is entitled, reflecting the percentage actually retained, if any, from payments to the Contractor on account of such Subcontractor's Work. The Contractor shall, by an appropriate agreement with each Subcontractor, require each Subcontractor to make payments to his or her Sub-subcontractors in similar manner.
- **9.5.3** The Architect may, on request and at his discretion, furnish to any Subcontractor, if practicable, information regarding the percentages of completion or the amounts applied for by the Contractor and the action taken thereon by the Architect on account of Work done by such Subcontractor.
- **9.5.4** Neither the State nor the Architect shall have any obligation to pay or to see to the payment of any moneys to any Subcontractor.
- **9.5.5** No Certificate for a progress payment, nor any progress payment, nor any partial or entire use or occupancy of the Project by the State, shall constitute an acceptance of any Work not in accordance with the Contract Documents.

#### 9.6 PAYMENTS WITHHELD

- **9.6.1** The Architect may decline to certify payment and may withhold his or her Certificate in whole or in part, to the extent necessary reasonably to protect the State, if in the Architect's opinion he or she is unable to make representations to the State as provided in Subparagraph 9.4.2 of these General Conditions. If the Architect is unable to make representations to the State as provided in the aforementioned Subparagraph 9.4.2 and to certify payment in the amount of the Application, he or she will notify the Contractor as provided in Subparagraph 9.4.1 of these General Conditions. If the Contractor and the Architect cannot agree on a revised amount, the Architect will promptly issue a Certificate for Payment for the amount for which he or she is able to make such representations to the State. The Architect may also decline to certify payment or, because of subsequently discovered evidence or subsequent observations, he or she may nullify the whole or any part of any Certificate for Payment previously issued, to such extent as may be necessary in his or her opinion to protect the State from loss because of:
  - (a) Defective Work not remedied,
  - (b) Third party claims filed or reasonable evidence indicating probable filing of such claims,
  - (c) Failure of the Contractor to make payments properly to Subcontractors or for labor, materials or equipment,
  - (d) Reasonable evidence that the Work cannot be completed for the unpaid balance of the Contract Sum,

- (e) Damage to the State or another contractor,
- (f) Reasonable evidence that the Work will not be completed within the Contract Time, or
- (g) Persistent failure to carry out the Work in accordance with the Contract Documents.
- **9.6.2** When the above grounds in Subparagraph 9.6.1 are removed, payment shall be made for amounts withheld because of them.

#### 9.7 FAILURE OF PAYMENT

**9.7.1** If the Architect does not issue a Certificate for Payment, through no fault of the Contractor, within seven days after receipt of the Contractor's Application for Payment, or if the State does not pay the Contractor within seven days after the date established in the Contract Documents any amount certified by the Architect or awarded by alternative dispute resolution proceedings, then the Contractor may, upon seven additional days written notice to the State and the Architect, stop the Work until payment of the amount owing has been received. The Contract Sum shall be increased by the amount of the Contractor's reasonable costs of shut-down, delay and start-up, which shall be effected by appropriate Change Order in accordance with Paragraph 12.3.

#### 9.8 SUBSTANTIAL COMPLETION

- **9.8.1** When the Contractor considers that the Work, or a designated portion thereof which has been accepted in writing to by the State, is substantially complete as defined in Subparagraph 8.1.3 of these General Conditions, the Contractor shall prepare for submission to the Architect a list of items to be completed or corrected. The failure to include any items on such list does not alter the responsibility of the Contractor to complete all Work in accordance with the Contract Documents. When the Architect on the basis of an inspection determines that the Work or designated portion thereof is substantially complete, he will then prepare a Certificate of Substantial Completion which shall establish the Date of Substantial Completion, shall state the responsibilities of the State and the Contractor for security, maintenance, heat, utilities, damage to the Work, and insurance, and shall fix the time, which shall not exceed thirty (30) days, within which the Contractor shall complete the items listed therein. Warranties required by the Contract Documents shall commence on the Date of Substantial Completion of the Work or designated portion thereof unless otherwise provided in the Certificate of Substantial Completion. The Certificate of Substantial Completion shall be submitted to the State and the Contractor for their written acceptance of the responsibilities assigned to them in such Certificate. The Work shall not be considered substantially complete prior to the receipt of approved O & M manuals and record drawings as required by paragraph 4.11 and as required elsewhere in the Contract Documents.
- **9.8.2** Upon Substantial Completion of the Work or designated portion thereof and upon application by the Contractor and certification by the Architect, the State shall make payment, reflecting adjustment in retainage, if any, for such Work or portion thereof, as provided in the Contract Documents.

#### 9.9 FINAL COMPLETION AND FINAL PAYMENT

**9.9.1** Upon receipt of written notice that the Work is ready for final inspection and acceptance and upon receipt of a final Application for Payment, the Architect will promptly make such

inspection and, when he or she finds the Work acceptable under the Contract Documents and the Contract fully performed, he or she will promptly issue a final Certificate for Payment stating that to the best of his / her knowledge, information and belief, and on the basis of his or her observations and inspections, the Work has been completed in accordance with the terms and conditions of the Contract Documents and that the entire balance found to be due the Contractor, and noted in said final Certificate, is due and payable. The Architect's final Certificate for Payment will constitute a further representation that the Contractor has fulfilled the conditions entitling him or her to final payment as set forth in Subparagraph 9.9.2 of these General Conditions.

- **9.9.2** Neither the final payment nor the remaining retained percentage shall become due until the Contractor submits to the Architect (1) an affidavit that all payrolls, bills for materials and equipment, and other indebtedness connected with the Work for which the State or its property might in any way be responsible, have been paid or otherwise satisfied; (2) consent of surety, if any, to final payment; (3) Other data establishing payment or satisfaction of all such obligations, including, but not limited to, receipts, releases, or other supporting documentation. and (4) final waivers of liens arising out of the Contract, to the extent and in such form as may be designated by the State. If any Subcontractor refuses to furnish a release or waiver required by the State, the Contractor may furnish a bond satisfactory to the State to indemnify him against any such lien. If any such lien remains unsatisfied after all payments are made, the Contractor shall refund to the State all moneys that the latter may be compelled to pay in discharging such lien, including all costs and reasonable attorneys' fees.
- **9.9.3** If, after Substantial Completion of the Work, final completion thereof is materially delayed through no fault of the Contractor or by the issuance of Change Orders affecting final completion, and the Architect so confirms, the State shall, upon application by the Contractor and certification by the Architect, and without terminating the Contract, make payment of the balance due for that portion of the Work fully completed and accepted. If the remaining balance for Work not fully completed or corrected is less than the retainage stipulated in the Contract Documents, and if bonds have been furnished as required by the Contract Documents, the written consent of the surety to the payment of the balance due for that portion of the Work fully completed and accepted shall be submitted by the Contractor to the Architect prior to certification of such payment. Such payment shall be made under the terms and conditions governing final payment, except that it shall not constitute a waiver of claims.
- **9.9.4** The making of final payment shall constitute a waiver of all claims by the State except those arising from:
  - (a) unsettled liens,
  - (b) faulty or defective Work appearing after Substantial Completion,
  - (c) failure of the Work to comply with the requirements of the Contract Documents, or
  - (d) terms of any special warranties required by the Contract Documents.
- **9.9.5** The acceptance of final payment shall constitute a waiver of all claims by the Contractor except those previously made in writing and identified by the Contractor as unsettled at the time of the final Application for Payment.

#### PROTECTION OF PERSONS AND PROPERTY

#### 10.1 SAFETY PRECAUTIONS AND PROGRAMS

The Contractor shall be responsible for initiating, maintaining and supervising all safety precautions and programs in connection with the Work.

#### 10.2 SAFETY OF PERSONS AND PROPERTY

- **10.2.1** The Contractor shall take all reasonable precautions for the safety of, and shall provide all reasonable protection to prevent damage, injury or loss to:
  - (a) all employees on the Work and all other persons who may be affected thereby;
  - (b) all the Work and all materials and equipment to be incorporated therein, whether in storage on or off-the site, under the care; custody or control of the Contractor or any of his Subcontractors or Sub-subcontractors; and
  - (c) other property at the site or adjacent thereto, including trees, shrubs; lawns, walks, pavements, roadways, structures and utilities not designated for removal, relocation or replacement in the course of construction.
- **10.2.2** The Contractor shall give all notices and comply with all applicable laws, ordinances, rules, regulations and lawful orders of any public authority bearing on the safety of persons or property or their protection from damage, injury or loss.
- **10.2.3** The Contractor shall erect and maintain, as required by existing conditions and progress of the Work, all reasonable safeguards for safety and protection, including posting danger signs and other warnings against hazards, promulgating safety regulations and notifying owners and users of adjacent utilities.
- **10.2.4** When the use or storage of explosives or other hazardous materials or equipment is necessary for the execution of the Work, the Contractor shall: (1) exercise the utmost care and shall carry on such activities under the supervision of properly qualified personnel; and (2) give the State notice, in writing, seven (7) days in advance of the planned activity of the Contractor's intent to store and/ or use explosives or other hazardous materials or equipment on the project site.
- **10.2.5** When the project involves the placement or replacement of roofing systems of all types on State- owned buildings, including flat, low-pitched and steep roofs, Contractor must comply with all requirements of the latest edition of VOSHA Safety and Health Standards for Construction (29 CFR 1926) including, but not limited to, Subpart M Fall Protection. Prior to execution of a contract by BGS, contractors engaged in placement or replacement of roofing systems of all types must submit a signed certification statement attesting to their intention to comply with VOSHA Fall Protection Regulations. Such certification shall be submitted on a State-provided form along with proof of insurance.
- **10.2.6** The Contractor shall promptly remedy all damage or loss (consistent with Contractor's obligations described in Attachment C) to any property referred to in Clauses 10.2.1(b) and 10.2.1(c) of these General Conditions caused in whole or in part by the Contractor, any

Subcontractor, any Sub-subcontractor, or anyone directly or indirectly employed by any of them, or by anyone for whose acts any of them may be liable and for which the Contractor is responsible under Clauses 10.2.1(b) and 10.2.1(c), except damage or loss attributable to the acts or omissions of the State or Architect or anyone directly or indirectly employed by either of them, or by anyone for whose acts either of them may be liable, and not attributable to the fault or negligence of the Contractor. The foregoing obligations of the Contractor are in addition to his obligations referenced under Paragraph 4.18.

- **10.2.7** The Contractor shall designate a responsible member of his organization at the site whose duty shall be the prevention of accidents. This person shall be the Contractor's superintendent unless otherwise designated by the Contractor in writing to the State and the Architect.
- **10.2.8** The Contractor shall not load or permit any part of the Work to be loaded so as to endanger its safety.

#### 10.3 EMERGENCIES

**10.3.1** In any emergency affecting the safety of persons or property, the Contractor shall act to prevent threatened damage, injury or loss. Any additional compensation or extension of time claimed by the Contractor on account of emergency work shall be determined as provided in Article 12 for Changes in the Work.

#### **ARTICLE 11**

#### **INSURANCE**

#### Insurance obligations of the parties to this agreement are fully set forth below:

Attachment "C", numbered paragraph 8 has been deleted in its entirety and the following provisions are applicable:

#### 11 INSURANCE – GENERAL REQUIREMENTS

- 11.1 Insurance obtained by the Contractor to cover the below-listed requirements shall be procured from an insurance company registered and licensed to do business in the State of Vermont. Before the Contract is executed, the Contractor shall file with the State a certificate of insurance, executed by an insurance company or its licensed agent(s), on form(s) satisfactory to the State, stating that with respect to the Contract awarded, the Contractor carries insurance in accordance with the following requirements:
  - (a) Workers' Compensation Insurance: With respect to all operations performed, the Party shall carry workers' compensation insurance in accordance with the laws of the State of Vermont. Vermont will accept an out-of-state employer's workers' compensation coverage while operating in Vermont provided that the insurance carrier is licensed to write insurance in Vermont and an amendatory endorsement is added to the policy adding Vermont for coverage purposes. Otherwise, the party shall secure a Vermont workers' compensation policy, if necessary to comply with Vermont law.

(b) Commercial General Liability (CGL) Insurance: With respect to all operations performed by the Contractor and any Subcontractors, the Contractor shall carry Commercial General Liability (CGL) Insurance affording all major divisions of coverage including, but not limited to:

Premises Operations
Independent Contractors' Protective Products and Completed Operations
Personal Injury Liability
Contractual Liability
Broad Form Property Damage
Explosion, Collapse and Underground (XCU) Coverage
Fire Legal Liability

State may require additional specific liability coverage(s) when applicable.

Limits of Coverage shall be not less than:

\$1,000,000 Each Occurrence \$2,000,000 General Aggregate Applying, In Total To This Project Only \$1,000,000 Products/Completed Operations Aggregate \$1,000,000 Personal Injury and Advertising Liability \$50,000 Fire Legal Liability \$5,000 Medical Expense

Aggregate Limits of Insurance (Per Project) Form must be included (or as excepted in f below) with the Commercial General Liability Policy.

(c) Automobile Liability Insurance: Contractor shall carry Automobile Liability Insurance covering all motor vehicles, including owned, hired, borrowed and non-owned vehicles, used in connection with the project. Limits of Coverage shall be not less than:

\$1,000,000 Combined Single Limit

- (d) The Contractor shall require each Subcontractor, if any, to maintain the same Workers' Compensation, Commercial General Liability, and Automobile Liability coverage as specified in paragraphs (a), (b), and (c) above.
- (e) Scope of Insurance & Special Hazards: The insurance required under subparagraphs (b), (c), and (d) above shall provide protection for the Contractor and the Subcontractors, if any, against damage claims which may arise from work being performed. Special hazards, may be encountered by the insured or by any person directly or indirectly employed by the Contractor or a Subcontractor, and may be required.
- (f) If Contractor is unable to provide Aggregate Limits of Insurance (Per Project) as required in (b) above, Contractor shall purchase an Owner's Protective Liability Insurance Policy in the name of the State of Vermont to cover all exposures, including bodily injury and death, arising out of and in the course of this Contract. Limits of liability shall be the same as those required of the Contractor under paragraph (b) above.

Additional Insured. The General Liability and Property Damage coverages required for performance of this Agreement shall include the State of Vermont and its agencies, departments, officers and employees as Additional Insureds. If performance of this Agreement involves construction, or the transport of persons or hazardous materials, then the required Automotive Liability coverage shall include the State of Vermont and its agencies, departments, officers and employees as Additional Insureds. Coverage shall be primary and non-contributory with any other insurance and self-insurance.

Notice of Cancellation or Change. There shall be no cancellation, change, potential exhaustion of aggregate limits or non-renewal of insurance coverage(s) without thirty (30) days written prior written notice to the State of Vermont, Office of Purchasing & Contracting.

**11.2 Insurance Requirements for Builder's Risk Insurance.** The State of Vermont does not require Contractor to obtain Builder's Risk insurance for every construction project. The State of Vermont will evaluate each project prior to accepting bids to determine if the State will require the Contractor to obtain and maintain Builder's Risk insurance for the project.

Any project that involves NEW CONSTRUCTION will require the Contractor to obtain and maintain Builder's Risk insurance for the project.

At the discretion of the State, Contractor may not be required to obtain and maintain Builder's Risk insurance on CONSTRUCTION RENOVATION projects. Whether or not the State is going to require Contractor to obtain and maintain Builder's Risk insurance for a particular CONSTRUCTION RENOVATION project will be determined by the State prior to bidding the project and the requirement will be included in the Instructions to Bidders.

- 11.2.1 BUILDER'S RISK FOR NEW CONSTRUCTION PROJECTS. If the project is NEW CONSTRUCTION, then the following provisions apply:
  - (1) The State will require Contractor to purchase and maintain Builder's Risk Insurance. The Contractor shall name the Contractor and the State of Vermont as their interest may occur. Other parties shall be insured as the State may reasonably require.
  - (2) Contractor shall effect and maintain insurance on portions of the work stored off-site, on site and in transit. Boiler & Machinery Insurance may be used in conjunction with this coverage if it is required to meet the testing requirement.
  - (3) Any deductible applicable to loss covered by insurance provided hereunder shall be borne by the Contractor.
  - (4) Except as provided for in (1) above the State and Contractor waive all rights against each other and the Subcontractor, Sub-subcontractors, agents and employees of the other.
- 11.2. 2 BUILDER'S RISK FOR CONSTRUCTION RENOVATION PROJECTS. If the project is CONSTRUCTION RENOVATION of an existing structure, and the State has already determined that Contractor will not be obligated to obtain and maintain Builder's Risk insurance, then the following provisions apply:

- (1) The State will maintain property insurance upon the construction site and will not require Contractor to purchase and maintain Builder's Risk Insurance upon the entire work at the site.
- (2) The decision of the State to waive the requirement that the Contractor maintain builder's risk coverage does not waive Contractor's liability for damage to the State's real and personal property. Contractor's liability for loss to the State's real and personal property will be limited to the first \$100,000 of each and every property loss at the work site provided such loss is covered under the State's property insurance coverage. If the Contractor elects to meet this obligation by purchase of commercial insurance, this insurance shall name the Contractor and the State of Vermont as Named Insureds and shall include the interests of the Contractor and Subcontractors. Other parties shall be insured as the State may reasonably require.
- (3) Contractor shall effect and maintain insurance on portions of the work stored off-site, on site and in transit. Boiler & Machinery Insurance may be used in conjunction with this coverage if it is required to meet the testing requirement. The cost of any deductible applicable to loss covered by insurance provided hereunder shall be borne by the Contractor.
- (4) Except as provided for in 11.2.2 (1)-(3) above the State and Contractor waive all rights against each other and the Subcontractor, Sub-subcontractors, agents and employees of the other.

#### 11.2.3 Builder's Risk – additional provisions

The insurance specified shall be maintained in force until final acceptance of the project by the State.

- (5) Umbrella Excess Liability Policies may be used in conjunction with primary policies to comply with any of the limit requirements specified above.
- (6) "Claims-made" coverage forms are not acceptable without the express written prior consent of the State. Each policy furnished shall contain a rider or non-cancellation clause reading in substance as follows:
  - Anything herein to the contrary notwithstanding, notice of any cancellation, termination or alteration to the insurance contracts must be delivered by registered mail to the Commissioner, Department of Buildings and General Services, State of Vermont, Montpelier, Vermont, at least 60 days before effective cancellation, termination or alteration date unless all work required to be performed under the terms of the Contract is satisfactorily completed as evidenced by the formal acceptance by the State of Vermont.
- (7) No warranty is made that the coverages and limits listed herein are adequate to cover and protect the interests of the Contractor for the Contractor's operations. These are solely minimums that have been set to protect the interests of the State.
- **11.3** The State shall have power to adjust and settle any loss with the insurers.

#### **CHANGES IN THE WORK**

#### 12.1 CHANGE ORDERS

**12.1.1** A Change Order is a written order to the Contractor signed by the State, the Architect, and the Contractor issued after execution of the Contract, authorizing a change in the Work or an adjustment in the Contract Sum or the Contract Time. The Contract Sum and the Contract Time may be changed only by Change Order. A Change Order signed by the Contractor indicates his agreement therewith, including the adjustment in the Contract Sum or the Contract Time. A Proposed Change Order (PCO) is a written request submitted to the Architect by either the Contractor or the State requesting a change to the contract price and / or schedule. A Construction Change Directive (CCD) is a written document prepared by the State and / or Architect directing the Contractor to make changes in the Work where the State and the Contractor have not reached an agreement on proposed changes in the contract sum or contract time. A Construction Change Directive becomes part of a PCO.

Payment for changes in the Work will only be made after a Change Order is signed by the Contractor and Architect and executed by the State.

**12.1.2** The State, without invalidating the Contract, may order changes in the Work within the general scope of the Contract consisting of additions, deletions or other revisions, the Contract Sum and the Contract Time being adjusted accordingly. All such changes in the Work REQUIRE AGREEMENT IN WRITING BY THE ACHITECT, STATE AND CONTRACTOR AND shall be authorized by A PROPOSED Change Order (PCO), and shall be performed under the applicable conditions of the Contract Documents. The State will prescribe a Proposed Change Order form for use by the Contractor. The Contractor shall submit all requests for Proposed Change Orders on the prescribed form.

The process, by which changes in the Work within the general scope of the Contract consisting of additions, deletions or other revisions, is as follows:

- (1) A Proposed Change Order (PCO) is submitted to the Architect for review. Said PCO is to be submitted on the State approved form.
- (2) If the PCO originated with the Contractor, the Architect will review the PCO and if the Architect approves the PCO, it is then presented to the State for approval.
- (3) The PCO is signed by the Project Manager.
- (4) The Project Manager shall coordinate with the Contractor through email or other written documentation every 30 days as to the merit of issuing a Change Order. If the Project Manager and Contractor agree that a Change Order should be submitted, the Architect will compile all approved PCOs and incorporate them into one (1) Change Order.
- (5) The Change Order is then submitted to the Project Manager who immediately processes the CO with OPC as required by Bulletin 3.5 and BGS' Contracting Plan.
- **12.1.3** The cost or credit to the State resulting from a change in the Work shall be determined in one or more of the following ways:
  - (a) by mutual acceptance of a lump sum properly itemized and supported by sufficient substantiating data to permit evaluation;
  - **(b)** by unit prices stated in the Contract Documents or subsequently agreed upon;
  - (c) by cost to be determined in a manner agreed upon by the parties and a mutually

acceptable fixed or percentage fee; or **(d)** by the method provided in Subparagraph 12.1.4.

12.1.4 If none of the methods set forth in Clauses 12.1.3(a) –(d) is agreed upon, or cannot be agreed upon at the time due to the nature of the change, emergent conditions, or lack of required information, the Contractor, provided he receives a Construction Change Directive (CCD)) signed by the State, shall promptly proceed with the Work involved. Upon receipt of the CCD, the Contractor shall proceed with implementation of the CCD. The cost of such Work shall then be recommended by the Architect on the basis of the reasonable expenditures and savings of those performing the Work attributable to the change, including, in the case of an increase in the Contract Sum, a reasonable allowance for overhead and profit. In such case, and also under Clauses 12.1.3(c) and 12.1.3(d) above, the Contractor shall keep and present, in such form as the Architect may prescribe, an itemized accounting together with appropriate supporting data for inclusion in the CCD. Unless otherwise provided in the Contract Documents, cost shall be limited to the following: cost of materials, cost of delivery; cost of labor, including social security, medicare and unemployment insurance, and fringe benefits required by agreement or custom; workers' or workmen's compensation insurance; bond premiums; rental value of equipment and machinery; and the additional costs of supervision and field office personnel directly attributable to the change. Pending final determination of cost to the State, payments on account shall be made on the State's Certificate for Payment. The amount of credit to be allowed by the Contractor to the State for any deletion or change which results in a net decrease in the Contract Sum will be the amount of the actual net cost as confirmed by the Architect. When both additions and credits covering related Work or substitutions are involved in any one change, the allowance for Overhead and profit shall be figured on the basis of the net increase, if any, with respect to that change. It is the Project Manager's responsibility to ensure that All CCD's are converted / incorporated into a PCO within Thirty (30) days of completion of the changes to the Work that was the subject of the CCD. The parties agree to utilize the CCD form provided or approved by the State.

**12.1.5** If unit prices are stated in the Contract Documents or subsequently agreed upon, and if the quantities originally contemplated are so changed in a Proposed Change Order that application of the agreed unit prices to the quantities of Work proposed will cause substantial in- equity to the State or the Contractor, the applicable unit prices shall be equitably adjusted.

#### 12.2 CONCEALED CONDITIONS

12.2.1 Should concealed conditions encountered in the performance of the Work below the surface of the ground, or should concealed or unknown conditions in an existing structure be at variance with the conditions indicated by the Contract Documents or should unknown physical conditions below the surface of the ground or should concealed or unknown conditions in an existing structure of an unusual nature, differing materially from those ordinarily encountered and generally recognized as inherent in work of the character provided for in this Contract, be encountered, the Contract Sum shall be equitably adjusted by Change Order upon claim by either

party made within twenty days after the first observance of the conditions.

#### 12.3 CLAIMS FOR ADDITIONAL COST

- 12.3.1 If the Contractor wishes to make a claim for an increase in the Contract Sum, he shall give the Architect written notice thereof within twenty days after the occurrence of the event giving rise to such claim. This notice shall be given by the Contractor before proceeding to execute the Work, except in an emergency endangering life or property in which case the Contractor shall proceed in accordance with Paragraph 10.3 of these General Conditions. No such claim shall be valid unless so made. If the State and the Contractor cannot agree on the amount of the adjustment in the Contract Sum, it shall be determined by the Architect. Any change in the Contract Sum resulting from such claim shall be authorized by Change Order.
- **12.3.2** If the Contractor claims that additional cost is involved because of, but not limited to, (1) any order by the State to stop the Work pursuant to Paragraph 3.3 of these General Conditions where the Contractor was not at fault, or (2) failure of payment by the State pursuant to Paragraph 9.7(of these General Conditions), the Contractor shall make such claim as provided in Subparagraph 12.3.1 of this agreement.
- **12.3.3** The Architect will have authority to order minor changes in the Work not involving an adjustment in the Contract Sum or an extension of the Contract Time and not inconsistent with the intent of the Contract Documents. Such changes shall be effected by written order and shall be binding on the State and Contractor within 10 days unless the Contractor or the State objects to the change in writing.

#### **ARTICLE 13**

#### UNCOVERING AND CORRECTION OF WORK

#### 13.1 UNCOVERING OF WORK

- **13.1.1** If any portion of the Work should be covered contrary to the request of the Architect or to requirements specifically expressed in the Contract Documents, it must, if required in writing by the Architect, be uncovered for his or her observation and shall be replaced at the Contractor's expense.
- 13.1.2 If any other portion of the Work has been covered which the Architect has not specifically requested to observe prior to being covered, the Architect may request to see such Work and it shall be uncovered by the Contractor. If such Work is found in accordance with the Contract Documents, the cost of uncovering and replacement shall, by appropriate Change Order, be charged to the State. If such Work is found not to be in accordance with the Contract Documents, the Contractor shall pay such costs unless it is found that this condition was caused by the State or a separate contractor as provided in Article 6 of these General Conditions, in which event the State shall be responsible for the payment of such costs.

#### 13.2 CORRECTION OF WORK

**13.2.1** The Contractor shall be responsible for correcting all Work which the Architect has found to be defective or which fails to conform to the Contract Documents whether observed before or after Substantial Completion and whether or not fabricated, installed or completed. The Contractor shall bear all costs of correcting such rejected Work, including compensation for the Architect's and the State's additional services made necessary thereby.

The Architect, upon a finding of defect or failure to conform, shall immediately notify the State and Contractor, in writing, of the defect. The Contractor shall begin correcting the defective or non-conforming Work within ten (10) days unless the State agrees to a Change Order which reflects the reduction in Contract Sum due to the value of diminishment of the defective or nonconforming Work.

- 13.2.2 If, within one year after the Date of Substantial Completion of the Work or designated portion thereof or within one year after acceptance by the State of designated equipment or within such longer period of time as may be prescribed by law or by the terms of any applicable special warranty required by the Contract Documents, any of the Work is found to be defective or not in accordance with the Contract Documents, the Contractor shall correct it promptly after receipt of a written notice from the State to do so unless the State has previously given the Contractor a written acceptance of such condition. This obligation shall survive termination of the Contract. The State shall give such notice promptly after discovery of the condition.
- **13.2.3** The Contractor shall remove from the site all portions of the Work which are defective or non-conforming and which have not been corrected under Subparagraphs 4.5.1, 13.2.1 and 13.2.2 of these General Conditions, unless removal is waived by the State.
- **13.2.4** If the Contractor fails to correct defective or nonconforming Work as provided in Subparagraphs 4.5.1, 13.2.1 and 13.2.2, the State may correct it in accordance with Paragraph 3.4 of these General Conditions.
- 13.2.5 If the Contractor does not proceed with the correction of such defective or non- conforming Work within a reasonable time fixed by written notice from the Architect, the State may remove it and may store the materials or equipment at the expense of the Contractor. If the Contractor does not pay the cost of such removal and storage within ten days thereafter, the State may upon ten additional days written notice sell such Work at auction or at private sale and shall account for the net proceeds thereof, after deducting all the costs that should have been borne by the Contractor, including compensation for the Architect's and the State's additional services and expenses made necessary thereby. If such proceeds of sale do not cover all costs which the Contractor should have borne, the difference shall be charged to the Contractor and an appropriate Change Order shall be issued. If the payments then or thereafter due the Contractor are not sufficient to cover such amount, the Contractor shall pay the difference to the State.
- **13.2.6** The Contractor shall bear the cost of making good all work of the State or separate contractors destroyed or damaged by such correction or removal.
- 13.2.7 Nothing contained in this Paragraph 13.2 shall be construed to establish a period of limitation with respect to any other obligation which the Contractor might have under the

Contract Documents, including Paragraph 4.5 of these General Conditions. The establishment of the time period of one year after the Date of Substantial Completion or such longer period of time as may be prescribed by law or by the terms of any warranty required by the Contract Documents relates only to the specific obligation of the Contractor to correct the Work, and has no relationship to the time within which his obligation to comply with the Contract Documents may be sought to be enforced, nor to the time within which proceedings may be commenced to establish the Contractor's liability with respect to his obligations other than specifically to correct the Work.

#### 13.3 ACCEPTANCE OF DEFECTIVE OR NON-CONFORMING WORK

13.3.1 The State may accept defective or nonconforming Work pursuant to Paragraph 13.2.1 of this agreement. If the State elects to accept the defective or nonconforming Work, a Change Order will be issued to reflect a reduction in the Contract Sum. The Architect will recommend to the State the value of diminishment of the defective or nonconforming Work. Such adjustment shall be effected whether or not final payment has been made.

#### **ARTICLE 14**

#### TERMINATION AND SUSPENSION OF THE CONTRACT

#### 14.1 TERMINATION BY THE CONTRACTOR

If the Work is stopped for a period of thirty days under an order of any court or other public authority having jurisdiction, or as a result of an act of government, such as a declaration of a national emergency making materials unavailable, through no act or fault of the Contractor or a Subcontractor or their agents or employees or any other persons performing any of the Work under a contract with the Contractor, or if the Work should be stopped for a period of thirty days by the Contractor because the Architect has not issued a Certificate for Payment as provided in Paragraph 9.7 of these General Conditions or because the State has not made payment thereon as provided in Paragraph 9.7, then the Contractor may, upon seven additional days written notice to the State and the Architect, terminate the Contract and recover from the State payment for all Work executed and for any proven loss sustained upon any materials, equipment, tools, construction equipment and machinery, including reasonable profit and damages.

#### 14.2 TERMINATION BY THE STATE FOR CAUSE

14.2.1 If the Contractor is adjudged a bankrupt, or if he or she makes a general assignment for the benefit of his/ her creditors, or if a receiver is appointed on account of his/ her insolvency, or if he or she persistently or repeatedly refuses or fails, except in cases for which extension of time is provided, to supply enough properly skilled workmen or proper materials, or if he or she fails to make prompt payment to Subcontractors or for materials or labor, or persistently disregards laws, ordinances, rules, regulations or orders of any public authority having jurisdiction, or otherwise is guilty of a substantial violation of a provision of the Contract Documents, then the State, upon certification by the Architect that sufficient cause exists to justify such action, may, without prejudice to any right or remedy and after giving the Contractor and his surety, if any, seven days written notice, terminate the employment of the Contractor and take possession of the

site and of all materials, equipment, tools, construction equipment and machinery thereon owned by the Contractor and may finish the Work by whatever method he may deem expedient. In such case the Contractor shall not be entitled to receive any further payment until the Work is finished.

**14.2.2** If the costs of finishing the Work, including compensation for the Architect's and State's additional services and expenses made necessary thereby, exceed the unpaid balance of the Contract Sum, the Contractor shall pay the difference to the State.

If the unpaid balance of the Contract Sum exceeds the costs of finishing the Work, including compensation for the Architect's and the State's additional services and expenses made necessary thereby, such excess shall be paid to the Contractor.

The amount owed by Contractor to the State shall be certified by the Architect, upon application, in the manner provided in section 9.4 of these General Conditions, and this obligation for payment shall survive the termination of the Contract.

#### 14.3 TERMINATION BY STATE FOR CONVENIENCE

- 14.3.1 The State may, at any time, terminate this agreement for the State's convenience and without cause.
- 14.3.2 Upon receipt of written notice from the State of such termination for the State's convenience, the Contractor shall:
  - (a) cease operations as directed by the State in the notice;
  - (b) take actions necessary, or that the State may direct, for the protection and preservation of the Work; and
  - (c) except for Work directed to be performed prior to the effective date of termination stated in the notice, terminate all existing subcontracts and purchase orders and enter into no further subcontracts and purchase orders.
- 14.3.3 In case of such termination for the State's convenience, the Contractor shall be entitled to receive payment for Work executed, and costs incurred by reason of such termination, along with reasonable overhead and profit on the Work not executed.

#### 14.4 SUSPENSION BY STATE FOR CONVEVIENCE

- 14.4.1 The State may, without cause, order the Contractor in writing to suspend, delay or interrupt the Work in whole or in part for such period of time as the State may determine.
- 14.4.2 The Contract Sum and Contract Time shall be adjusted for increases in the cost and time caused by suspension, delay or interruption as described in section 14.4.1. Adjustment of the Contract Sum shall include profit. No adjustment shall be made to the extent:
  - (a) that performance is, was or would have been so suspended, delayed or interrupted by another cause for which the Contractor is responsible; or
  - (b) that an equitable adjustment is made or denied under another provision of this Contract.

# STATE OF VERMONT CONTRACT AMENDMENT

It is hereby agreed by and between the State of Vermont, Department of Buildings & General Services (the "State") and Limelite Restoration Services Inc., with a principal place of business in Irasburg, VT (the "Contractor") that the contract between them originally dated as of March 1, 2019, Contract #37787, as amended to date, (the "Contract") is hereby amended as follows:

- I. <u>Contract Term</u>. The Contract end date, wherever such reference appears in the Contract, shall be changed from February 28, 2020 to February 28, 2021.
- II. <u>Attachment B, Payment Provisions</u>. Section 4 of Attachment B has been amended to allow for the Consumer Price Index unadjusted percentage change increase, of 2.1%, for the prior 12 months of the contract.

Job Title	Hourly Rate	Hourly Rate for off-hours,	
		weekends, holidays	
Supervisor	\$66.37	\$75.58	
Technician	\$56.16	\$66.37	
Labor	\$40.84	\$51.05	
Administrator	\$30.63	\$35.74	

<u>Taxes Due to the State</u>. Contractor certifies under the pains and penalties of perjury that, as of the date this contract amendment is signed, the Contractor is in good standing with respect to, or in full compliance with a plan to pay, any and all taxes due the State of Vermont.

<u>Child Support (Applicable to natural persons only; not applicable to corporations, partnerships or LLCs)</u>. Contractor is under no obligation to pay child support or is in good standing with respect to or in full compliance with a plan to pay any and all child support payable under a support order as of the date of this amendment.

<u>Certification Regarding Suspension or Debarment</u>. Contractor certifies under the pains and penalties of perjury that, as of the date this contract amendment is signed, neither Contractor nor Contractor's principals (officers, directors, owners, or partners) are presently debarred, suspended, proposed for debarment, declared ineligible or excluded from participation in federal programs, or programs supported in whole or in part by federal funds.

Contractor further certifies under pains and penalties of perjury that, as of the date this contract amendment is signed, Contractor is not presently debarred, suspended, nor named on the State's debarment list at: <a href="http://bgs.vermont.gov/purchasing-contracting/debarment">http://bgs.vermont.gov/purchasing-contracting/debarment</a>

This document consists of 2 pages. Except as modified by this Amendment No. 2, all provisions of the Contract remain in full force and effect.

[Remainder of Page Intentionally Left Blank]

Revision Date: 12/12/2018

The signatures of the undersigned indicate that each has read and agrees to be bound by this Amendment to the Contract.

STATE OF VERMONT	LIMELITE RESTORATION SE	RVICES, INC.
By:	By:	
Name:	Name:	_
Title:	Title:	
Date:	<b>Date:</b>	

## STATE OF VERMONT CONTRACT AMENDMENT

It is hereby agreed by and between the State of Vermont, Department of Buildings and General Services (the "State") and LimeLite Restoration Services, Inc. with a principal place of business in Irasburg, VT (the "Contractor") that the contract between them originally dated as of March 1, 2019, Contract #37787, as amended to date, (the "Contract") is hereby amended as follows:

I. <u>Attachment A, Scope of Services</u>. The scope of services is hereby deleted in its entirety and replaced as set forth in the attachment to this Amendment.

<u>Taxes Due to the State</u>. Contractor certifies under the pains and penalties of perjury that, as of the date this contract amendment is signed, the Contractor is in good standing with respect to, or in full compliance with a plan to pay, any and all taxes due the State of Vermont.

<u>Child Support (Applicable to natural persons only; not applicable to corporations, partnerships or LLCs)</u>. Contractor is under no obligation to pay child support or is in good standing with respect to or in full compliance with a plan to pay any and all child support payable under a support order as of the date of this amendment.

<u>Certification Regarding Suspension or Debarment</u>. Contractor certifies under the pains and penalties of perjury that, as of the date this contract amendment is signed, neither Contractor nor Contractor's principals (officers, directors, owners, or partners) are presently debarred, suspended, proposed for debarment, declared ineligible or excluded from participation in federal programs, or programs supported in whole or in part by federal funds.

Contractor further certifies under pains and penalties of perjury that, as of the date this contract amendment is signed, Contractor is not presently debarred, suspended, nor named on the State's debarment list at: <a href="http://bgs.vermont.gov/purchasing-contracting/debarment">http://bgs.vermont.gov/purchasing-contracting/debarment</a>

This document consists of 2 pages. Except as modified by this Amendment No. 1, all provisions of the Contract remain in full force and effect.

[Remainder of Page Intentionally Left Blank]

Revision Date: 12/12/2018

The signatures of the undersigned indicate that each has read and agrees to be bound by this Amendment to the Contract.

STATE OF VERMONT		LIMELITE RESTORATION SERVICES		
By:		By:		
Name:Christop		Name:		
Commis Title: Building	sioner s and General Services	Title:		
Date:		Date:		

Revision Date: 12/12/2018

#### ATTACHMENT A – STATEMENT OF WORK

The Contractor shall provide all labor, materials and equipment necessary to satisfactorily complete Abatement and Mold Remediation services as identified below.

#### Contractor shall provide:

- 1.1 This Contract may be utilized on an emergency or as-needed basis for the abatement of building materials containing lead or asbestos, or for the remediation of mold. Projects are not to exceed \$150,000.00, including all costs associated with the individual project, such as labor, material, equipment, etc.
- 1.2 All Contractor work performed under this contract shall be planned and scheduled by Project Managers from the Agency/Department coordinating the work. The Project Manager will work closely with both Contractor and the Agency/Department requiring the work and will approve all invoices for work completed under this Contract.
  - 1.2.1 If the Contractor is selected to perform work on a project, the Agency/Department will issue a Purchase Order (PO) with a project Statement of Work and the Project Manager shall provide a PO number to the Contractor to reference on their invoice for services completed under that specific work assignment.
  - 1.2.2 The resultant Purchase Order will be administered by the Agency/Department coordinating the work.
- 1.3 Contractor shall obtain all required permits from the Vermont Department of Health (VDH) Asbestos and Lead Regulatory Program and, if applicable, from the municipal jurisdiction, prior to performing abatement under this contract.
- 1.4 Contractor shall provide notice to VDH as required by permit or by the Vermont Regulations for Asbestos Control or other applicable local, State, or federal regulations.
- 1.5 Contractor's typical working hours under this Contract will range between 6:00 AM and 5:00 PM, Monday through Friday, but occasions may arise which would require work to be performed before or after these hours, on weekends, or Federal government observed holidays. The typical working hours may vary by the type of facility or the operational needs of the Agency/Department where work is being performed and will be established at the beginning of each project.
- 1.6 Contractor shall notify the Project Manager of any maintenance related issues that are discovered while performing work.
- 1.7 All unclaimed work articles found in or about the work area by the Contractor shall be turned in immediately to the Project Manager, with the location where the article was found.
- 1.8 Contractor acknowledges that security procedures in some State buildings require a background clearance be performed on any contractor working inside the building, prior to beginning work. Projects at Correctional Facilities, Courthouses, and Public Safety buildings may all require clearances.
- 1.9 Contractor shall secure and pay for any permits and inspections required by the authorities having jurisdiction, or for warranty purposes. Contractor shall ensure that any inspections are made by the appropriate State or local authority having jurisdiction, or manufacturer from which the warranty will be issued.

1.10 Subcontractors, if required, shall be approved in writing by the Project Manager prior to performing work as part of the contract.

#### 1.11 SITE SUPERVISION:

- 1.11.1 Contractor shall provide adequate supervision of his employees to ensure complete and satisfactory performance of all work in accordance with the terms of the contract. Contractor shall have a responsible supervisor on the job at all times when the work of the contract is being carried out.
- 1.11.2 Contractor's site supervisor shall be responsible for communication with the State's representatives and shall meet with the Project Manager at the site on a weekly basis to discuss project status, including any problems, ideas, or concerns related to the project work.
- 1.11.3 Contractor and its employees shall be subject to all applicable State and Federal statutes and regulations for the conduct of personnel.
- 1.11.4 The Contractor shall provide adequate supervision of his/her subcontractors and their employees at all times.

#### 1.12 WORKMANSHIP AND MATERIALS:

- 1.12.1 Contractor shall furnish all supervision, labor, transportation, materials, tools and equipment necessary to satisfactorily complete the service in a manner consistent with the Project Manager's plan and schedule. Contractor's equipment shall be of the size and type appropriate for completing the various types of work described in this RFP. Contractor shall ensure that any equipment considered by the Project Manager to be improper or inadequate for this purpose is removed from the site and replaced with satisfactory equipment.
- 1.12.2 All work performed under this contract shall be completed in accordance with local, state, and national codes and standards, and other recognized industry standards associated with the work.
- 1.12.3 Contractor and their employees shall be certified by the VDH Asbestos and Lead Regulatory Program. For mold remediation work, Contractor shall be certified by the American Council of Accredited Certifications or equivalent certification body.
- 1.12.4 Lead or asbestos abatement work performed under this contract shall be in accordance with the regulations of the VDH Asbestos and Lead Regulatory Program, United States Occupational Health and Safety Agency (OSHA), Vermont OSHA (VOSHA), and United States Environmental Protection Agency (USEPA).
- 1.12.5 Contractor shall perform mold remediation services in accordance with a project-specific Mold Remediation Work Plan prepared for the State of Vermont by a Professional Engineer or qualified Mold Remediation Contractor. If the Project Manager determines that the project does not require a Mold Remediation Work Plan, then the Contractor shall perform mold remediation work in accordance with their subtask scope of work and recognized industry standards applicable to the work including, but not limited to, the following documents or the most recent revisions thereof:
  - 1.12.5.1 American Industrial Hygiene Association (AIHA) Recognition, Evaluation, and Control of Indoor Mold, 2008
  - 1.12.5.2 American Conference of Governmental Industrial Hygienists (ACGIH) Bioaerosols Assessment and Control, 1999
  - 1.12.5.3 Institute of Inspection Cleaning and Restoration Certification (IICRC) S520 Standard and Reference Guide for Professional Mold Remediation, published 2015
  - 1.12.5.4 United States Occupational Safety and Health Administration (OSHA) A Brief Guide to Mold in the Workplace, revised November 2013

- 1.12.5.5 United States Environmental Protection Agency (USEPA) Mold Remediation in Schools and Commercial Buildings, published March 2001
- 1.12.6 The Contractor guarantees that all materials shall be of the best quality, that all work shall be done in a professional manner, and that all aspects of the project will be delivered in good working order, complete and perfect in every respect, and that all systems and materials necessary to make the project completely operational as contemplated by the above description of the project, even if those systems and materials are not specifically described in this Contract, shall be included in the contract price.
- 1.12.7 Contractor shall ensure that all supplies, equipment and machines shall be kept free of traffic lanes or other areas that may be hazardous. Contractor shall further ensure that all dirt and debris resulting from the work under this contract shall be disposed of at the end of each day or at the completion of work in each building.
- 1.12.8 Contractor shall, at no additional cost to the State, repair furnishings, equipment, facilities or other property of the State damaged by Contractor, its officers, employees, agents, contractors, subcontractors and invitees. Contractor acknowledges that the determination of the need for, and extent of, any repair work shall be made at the sole discretion of the Project Manager.
- 1.12.9 Contractor shall, upon request of the Project Manager, provide Safety Data Sheets for all chemicals to be used during the performance of work under this contract. Safety Data Sheets shall be provided no later than 72 hours prior to the performance of abatement or remediation work under this contract, except for emergency response work where the 72-hour requirement shall be waived.
- 1.12.10 It is the contractor's responsibility to contact Dig-Safe prior to beginning any excavation work. For private complexes, such as prisons, the contractor will need to contact a private underground utility locator prior to beginning work.
- 1.13 In the event of Contractor default, the State may procure the services, materials and/or supplies from other sources and hold Contractor responsible for any excess cost occasioned thereby, provided that, if public necessity requires the use of services, materials and/or supplies not conforming to the specifications, they may be accepted and payment therefore shall be made at a proper reduction in price.

	specifications, they may be accepted and pay price.	ment therefore shall be made at a proper reduction in
1.14	Types of abatement and mold remediation se	rvices contractor shall provide:
	☐ Asbestos Abatement ☐ Lead	d Abatement x Mold Remediation
	· · · · · · · · · · · · · · · · · · ·	o-hazard restoration services. Emergency heating up to dalism cleaning as well as graffiti and odor removal
1.15	BGS Maintenance Districts contractor s	nall provide service in:
	X District 1 – Montpelier/Barre/Berlin	X District 2 – Waterbury/Hyde Park/Middlesex
	☐ District 3 - Northwest	X District 4 – Northeast
	☐ District 5 - Southwest	☐ District 6 – Southeast
	☐ District 7 - W. Springfield, MA (VT Bu	uilding at Eastern States Exposition)

### CONTRACT

## State of Vermont

Buildings and General Services Office of Purchasing & Contracting 109 State St Montpelier VT 05609-3001 United States

Supplier 0000326118 Limelite Restoration Services Inc 124 Leaf Hill Drive Irasburg VT 05485 United States

Contract ID		Page
000000000000000	00000037787	1 of 2
Contract Dates		Origin
03/01/2019 to 02	2/28/2020	CP
Description:		Contract Maximum
Docopuo		
Abatement & Mol	d Remediation	\$300,000.00
	d Remediation  Buyer Phone	\$300,000.00 Contract Status

#### Phone #:

Line # Item ID	Item Desc	UOM	Unit Price	Max Qty	Max Amt
1	Retainer - Abatement & Mold Remediation Services	JOB	0.01000	0.00	300,000.00

#### STANDARD CONTRACT FOR SERVICES

- 1. Parties. This is a contract for services between the State of Vermont, Department of Buildings and General Services (hereinafter called "State"), and LimeLite Restoration Services, Inc. with a principal place of business in Irasburg, VT (hereinafter called "Contractor"). Contractor's form of business organization is a domestic profit corporation. It is Contractor's responsibility to contact the Vermont Department of Taxes to determine if, by law, Contractor is required to have a Vermont Department of Taxes Business Account Number.
- 2. Subject Matter. The subject matter of this contract is services generally on the subject of abatement and mold remediation services. Detailed services to be provided by Contractor are described in Attachment A.
- 3. Maximum Amount. In consideration of the services to be performed by Contractor, the State agrees to pay Contractor, in accordance with the payment provisions specified in Attachment B, a sum not to exceed \$300,000.00.
- 4. Contract Term. The period of Contractor's performance shall begin on March 1, 2019 and end on February 28, 2020 with an option to renew for an additional three (3) twelve (12) month periods.
- 5. Prior Approvals. This Contract shall not be binding unless and until all requisite prior approvals have been obtained in accordance with current State law, bulletins, and interpretations.
- 6. Amendment. No changes, modifications, or amendments in the terms and conditions of this contract shall be effective unless reduced to writing, numbered and signed by the duly authorized representative of the State and Contractor.
- 7. Termination for Convenience. This contract may be terminated by the State at any time by giving written notice at least thirty (30) days in advance. In such event, Contractor shall be paid under the terms of this contract for all services provided to and accepted by the State prior to the effective date of termination.
- 8. Attachments. This contract consists of 56 pages including the following attachments which are incorporated herein:

Attachment A - Statement of Work

Attachment B - Payment Provisions

Attachment C - "Standard State Provisions for Contracts and Grants" a preprinted form (revision date 12/15/2017)

Attachment D - General Conditions for Construction Contracts

Attachment D1 - SOW - RFP Form

Attachment E - Security Specifications

Section 01900 - Security Requirements for Correctional Facilities and Background Check Form

Section 01901 - Security Requirements for Courthouses and Background Check Form

Informational Sheet for Public Safety Buildings and Background Check Form

#### Attachment F - BGS Maintenance District Map

- 9. Order of Precedence. Any ambiguity, conflict or inconsistency between the documents comprising this contract shall be resolved according to the following order of precedence:
- (1) Standard Contract
- (2) Attachment D General Conditions for Construction Contracts
- (3) Attachment C (Standard State Provisions for Contracts and Grants)
- (4) Attachment E Security Specifications
- (5) Attachment A Statement of Work
- (6) Attachment B Payment Provisions
- (7) Attachment D1 SOW RFP Form

## **CONTRACT**

## **State of Vermont**

Buildings and General Services Office of Purchasing & Contracting 109 State St Montpelier VT 05609-3001 United States



Supplier 0000326118 Limelite Restoration Services Inc 124 Leaf Hill Drive Irasburg VT 05485 United States

Contract ID Page 000000000000000000037787 2 of Origin **Contract Dates** 03/01/2019 to 02/28/2020 CP Description: **Contract Maximum** Abatement & Mold Remediation \$300,000.00 **Buyer Name** Contract Status **Buyer Phone** Erin Marie Collier Approved

	Р	hone #:				
				Unit	Max	Max
Line #	Item ID	Item Desc	UOM	Price	Qty	Amt
(8) Atta	chment F	- BGS Maintenance District Map				

WE THE UNDERSIGNED PARTIES AGREE TO BE BOUND BY THIS CONTRACT

By the STATE of VERMONT

Date:

Signature:

Name:

Title:

Email:

Email:

Email:

Email:

#### ATTACHMENT A - STATEMENT OF WORK

The Contractor shall provide all labor, materials and equipment necessary to satisfactorily complete abatement and mold remediation services as identified below.

#### Contractor shall provide:

- 1.1 This Contract may be utilized for small abatement and mold remediation service projects, emergency repairs, and/or on an as-needed basis. Projects are not to exceed \$150,000.00, including all costs associated with the individual project, such as labor, material, equipment, etc.
- 1.2 All Contractor work performed under this contract shall be planned and scheduled by Project Managers from the Agency/Department coordinating the work. The Project Manager will work closely with both Contractor and the Agency/Department requiring the work and will approve all invoices for work completed under this Contract.
  - 1.2.1 If the Contractor is selected to perform work on a project, the Agency/Department will issue a Purchase Order (PO) with a project Statement of Work and the Project Manager shall provide a PO number to the Contractor to reference on their invoice for services completed under that specific work assignment.
  - 1.2.2 The resultant PO will be administered by the Agency/Department.
- 1.3 Contractor's typical working hours under this Contract will range between 6:00 AM and 5:00 PM, Monday through Friday, but occasions may arise which would require work to be performed before or after these hours, on weekends, or on Federal government observed holidays. The typical working hours may vary by the type of facility or the operational needs of the Agency/Department where work is being performed and, if typical work hours will vary from the above, such hours will be established by the State at the beginning of each project.
- 1.4 Contractor shall notify the Project Manager of any maintenance related issues that are discovered while performing work.
- 1.5 All unclaimed work articles found in or about the work area by the Contractor shall be turned in immediately to the Project Manager, with the location where the article was found.
- 1.6 Contractor acknowledges that security procedures in some State buildings require a background clearance be performed on any contractor working inside the building, prior to beginning work. Projects at Correctional Facilities, Courthouses, and Public Safety buildings may all require clearances.
- 1.7 Contractor shall secure and pay for any permits and inspections required by the authorities having jurisdiction, or for warranty purposes. Contractor shall ensure that any inspections are made by the appropriate State or local authority having jurisdiction, or manufacturer from which the warranty will be issued.
- 1.8 Subcontractors, if required, shall be approved in writing by the Project Manager prior to performing work as part of the contract.

#### 1.9 SITE SUPERVISION:

1.9.1 Contractor shall provide adequate supervision of his employees to ensure complete and satisfactory performance of all work in accordance with the terms of the contract. Contractor shall have a responsible supervisor on the job at all times when the work of the contract is being carried out.

- 1.9.2 Contractor's site supervisor shall be responsible for communication with the State's representatives and shall meet with the Project Manager at the site on a weekly basis to discuss project status, including any problems, ideas, or concerns related to the project work.
- 1.9.3 Contractor and its employees shall be subject to all applicable State and Federal statutes and regulations for the conduct of personnel.
- 1.9.4 The Contractor shall provide adequate supervision of his/her subcontractors and their employees at all times.

#### 1.10 WORKMANSHIP AND MATERIALS:

- 1.10.1 Contractor shall furnish all supervision, labor, transportation, materials, tools and equipment necessary to satisfactorily complete the service in a manner consistent with the Project Manager's plan and schedule. Contractor's equipment shall be of the size and type appropriate for completing the various types of work described in this RFP. Contractor shall ensure that any equipment considered by the Project Manager to be improper or inadequate for this purpose is removed from the site and replaced with satisfactory equipment.
- 1.10.2 All work performed under this contract shall be completed in accordance with local, state, and national codes and standards, and other recognized industry standards associated with the work.
- 1.10.3 The Contractor guarantees that all materials shall be of the best quality, that all work shall be done in a professional manner, and that all aspects of the project will be delivered in good working order, complete and perfect in every respect, and that all systems and materials necessary to make the project completely operational as contemplated by the above description of the project, even if those systems and materials are not specifically described in this Contract, shall be included in the contract price.
- 1.10.4 Contractor shall ensure that all supplies, equipment and machines shall be kept free of traffic lanes or other areas that may be hazardous. Contractor shall further ensure that all dirt and debris resulting from the work under this contract shall be disposed of at the end of each day or at the completion of work in each building.
- 1.10.5 Contractor shall, at no additional cost to the State, repair furnishings, equipment, facilities or other property of the State damaged by Contractor, its officers, employees, agents, contractors, subcontractors and invitees. Contractor acknowledges that the determination of the need for, and extent of, any repair work shall be made at the sole discretion of the Project Manager.
- 1.11 In the event of Contractor default, the State may procure the services, materials and/or supplies from other sources and hold Contractor responsible for any excess cost occasioned thereby, provided that, if public necessity requires the use of services, materials and/or supplies not conforming to the specifications, they may be accepted, and payment therefore shall be made at a proper reduction in price.

1.12	12 Types of abatement and mold remediation services contractor shall provide:		
	☐ Asbestos Abatement	☐ Lead Abatement	X Mold Remediation
	`		oration services. Emergency heatingup to as well as graffiti and odor removal services.

1.13 BGS Maintenance Districts contractor shall provide service in:

 ${f X}$  District 1 – Montpelier/Barre/Berlin  ${f X}$  District 2 – Waterbury/Hyde Park/Middlesex

 $\square$  District 3 - Northwest X District 4 - Northeast

☐ District 5 - Southwest ☐ District 6 - Southeast

☐ District 7 - W. Springfield, MA (VT Building at Eastern States Exposition)

#### ATTACHMENT B - PAYMENT PROVISIONS

- 1. The maximum dollar amount payable under this contract is not intended as any form of a guaranteed amount. Contractor will be paid for products or services actually delivered or performed, as specified in Attachment A, up to the maximum allowable amount specified on page 1 of this contract.
- 2 Prior to commencement of work and release of any payments, Contractor shall submit to the State:
  - 2.1 a certificate of insurance consistent with the requirements set forth in Attachment C, Section 8 (Insurance), and with any additional requirements for insurance as may be set forth elsewhere in this contract; and
  - 2.2 a current IRS Form W-9 (signed within the last six months).
- 3 Payment terms are **Net 30** days from the date the State receives an error-free invoice with all necessary and complete supporting documentation.
- 4 In consideration of the services performed by Contractor, the State shall pay Contractor in accordance with the following schedule of rates. These rates are inclusive of all fees and expenses including mileage and travel time:

Job Title	Hourly Rate	Hourly Rate for off hours, weekends, holidays
Supervisor	\$65.00	\$75.00
Technician	\$55.00	\$65.00
Labor	\$40.00	\$50.00
Administrator	\$30.00	\$35.00

- 5 Contractor Material Mark-up: The State will not consider any contractor's material mark-up exceeding 10% over Contractor's actual cost.
- 6 The State will not consider any mark-up exceeding 5%, by the contractor, on any work performed by subcontractors.
- 7 The State will not consider any subcontractor's material mark-up exceeding 10% over the subcontractor's actual cost.
- 8 Contractor shall submit detailed invoices itemizing all work performed during the invoice period, including the dates of service, rates of pay, hours of work performed, and any other information and/or documentation appropriate and sufficient to substantiate the amount invoiced for payment by the State.
  - 8.1 Services may be issued on a time and materials basis or a fixed price. The following information is required on all invoices:
    - 8.1.1 All invoices must include the Contract # and numbered invoice for this contract;
    - 8.1.2 Time frame indicated of when work was performed;
    - 8.1.3 Copy of quote originally submitted;
    - 8.1.4 The agreed to markup for profit and overhead unless a previously agreed to billing schedule was approved in the contract;

- 8.1.5 Certification that the contractor has no ownership (majority or minority) in any subcontractor they claim for profit and overhead;
- 8.1.6 Provide supporting documentation of material costs, in accordance with the percentage specified in the contract. This supporting documentation is required for verification.
- 8.2 For projects billed on a Time & Materials basis, the following additional information must be included:
  - 8.2.1 Invoices shall include description of work, # of hours worked if applicable, including copies of time sheets and a certified payroll following the USDOL form (or comparable);
  - 8.2.2 Copies of <u>original</u> receipts for all materials purchased or costs incurred as a result of the scope of work;
- 9 Contractor shall submit invoices to the State in accordance with the schedule set forth in this Attachment B. Unless a more particular schedule is provided herein, invoices shall be submitted not more frequently than monthly. Invoices shall be submitted to the State Agency/Department requesting services.
- 10 Contract shall be for a period of one (1) year with an option to renew for an additional three (3) twelve (12) month periods. If renewed, on each anniversary of the Contract Start Date, the hourly rates shall be increased by the unadjusted percentage change for the twelve (12) months prior, from the Consumer Price Index (CPI) reporting data. Should the percentage change be negative, the State reserves the right to adjust the yearly contract amount accordingly.

## ATTACHMENT C: STANDARD STATE PROVISIONS FOR CONTRACTS AND GRANTS REVISED DECEMBER 15, 2017

- **1. Definitions:** For purposes of this Attachment, "Party" shall mean the Contractor, Grantee or Subrecipient, with whom the State of Vermont is executing this Agreement and consistent with the form of the Agreement. "Agreement" shall mean the specific contract or grant to which this form is attached.
- **2. Entire Agreement:** This Agreement, whether in the form of a contract, State-funded grant, or Federally-funded grant, represents the entire agreement between the parties on the subject matter. All prior agreements, representations, statements, negotiations, and understandings shall have no effect.
- **3. Governing Law, Jurisdiction and Venue; No Waiver of Jury Trial:** This Agreement will be governed by the laws of the State of Vermont. Any action or proceeding brought by either the State or the Party in connection with this Agreement shall be brought and enforced in the Superior Court of the State of Vermont, Civil Division, Washington Unit. The Party irrevocably submits to the jurisdiction of this court for any action or proceeding regarding this Agreement. The Party agrees that it must first exhaust any applicable administrative remedies with respect to any cause of action that it may have against the State with regard to its performance under this Agreement. Party agrees that the State shall not be required to submit to binding arbitration or waive its right to a jury trial.
- **4. Sovereign Immunity:** The State reserves all immunities, defenses, rights or actions arising out of the State's sovereign status or under the Eleventh Amendment to the United States Constitution. No waiver of the State's immunities, defenses, rights or actions shall be implied or otherwise deemed to exist by reason of the State's entry into this Agreement.
- **5.** No Employee Benefits For Party: The Party understands that the State will not provide any individual retirement benefits, group life insurance, group health and dental insurance, vacation or sick leave, workers compensation or other benefits or services available to State employees, nor will the State withhold any state or Federal taxes except as required under applicable tax laws, which shall be determined in advance of execution of the Agreement. The Party understands that all tax returns required by the Internal Revenue Code and the State of Vermont, including but not limited to income, withholding, sales and use, and rooms and meals, must be filed by the Party, and information as to Agreement income will be provided by the State of Vermont to the Internal Revenue Service and the Vermont Department of Taxes.
- **6. Independence:** The Party will act in an independent capacity and not as officers or employees of the State.
- **7. Defense and Indemnity:** The Party shall defend the State and its officers and employees against all third party claims or suits arising in whole or in part from any act or omission of the Party or of any agent of the Party in connection with the performance of this Agreement. The State shall notify the Party in the event of any such claim or suit, and the Party shall immediately retain counsel and otherwise provide a complete defense against the entire claim or suit. The State retains the right to participate at its own expense in the defense of any claim. The State shall have the right to approve all proposed settlements of such claims or suits.

After a final judgment or settlement, the Party may request recoupment of specific defense costs and may file suit in Washington Superior Court requesting recoupment. The Party shall be entitled to recoup costs only upon a showing that such costs were entirely unrelated to the defense of any claim arising from an act or omission of the Party in connection with the performance of this Agreement.

The Party shall indemnify the State and its officers and employees if the State, its officers or employees become legally obligated to pay any damages or losses arising from any act or omission of the Party or an agent of the Party in connection with the performance of this Agreement.

Notwithstanding any contrary language anywhere, in no event shall the terms of this Agreement or any document furnished by the Party in connection with its performance under this Agreement obligate the State to (1) defend or indemnify the Party or any third party, or (2) otherwise be liable for the expenses or reimbursement, including attorneys' fees, collection costs or other costs of the Party or any third party.

**8. Insurance:** Before commencing work on this Agreement the Party must provide certificates of insurance to show that the following minimum coverages are in effect. It is the responsibility of the Party to maintain current certificates of insurance on file with the State through the term of this Agreement. No warranty is made that the coverages and limits listed herein are adequate to cover and protect the interests of the Party for the Party's operations. These are solely minimums that have been established to protect the interests of the State.

Workers Compensation: With respect to all operations performed, the Party shall carry workers' compensation insurance in accordance with the laws of the State of Vermont. Vermont will accept an out-of-state employer's workers' compensation coverage while operating in Vermont provided that the insurance carrier is licensed to write insurance in Vermont and an amendatory endorsement is added to the policy adding Vermont for coverage purposes. Otherwise, the party shall secure a Vermont workers' compensation policy, if necessary to comply with Vermont law.

General Liability and Property Damage: With respect to all operations performed under this Agreement, the Party shall carry general liability insurance having all major divisions of coverage including, but not limited to:

Premises - Operations

**Products and Completed Operations** 

Personal Injury Liability

**Contractual Liability** 

The policy shall be on an occurrence form and limits shall not be less than:

\$1,000,000 Each Occurrence

\$2,000,000 General Aggregate

\$1,000,000 Products/Completed Operations Aggregate

\$1,000,000 Personal & Advertising Injury

Automotive Liability: The Party shall carry automotive liability insurance covering all motor vehicles, including hired and non-owned coverage, used in connection with the Agreement. Limits of coverage shall not be less than \$500,000 combined single limit. If performance of this Agreement involves construction, or the transport of persons or hazardous materials, limits of coverage shall not be less than \$1,000,000 combined single limit.

Additional Insured. The General Liability and Property Damage coverages required for performance of this Agreement shall include the State of Vermont and its agencies, departments, officers and employees as Additional Insureds. If performance of this Agreement involves construction, or the transport of persons or hazardous materials, then the required Automotive Liability coverage shall include the State of Vermont and its agencies, departments, officers and employees as Additional Insureds. Coverage shall be primary and non-contributory with any other insurance and self-insurance.

*Notice of Cancellation or Change*. There shall be no cancellation, change, potential exhaustion of aggregate limits or non-renewal of insurance coverage(s) without thirty (30) days written prior written notice to the State.

- **9. Reliance by the State on Representations:** All payments by the State under this Agreement will be made in reliance upon the accuracy of all representations made by the Party in accordance with this Agreement, including but not limited to bills, invoices, progress reports and other proofs of work.
- **10. False Claims Act:** The Party acknowledges that it is subject to the Vermont False Claims Act as set forth in 32 V.S.A. § 630 *et seq.* If the Party violates the Vermont False Claims Act it shall be liable to the State for civil penalties, treble damages and the costs of the investigation and prosecution of such violation, including attorney's fees, except as the same may be reduced by a court of competent jurisdiction. The Party's liability to the State under the False Claims Act shall not be limited notwithstanding any agreement of the State to otherwise limit Party's liability.
- 11. Whistleblower Protections: The Party shall not discriminate or retaliate against one of its employees or agents for disclosing information concerning a violation of law, fraud, waste, abuse of authority or acts threatening health or safety, including but not limited to allegations concerning the False Claims Act. Further, the Party shall not require such employees or agents to forego monetary awards as a result of such disclosures, nor should they be required to report misconduct to the Party or its agents prior to reporting to any governmental entity and/or the public.
- **12.** Location of State Data: No State data received, obtained, or generated by the Party in connection with performance under this Agreement shall be processed, transmitted, stored, or transferred by any means outside the continental United States, except with the express written permission of the State.
- 13. Records Available for Audit: The Party shall maintain all records pertaining to performance under this agreement. "Records" means any written or recorded information, regardless of physical form or characteristics, which is produced or acquired by the Party in the performance of this agreement. Records produced or acquired in a machine readable electronic format shall be maintained in that format. The records described shall be made available at reasonable times during the period of the Agreement and for three years thereafter or for any period required by law for inspection by any authorized representatives of the State or Federal Government. If any litigation, claim, or audit is started before the expiration of the three-year period, the records shall be retained until all litigation, claims or audit findings involving the records have been resolved.
- **14. Fair Employment Practices and Americans with Disabilities Act:** Party agrees to comply with the requirement of 21 V.S.A. Chapter 5, Subchapter 6, relating to fair employment practices, to the full extent applicable. Party shall also ensure, to the full extent required by the Americans with Disabilities Act of 1990, as amended, that qualified individuals with disabilities receive equitable access to the services, programs, and activities provided by the Party under this Agreement.
- **15. Set Off:** The State may set off any sums which the Party owes the State against any sums due the Party under this Agreement; provided, however, that any set off of amounts due the State of Vermont as taxes shall be in accordance with the procedures more specifically provided hereinafter.

#### 16. Taxes Due to the State:

- **A.** Party understands and acknowledges responsibility, if applicable, for compliance with State tax laws, including income tax withholding for employees performing services within the State, payment of use tax on property used within the State, corporate and/or personal income tax on income earned within the State.
- **B.** Party certifies under the pains and penalties of perjury that, as of the date this Agreement is signed, the Party is in good standing with respect to, or in full compliance with, a plan to pay any and all taxes due the State of Vermont.

- C. Party understands that final payment under this Agreement may be withheld if the Commissioner of Taxes determines that the Party is not in good standing with respect to or in full compliance with a plan to pay any and all taxes due to the State of Vermont.
- **D.** Party also understands the State may set off taxes (and related penalties, interest and fees) due to the State of Vermont, but only if the Party has failed to make an appeal within the time allowed by law, or an appeal has been taken and finally determined and the Party has no further legal recourse to contest the amounts due.
- **17. Taxation of Purchases:** All State purchases must be invoiced tax free. An exemption certificate will be furnished upon request with respect to otherwise taxable items.
- **18. Child Support:** (Only applicable if the Party is a natural person, not a corporation or partnership.) Party states that, as of the date this Agreement is signed, he/she:
  - A. is not under any obligation to pay child support; or
  - **B.** is under such an obligation and is in good standing with respect to that obligation; or
  - **C.** has agreed to a payment plan with the Vermont Office of Child Support Services and is in full compliance with that plan.

Party makes this statement with regard to support owed to any and all children residing in Vermont. In addition, if the Party is a resident of Vermont, Party makes this statement with regard to support owed to any and all children residing in any other state or territory of the United States.

**19. Sub-Agreements:** Party shall not assign, subcontract or subgrant the performance of this Agreement or any portion thereof to any other Party without the prior written approval of the State. Party shall be responsible and liable to the State for all acts or omissions of subcontractors and any other person performing work under this Agreement pursuant to an agreement with Party or any subcontractor.

In the case this Agreement is a contract with a total cost in excess of \$250,000, the Party shall provide to the State a list of all proposed subcontractors and subcontractors' subcontractors, together with the identity of those subcontractors' workers compensation insurance providers, and additional required or requested information, as applicable, in accordance with Section 32 of The Vermont Recovery and Reinvestment Act of 2009 (Act No. 54).

Party shall include the following provisions of this Attachment C in all subcontracts for work performed solely for the State of Vermont and subcontracts for work performed in the State of Vermont: Section 10 ("False Claims Act"); Section 11 ("Whistleblower Protections"); Section 12 ("Location of State Data"); Section 14 ("Fair Employment Practices and Americans with Disabilities Act"); Section 16 ("Taxes Due the State"); Section 18 ("Child Support"); Section 20 ("No Gifts or Gratuities"); Section 22 ("Certification Regarding Debarment"); Section 30 ("State Facilities"); and Section 32.A ("Certification Regarding Use of State Funds").

- **20.** No Gifts or Gratuities: Party shall not give title or possession of anything of substantial value (including property, currency, travel and/or education programs) to any officer or employee of the State during the term of this Agreement.
- **21. Copies:** Party shall use reasonable best efforts to ensure that all written reports prepared under this Agreement are printed using both sides of the paper.
- **22. Certification Regarding Debarment:** Party certifies under pains and penalties of perjury that, as of the date that this Agreement is signed, neither Party nor Party's principals (officers, directors, owners, or partners) are presently debarred, suspended, proposed for debarment, declared ineligible or excluded from participation in Federal programs, or programs supported in whole or in part by Federal funds.

Party further certifies under pains and penalties of perjury that, as of the date that this Agreement is signed, Party is not presently debarred, suspended, nor named on the State's debarment list at: http://bgs.vermont.gov/purchasing/debarment

- **23.** Conflict of Interest: Party shall fully disclose, in writing, any conflicts of interest or potential conflicts of interest.
- **24. Confidentiality:** Party acknowledges and agrees that this Agreement and any and all information obtained by the State from the Party in connection with this Agreement are subject to the State of Vermont Access to Public Records Act, 1 V.S.A. § 315 et seq.
- **25. Force Majeure:** Neither the State nor the Party shall be liable to the other for any failure or delay of performance of any obligations under this Agreement to the extent such failure or delay shall have been wholly or principally caused by acts or events beyond its reasonable control rendering performance illegal or impossible (excluding strikes or lock-outs) ("Force Majeure"). Where Force Majeure is asserted, the nonperforming party must prove that it made all reasonable efforts to remove, eliminate or minimize such cause of delay or damages, diligently pursued performance of its obligations under this Agreement, substantially fulfilled all non-excused obligations, and timely notified the other party of the likelihood or actual occurrence of an event described in this paragraph.
- **26. Marketing:** Party shall not refer to the State in any publicity materials, information pamphlets, press releases, research reports, advertising, sales promotions, trade shows, or marketing materials or similar communications to third parties except with the prior written consent of the State.

#### 27. Termination:

- A. Non-Appropriation: If this Agreement extends into more than one fiscal year of the State (July 1 to June 30), and if appropriations are insufficient to support this Agreement, the State may cancel at the end of the fiscal year, or otherwise upon the expiration of existing appropriation authority. In the case that this Agreement is a Grant that is funded in whole or in part by Federal funds, and in the event Federal funds become unavailable or reduced, the State may suspend or cancel this Grant immediately, and the State shall have no obligation to pay Subrecipient from State revenues.
- **B.** Termination for Cause: Either party may terminate this Agreement if a party materially breaches its obligations under this Agreement, and such breach is not cured within thirty (30) days after delivery of the non-breaching party's notice or such longer time as the non-breaching party may specify in the notice.
- C. Termination Assistance: Upon nearing the end of the final term or termination of this Agreement, without respect to cause, the Party shall take all reasonable and prudent measures to facilitate any transition required by the State. All State property, tangible and intangible, shall be returned to the State upon demand at no additional cost to the State in a format acceptable to the State.
- **28.** Continuity of Performance: In the event of a dispute between the Party and the State, each party will continue to perform its obligations under this Agreement during the resolution of the dispute until this Agreement is terminated in accordance with its terms.
- **29. No Implied Waiver of Remedies:** Either party's delay or failure to exercise any right, power or remedy under this Agreement shall not impair any such right, power or remedy, or be construed as a waiver of any such right, power or remedy. All waivers must be in writing.
- **30. State Facilities:** If the State makes space available to the Party in any State facility during the term of this Agreement for purposes of the Party's performance under this Agreement, the Party shall only use the space in accordance with all policies and procedures governing access to and use of State facilities which

shall be made available upon request. State facilities will be made available to Party on an "AS IS, WHERE IS" basis, with no warranties whatsoever.

- **31. Requirements Pertaining Only to Federal Grants and Subrecipient Agreements:** If this Agreement is a grant that is funded in whole or in part by Federal funds:
  - **A.** Requirement to Have a Single Audit: The Subrecipient will complete the Subrecipient Annual Report annually within 45 days after its fiscal year end, informing the State of Vermont whether or not a Single Audit is required for the prior fiscal year. If a Single Audit is required, the Subrecipient will submit a copy of the audit report to the granting Party within 9 months. If a single audit is not required, only the Subrecipient Annual Report is required.
    - For fiscal years ending before December 25, 2015, a Single Audit is required if the subrecipient expends \$500,000 or more in Federal assistance during its fiscal year and must be conducted in accordance with OMB Circular A-133. For fiscal years ending on or after December 25, 2015, a Single Audit is required if the subrecipient expends \$750,000 or more in Federal assistance during its fiscal year and must be conducted in accordance with 2 CFR Chapter I, Chapter II, Part 200, Subpart F. The Subrecipient Annual Report is required to be submitted within 45 days, whether or not a Single Audit is required.
  - **B. Internal Controls:** In accordance with 2 CFR Part II, §200.303, the Party must establish and maintain effective internal control over the Federal award to provide reasonable assurance that the Party is managing the Federal award in compliance with Federal statutes, regulations, and the terms and conditions of the award. These internal controls should be in compliance with guidance in "Standards for Internal Control in the Federal Government" issued by the Comptroller General of the United States and the "Internal Control Integrated Framework", issued by the Committee of Sponsoring Organizations of the Treadway Commission (COSO).
  - C. Mandatory Disclosures: In accordance with 2 CFR Part II, §200.113, Party must disclose, in a timely manner, in writing to the State, all violations of Federal criminal law involving fraud, bribery, or gratuity violations potentially affecting the Federal award. Failure to make required disclosures may result in the imposition of sanctions which may include disallowance of costs incurred, withholding of payments, termination of the Agreement, suspension/debarment, etc.

## 32. Requirements Pertaining Only to State-Funded Grants:

- **A.** Certification Regarding Use of State Funds: If Party is an employer and this Agreement is a State-funded grant in excess of \$1,001, Party certifies that none of these State funds will be used to interfere with or restrain the exercise of Party's employee's rights with respect to unionization.
- **B.** Good Standing Certification (Act 154 of 2016): If this Agreement is a State-funded grant, Party hereby represents: (i) that it has signed and provided to the State the form prescribed by the Secretary of Administration for purposes of certifying that it is in good standing (as provided in Section 13(a)(2) of Act 154) with the Agency of Natural Resources and the Agency of Agriculture, Food and Markets, or otherwise explaining the circumstances surrounding the inability to so certify, and (ii) that it will comply with the requirements stated therein.

(End of Standard Provisions)

#### ATTACHMENT D

# STATE OF VERMONT STANDARD FORM

# GENERAL CONDITIONS FOR CONSTRUCTION CONTRACTS (January 5, 2018)

The following general conditions are for use with all construction contracts with the State of Vermont, administered through the Department of Buildings and General Services.

These general conditions will be attached to, and become part of, the Contract Documents.

If there is a conflict between the terms and conditions stated in these general conditions and the terms and conditions contained in the State-Contractor Agreement; the terms of the State-Contractor Agreement control.

## **ARTICLE 1**

#### CONTRACT DOCUMENTS

#### 1.1 **DEFINITIONS**

The term "State-Contractor Agreement" refers to the specific agreement between the State of Vermont and a particular contractor whereby the State and the contractor identify, explain and agree to the specific contractual obligations of each party and identify the nature and scope of a particular project. The "State-Contractor Agreement" will utilize one of the following construction delivery methods: (i) A general contractor hired by the State to provide all services necessary for the construction of the particular project. The general contractor is responsible for completing the particular project for a fixed price; (ii) a construction manager hired by the State to provide professional services and act as a consultant to the State in the design development and construction phases and to deliver the particular project for a Guaranteed Maximum Price; or (iii) a design-build contractor hired by the State to provide both design and construction services, under a single contract, for a particular project.

## 1.1.1 THE CONTRACT DOCUMENTS

The Contract Documents consist of the State-Contractor Agreement, the Conditions of the Contract (General, Supplementary and other Conditions), the Drawings, the Specifications, and all Addenda issued prior to bid opening and any Change Orders after execution of the Contract.-

#### 1.1.2 THE CONTRACT

The Contract Documents form the contract for construction. This contract represents the entire and integrated agreement between the parties and supersedes all prior negotiations, representations, or agreements, either written or oral. The contract may only be amended by Change Order. The Contract Documents shall not be construed to create any contractual relationship of any kind between the Architect and the Contractor, but the Architect shall be entitled to performance of obligations intended for his benefit, and to enforcement thereof. Nothing contained in the Contract Documents shall create any contractual relationship between the State or the Architect and any Subcontractor or Sub-subcontractor.

#### **1.1.3** THE WORK

The Work comprises the completed construction required by the Contract Documents and includes all labor necessary to produce such construction, and all materials and equipment incorporated or to be incorporated in such construction.

#### 1.1.4 THE PROJECT

The Project is the total construction of which the Work performed under the Contract Documents may be the whole or a part.

#### 1.1.5 PROJECT MANAGER

The term "Project Manager" refers to the employee of the State who has been assigned responsibility for overseeing and managing the proper and timely implementation of the project.

### 1.2 EXECUTION & INTENT

- **1.2.1** Upon request, the State-Contractor Agreement will be signed in duplicate by the State and Contractor.
- **1.2.2** By executing the Contract, the Contractor represents that he or she has visited and examined the site; is fully informed as to the extent and character of the Project; familiarized himself or herself with the local conditions under which the Work is to be performed, and correlated his / her observations with the requirements of the Contract Documents and understands that no consideration will be given for any alleged misunderstanding of same.
- **1.2.3** The intent of the Contract Documents is to include all items necessary for the proper execution and completion of the Work. The Contract Documents are complementary, and what is required by any one shall be as binding as, if required by all. Work not covered in the Contract Documents will not be required unless it is consistent with the Contract Documents and is reasonably inferable from the Contract Documents as being necessary to produce the intended results. Words and abbreviations which have well-known technical or trade meanings are used in the Contract Documents in accordance with such recognized meanings.
- 1.2.4 The organization of the Specifications into divisions, sections and articles, and the arrangement of Drawings shall not control the Contractor in dividing the Work among

Subcontractors or in establishing the extent of Work to be performed by any trade.

#### 1.3 OWNERSHIP AND USE OF DOCUMENTS

**1.3.1** All drawings, specifications, estimates, and all other documents, including shop drawings and calculations, prepared at any time in connection with the Project, shall, upon payment for services in connection therewith, become the sole property of the State.

## **ARTICLE 2**

#### **ARCHITECT**

#### 2.1 **DEFINITION**

The term Architect, as used in this document, means a person, or entity, lawfully licensed to practice architecture or a person or entity lawfully licensed to provide professional engineering services and has been identified as such in the State-Contractor Agreement. The term Architect means the Architect or his / her authorized representative. These General Conditions are applicable whether or not an Architect is utilized for a particular project. In the event that an Architect has not been utilized for the project, the Project Manager will be responsible for administration of the contract as specifically enumerated in paragraph 2.2. All other functions, duties, or obligations of the Architect, under these General Conditions, in the absence of an Architect, will be performed by the Project Manager.

#### 2.2 ADMINISTRATION OF THE CONTRACT

- **2.2.1** The Architect will provide administration of the Contract as hereinafter described.
- **2.2.2** The Architect will be the State's representative during construction and until final payment is due. The Architect will advise and consult with the State. The State's instructions to the Contractor shall be forwarded through the Architect. The Architect will have authority to act on behalf of the State only to the extent provided in the Contract Documents, unless otherwise modified by written instrument in accordance with Subparagraph 2.2.10.
- **2.2.3** The Architect will visit the site at intervals appropriate to the stage of construction to familiarize himself or herself generally with the progress and quality of the Work and to determine in general if the Work is proceeding in accordance with the Contract Documents. However, the Architect will not be required to make exhaustive or continuous on-site inspections to check the quality or quantity of the Work. On the basis of his or her on-site observations as an architect, he or she will keep the State informed of the progress of the Work, and will endeavor to guard the State against defects and deficiencies in the Work of the Contractor.
- **2.2.4** The Architect will not be responsible for and will not have control or charge of construction means, methods, techniques, sequences or procedures, or for safety precautions and programs in connection with the Work, and he or she will not be responsible for the Contractor's failure to carry out the Work in accordance with the Contract Documents. The Architect will not be responsible for or have control or charge over the acts or omissions of the Contractor,

Subcontractors, or any of their agents or employees, or any other persons performing any of the Work.

- **2.2.5** The Architect shall at all times have access to the Work wherever it is in preparation and progress. The Contractor shall provide facilities for such access so the Architect may perform his or her functions under the Contract Documents.
- **2.2.6** Based on the Architect's observations of the Work and an evaluation of the Contractor's Applications for Payment, the Architect will determine the amounts owing to the Contractor and will issue Certificates for Payment in such amounts, as provided in Paragraph 9.4 of these General Conditions.
- **2.2.7** The Architect will review and approve or take other appropriate action upon Contractor's submittals such as Shop Drawings, Product Data and Samples, but only for conformance with the design concept of the Work and with the information given in the Contract Documents. Such action shall be taken with reasonable promptness so as to cause no delay.
- **2.2.8** The Architect will prepare Change Orders in accordance with Article 12. The Architect has authority to order minor changes in the Work with the consent of the owner.
- **2.2.9** The Architect will conduct inspections to determine the dates of Substantial Completion and final completion, will receive and forward to the State for the State's review written warranties and related documents required by the Contract and assembled by the Contractor, and will issue a final Certificate for Payment upon compliance with the requirements of Paragraph 9.9.
- **2.2.10** The duties, responsibilities and limitations of authority of the Architect as the State's representative during construction as set forth in the Contract Documents will not be modified or extended without written consent of the State and the Architect. In the event that the duties, responsibilities and limitations of authority of the Architect as State's representative during construction as set forth in the Contract Documents is modified or extended, the State will immediately notify the Contractor, in writing, of the modification or extension and the date that the modified or extended authority is effective. If the change of duties, responsibilities, or limitations of authority of the Architect, causes the Contractor to incur new, additional, and unexpected expenses completing the Work, then the State and the Contractor shall agree to an equitable adjustment to be implemented by Change Order to the State-Contractor Agreement.
- **2.2.11** In case of the termination of the employment of Architect, the State may appoint a subsequent architect in which case, the subsequent architect will have the same status under the Contract Documents as that of the former architect. In the event of termination of the Architect, the State will immediately notify the Contractor, in writing, of the termination and the appointment of a replacement architect, if any.
- 2.2.12 The Architect will be the interpreter of the requirements of the Contract Documents and will make recommendations to the State regarding the progress and quality of the Work and whether the Work is proceeding in accordance with the Contract Documents.
- 2.2.13 The Architect will render interpretations necessary for the proper execution and progress of the Work, with reasonable promptness and in accordance with any time limit agreed

upon, in writing, by the Architect, Contractor, and the State.

2.2.14 Claims, disputes and other matters in question between the Contractor and the State relating to the execution or progress of the Work or the interpretation of the Contract Documents shall be referred initially to the Architect for review. The Architect will review the nature of the claim, dispute or other matter, along with any documentation provided by the Contractor or the State, and within a reasonable time, the Architect will provide a written recommendation to the State.

## **ARTICLE 3**

#### **STATE**

## 3.1 **DEFINITION**

The State of Vermont, by and through the Department of Buildings and General Services is the entity identified as "State" in the State-Contractor Agreement and is referred to throughout the Contract Documents as the "State".

#### 3.2 RESPONSIBILITIES OF THE STATE

- 3.2.1 The State will identify a Project Manager for the Project. The Project Manager will be the sole point of contact between the Contractor and the State. The Project Manager will be the State's representative authorized to communicate the State's position and directions related to all contract work and to coordinate all change orders with the Commissioner of Buildings and General Services as deemed necessary.
- **3.2.2** If applicable, the State shall furnish all surveys describing the physical characteristics, legal limitations and utility locations for the site of the Project, and a legal description of the site.
- **3.2.3** Except as provided in Subparagraph 4.7.1, the State shall secure and pay for necessary approvals, easements, assessments and charges required for the construction, use or occupancy of permanent structures or for permanent changes in existing facilities.
- **3.2.4** Information or services under the State's control shall be furnished by the State with reasonable promptness to avoid delay in the orderly progress of the Work.
- **3.2.5** Unless otherwise provided in the Contract Documents, the Contractor will be furnished, free of charge, all copies of Drawings and Specifications reasonably necessary for the execution of the Work.
- **3.2.6** The State shall forward instructions to the Contractor through the Architect.
- **3.2.7** The foregoing are in addition to any other duties and responsibilities of the State enumerated in the Contract Documents including those responsibilities described in Articles 6, 9 and 11 of this agreement.

**3.2.8** The State may utilize a Clerk-of-the-Works to provide inspection and monitoring work on behalf of the State. If a Clerk-of-the-Works is utilized, it is the responsibility of the Architect and the Contractor to review and understand the Clerk's duties, responsibilities and limitations of authority. Upon request, the State shall provide a copy of the Clerk-of-the-Works contract to Architect and Contractor.

#### 3.3 STATE'S RIGHT TO STOP THE WORK

**3.3.1** If the Contractor fails to correct defective Work as required by Paragraph 13.2 or persistently fails to carry out the Work in accordance with the Contract Documents, the State, by a written order signed personally or by an agent specifically so empowered by the State in writing, may order the Contractor to stop the Work, or any portion thereof, until the cause for such order has been eliminated; however, this right of the State to the stop the Work shall not give rise to any duty on the part the State to exercise this right for the benefit of the Contractor or any other person or entity, except to the extent required by Subparagraph 6.1.3.

#### 3.4 STATE'S RIGHT TO CARRY OUT THE WORK

**3.4.1** If the Contractor defaults or neglects to carry out the Work in accordance with the Contract Documents and fails within seven days after receipt of written notice from the State to commence and continue correction of such default or neglect with diligence and promptness, the State may, after seven days following receipt by the Contractor of an additional written notice and without prejudice to any other remedy he may have, make good such deficiencies. In such case an appropriate Change Order shall be issued deducting from the payments then or thereafter due the Contractor the cost of correcting such deficiencies, including compensation for the Architect's and the State's additional services made necessary by such default, neglect or failure. Such action by the State and the amount charged to the Contractor are both subject to review by the Architect under Paragraph 2.2.14 of these General Conditions. If the payments then or thereafter due the Contractor are not sufficient to cover the amounts paid by the State to make good such deficiencies, then the Contractor shall pay the difference to the Owner.

#### **ARTICLE 4**

### **CONTRACTOR**

### 4.1 **DEFINITION**

The Contractor is the person or entity identified as such in the State Contractor Agreement and is referred to throughout the Contract Documents as Contractor. The term Contractor means the Contractor or his authorized representative.

## 4.2 REVIEW OF CONTRACT DOCUMENTS

**4.2.1** The Contractor shall carefully study and compare the Contract Documents and shall immediately report in writing to the Architect and the State any error, inconsistency or omission he may discover. The Contractor shall not be liable to the State or the Architect for any damage resulting from any such errors, inconsistencies or omissions in the Contract Documents. The

Contractor shall perform no portion of the Work at any time without Contract Documents or, where required, approved Shop Drawings, Product Data or Samples for such portion of the Work.

## 4.3 SUPERVISION AND CONSTRUCTION PROCEDURES

- **4.3.1** The Contractor shall supervise and direct the Work, using his / her best skill and attention. The Contractor shall be solely responsible for all construction means, methods, techniques, sequences and procedures and for coordinating all portions of the Work under the Contract. All aspects of the Project shall be subject to the inspection and approval of the State. Contractor guarantees to repair, replace, re-execute or otherwise correct any defect in workmanship, materials, or the like that fails to conform to the requirements of this Contract or that appears during the progress of the Work or within one year of final acceptance by the State.
- **4.3.2** The Contractor shall be responsible to the State for the acts and omissions of his / her employees, Subcontractors and their agents and employees, and other persons performing any of the Work under a contract with the Contractor.
- **4.3.3** The Contractor shall not be relieved from his / her obligations to perform the Work in accordance with the Contract Documents either by the activities or duties of the Architect in his / her administration of the Contract, or by inspections, tests or approvals required or performed under Paragraph 7.7 by persons other than the Contractor.
- 4.3.4 The Contractor shall acquaint himself (herself/itself) with the limits of the property or right-of-way of the State and shall not trespass on other property. The Contractor shall adequately protect the project, adjacent property and the public, and shall be responsible for any damage or injury due to the Contractor's act or neglect, and shall save the State harmless in respect thereto.
- 4.3.5 All work shall be done in such a manner as not to interfere with the State's operating functions. Contractor and his employees shall familiarize themselves and comply with all rules and regulations applicable to the project.
- 4.3.6 The Contractor shall keep the premises free from liens arising out of or from the Project. Contractor shall obtain and submit waivers of liens with a request for a progress or final payment.

## 4.4 LABOR AND MATERIALS

**4.4.1** Unless otherwise provided in the Contract Documents, the Contractor shall provide and pay for all labor, materials, equipment, tools, construction equipment and machinery, water, heat, utilities, transportation, and other facilities and services necessary for the proper execution and completion of the Work, whether temporary or permanent and whether or not incorporated or to be incorporated in the Work. Contractor guarantees that materials shall be of the best quality, that work shall be completed in a neat and workmanlike manner, that equipment will be installed in a first class manner, and that all aspects of the project will be delivered in good working order complete and perfect in every respect and that all systems and materials necessary to make the project a complete operating utility as contemplated by the above description of the project is included in the Contract price.

**4.4.2** The Contractor shall a t all times enforce strict discipline and good order among his employees and shall not employ on the Work any unfit person or anyone not skilled in the task assigned to him.

#### 4.5 WARRANTY

**4.5.1** The Contractor warrants to the State and the Architect that all materials and equipment furnished under this Contract will be new unless otherwise specified, and that all Work will be of good quality, free from faults and defects and in conformance with the Contract Documents. All Work not conforming to these requirements, including substitutions not properly approved and authorized, may be considered defective. If requested by the Architect or the State, the Contractor shall furnish satisfactory evidence as to the kind and quality of materials and equipment. This warranty is not limited by the provisions of Paragraph 13.2 of these General Conditions.

#### 4.6 TAXES

The State is exempt from all sales and federal excise taxes. Contractors will be responsible for the payment of any sales, consumer, use and other similar taxes for the Work or portions thereof provided by the Contractor which are legally enacted at the time bids are received, whether or not yet effective.

## 4.7 PERMITS, FEES AND NOTICES

- **4.7.1** Unless otherwise provided in the Contract Documents, the Contractor shall secure and pay for all permits and governmental fees, licenses and inspections necessary for the proper execution and completion of the Work which are customarily secured after execution of the Contract and which are legally required at the time the bids are received. Contractor shall comply with the regulations and requirements of any insurance company which issues a policy on any part of the work or site.
- **4.7.2** The Contractor shall give all notices and comply with all laws, ordinances, rules, regulations and lawful orders of any public authority bearing on the performance of the Work.
- **4.7.3** It is not the responsibility of the Contractor to make certain that the Contract Documents are in accordance with applicable laws, statutes, building codes and regulations. However, if the Contractor observes that any of the Contract Documents are at variance therewith in any respect, he / she shall promptly notify the Architect in writing, and any necessary changes shall be accomplished by appropriate modification.
- **4.7.4** If the Contractor performs any Work knowing it to be contrary to such laws, ordinances, rules and regulations, and without such notice to the Architect, he / she shall assume full responsibility therefor and shall bear all costs attributable thereto.
- 4.7.5 The Contractor agrees to comply with all the requirements of Title 21 V.S.A., Chapter 5, Subchapter 6 relating to fair employment practices and agrees further to include a similar provision in any and all subcontracts. A link to 21 V.S.A. Chapter 5, Subchapter 6 is provided: http://legislature.vermont.gov/statutes/section/21/005/00495

#### 4.8 ALLOWANCES

**4.8.1** The Contractor shall include in the Contract Sum all allowances stated in the Contract Documents. Items covered by these allowances shall be supplied for such amounts and by such persons as dictated by the process contained in the Contract Documents.

## **4.8.2** Unless otherwise provided in the Contract Documents:

- (a) these allowances shall cover the cost to the Contractor, less any applicable trade discount, of the materials and equipment required by the allowance delivered at the site, and all applicable taxes;
- (b) the Contractor's costs for unloading an d handling on the site, labor, installation costs, overhead, prof it and other expenses contemplated for the original allowance shall be included in the Contract Sum and not in the allowance:
- (c) whenever the cost is more than or less than the allowance, the Contract Sum shall be adjusted accordingly by Change Or der, the a mount of which will recognize changes, if any, in handling costs on site, labor, installation costs, overhead, profit and other expenses.

## 4.9 SUPERINTENDENT

The Contractor shall employ a competent superintendent and necessary assistants who shall be in attendance at the Project site during the progress of the Work. The superintendent shall represent the Contractor and all communications given to the superintendent shall be as binding as if given to the Contractor. All Communications shall be in writing when made or shall be confirmed in writing, by the Contractor, within twenty-four (24) hours of the communication.

#### 4.10 PROGRESS SCHEDULE

The Contractor, within ten (10) working days of receiving notice of the award of the contract, shall prepare and submit for the State's and Architect's information an estimated progress schedule for the Work. The progress schedule shall be related to the entire Project to the extent required by the Contract Documents, and shall provide for expeditious and practicable execution of the Work.

#### 4.11 DOCUMENTS AND SAMPLES AT THE SITE

**4.11.1** The Contractor shall maintain at the site for the State one record copy of all Drawings, Specifications, Addenda, Change Orders and other modifications, in good order and marked currently to record all changes made during construction, and approved Shop Drawings, Product Data and Samples. These shall be available to the Architect and the State for monthly review prior to approval of Contractor's monthly application for payment. Prior to substantial completion of the Work, the Contractor shall ensure that one record copy of all Drawings, Specifications, Addenda, Change Orders and other Modifications, in good order and marked currently to record all changes made during construction, and approved Shop Drawings, Product Data and Samples are delivered to the Architect for the State and a duplicate copy shall be delivered to the State for the State's use.

## 4.12 SHOP DRAWINGS, PRODUCT DATA AND SAMPLES

- **4.12.1** Shop Drawings are drawings, diagrams, schedules and other data specially prepared for the Work by the Contractor or any Subcontractor, manufacturer, supplier or distributor to illustrate some portion of the Work.
- **4.12.2** Product Data are illustrations, standard schedules, performance charts, instructions, brochures, diagrams and other information furnished by the Contractor to illustrate a material, product or system for some portion of the Work.
- **4.12.3** Samples are physical examples which illustrate materials, equipment or workmanship and establish standards by which the Work will be judged.
- **4.12.4** The Contractor shall review, approve and submit, with reasonable promptness and in such sequence as to cause no delay in the Work or in the work of the State or any separate contractor, all Shop Drawings, Product Data and Samples required by the Contract Documents.
- **4.12.5** By approving and submitting Shop Drawings, Product Data and Samples, the Contractor represents that he / she has determined and verified all materials, field measurements, and field construction criteria related thereto, or will do so, and that he / she has checked and coordinated the information contained within such submittals with the requirements of the Work and of the Contract Documents.
- **4.12.6** The Contractor shall not be relieved of responsibility for any deviation from the requirements of the Contract Documents by the Architect's approval of Shop Drawings, Product Data or Samples under Subparagraph 2.2.7 of these General Conditions unless the Contractor has specifically informed the Architect and the State in writing of such deviation at the time of submission and the Architect and the State has given written approval to the specific deviation. The Contractor shall not be relieved from responsibility for errors or omissions in the Shop Drawings, Product Data or Samples by the Architect's approval thereof.
- **4.12.7** The Contractor shall direct specific attention, in writing or on resubmitted Shop Drawings, Product Data or Samples, to revisions other than those requested by the Architect on previous submittals.
- **4.12.8** No portion of the Work requiring submission of a Shop Drawing, Product Data or Sample shall be commenced until the submittal has been approved by the Architect as provided in Subparagraph 2.2.7 of these General Conditions. All such portions of the Work shall be in accordance with approved submittals.

#### 4.13 USE OF SITE

**4.13.1** The Contractor shall confine operations at the site to areas permitted by law, including areas designated for operations or prohibited from operations by any applicable permit and or regulation, ordinances, permits and the Contract Documents and shall not unreasonably encumber the site with any materials or equipment.

## 4.14 CUTTING AND PATCHING OF WORK

- **4.14.1** The Contractor shall be responsible for all cutting, fitting or patching that may be required to complete the Work or to make its several parts fit together properly.
- **4.14.2** The Contractor shall not damage or endanger any portion of the Work or the work of the State or any separate contractors by cutting, patching or otherwise altering any work, or by excavation. The Contractor shall not cut or otherwise alter the work of the State or any separate contractor except with the written consent of the State and of such separate contractor. The Contractor shall not unreasonably withhold from the State or any separate contractor his consent to cutting or otherwise altering the Work.

#### 4.15 CLEANING UP

- **4.15.1** Before commencement of any on-site activities, Contractor is required to prepare and submit to the Project Manager a Construction Site Waste Reduction Plan. A sample plan is available at: <a href="http://www.anr.state.vt.us/dec/wastediv/recycling/CandD.htm">http://www.anr.state.vt.us/dec/wastediv/recycling/CandD.htm</a>. Failure to comply with this provision or a failure to comply with the plan itself will result in withholding of general conditions' money from the contractor's monthly requisition until Contractor has rectified the situation and is in full compliance with these provisions.
- **4.15.2** The Contractor at all times shall keep the premises free from accumulation of waste materials or rubbish. At the completion of the Work he shall remove all his waste materials and rubbish from and about the Project as well as all his tools, construction equipment, machinery and surplus materials and shall leave the premises in a clean and satisfactory condition.
- **4.15.3** If the Contractor fails to clean up at the completion of the Work, the State may do so as provided in Paragraph 3.4 and the cost thereof shall be charged to the Contractor.

#### 4.16 COMMUNICATIONS

**4.16.1** The Contractor shall forward all communications to the State through the Architect.

#### 4.17 ROYALTIES AND PATENTS

The Contractor shall pay all royalties and license fees. The Contractor shall defend all suits or claims for infringement of any patent rights and shall save the State harmless from loss on account thereof, except that the State shall be responsible for all such loss when a particular design, process or the product of a particular manufacturer or manufacturers is specified, but if the Contractor has reason to believe that the design, process or product specified is an infringement of a patent, The Contractor shall be responsible for such loss unless he promptly gives such information to the Architect.

#### 4.18 INDEMNIFICATION

The Contractor hereby agrees and consents to indemnify and hold harmless the State under the terms and conditions specified in "Attachment C".

#### **SUBCONTRACTORS**

## 5.1 **DEFINITION**

- **5.1.1** A Subcontractor is a person or entity who has a direct contract with the Contractor to perform any of the Work at the site. The term Subcontractor means a Subcontractor or his / her authorized representative. The term Subcontractor does not include any separate contractor or his / her subcontractors.
- **5.1.2** A Sub-subcontractor is a person or entity who has a direct or indirect contract with a Subcontractor to perform any of the Work at the site. The term Sub-subcontractor means a Sub-subcontractor or an authorized representative thereof.

## 5.2 AWARD OF SUBCONTRACTS AND OTHER CONTRACTS FOR PORTIONS OFTHE WORK

Unless other procedures are specified or required by the Contract Documents or the Bidding Documents, then the following provisions are applicable:

- **5.2.1** The Parties agree that it is in the best interests of both the Contractor and the State to determine and select subcontractors as soon as possible after the Contractor has been awarded the Project. Therefore, the Contractor agrees, to furnish to the State and the Architect in writing the names of the persons or entities (including those who are to furnish materials or equipment fabricated to a special design) proposed for each of the principal portions of the Work within Ten (10) working days of award of the Project to Contractor. The State will reply within Ten (10) working days to the Contractor in writing stating whether or not the State, after due investigation, has reasonable objection to any such proposed person or entity.
- **5.2.2** The Contractor shall not contract with any such proposed person or entity to whom the State has made reasonable objection under the provisions of Subparagraph 5.2.1 of these General Conditions. The Contractor shall not be required to contract with anyone to whom he has a reasonable objection.
- **5.2.3** If the State has reasonable objection to any such proposed person or entity, the Contractor shall submit a substitute to whom the State has no reasonable objection, and the Contract Sum shall be increased or decreased by the difference in cost occasioned by such substitution and an appropriate Change Order shall be issued; however, no increase in the Contract Sum shall be allowed for any such substitution unless the Contractor has acted promptly and responsively in submitting names as required by Subparagraph 5.2.1. of these General Conditions.
- **5.2.4** The Contractor shall make no substitution for any Subcontractor, person or entity previously selected if the State makes a reasonable objection to such substitution.

#### 5.3 SUBCONTRACTUAL RELATIONS

**5.3.1** By written agreement, the Contractor shall require each Subcontractor, to the extent of the Work to be performed by the Subcontractor, to be bound to the Contractor by the terms of the Contract Documents, and to assume toward the Contractor all the obligations and responsibilities

which the Contractor, by these Documents, assumes toward the State and the Architect. Said agreement shall preserve and protect the rights of the State and the Architect under the Contract Documents with respect to the Work to be performed by the Subcontractor so that the subcontracting thereof will not prejudice such rights, and shall allow to the Subcontractor, unless specifically provided otherwise in the Contractor Subcontractor agreement, the benefit of all rights, remedies and redress against the Contractor that the Contractor, by these Documents, has against the State. The Contractor shall require each Subcontractor to enter into similar agreements with his Sub-subcontractors. The Contractor shall make available to each proposed Subcontractor, prior to the execution of the Subcontract, copies of the Contract Documents to which the Subcontractor will be bound by this Paragraph 5.3, and identify to the Subcontractor any terms and conditions of the proposed Subcontract which may be at variance with the Contract Documents. Each Subcontractor shall similarly make copies of such Documents available to his / her Sub-subcontractors.

#### **ARTICLE 6**

#### WORK BY STATE OR BY SEPARATE CONTRACTORS

# 6.1 STATE'S RIGHT TO PERFORM WORK AND TO AWARD SEPARATE CONTRACTS

- **6.1.1** The State reserves the right to perform work related to the Project with its own forces, and to award separate contracts in connection with other portions of the Project or other work on the site under these or similar Conditions of the Contract. If the Contractor claims that delay or additional cost is involved because of such action by the State, he / she shall make such claim as provided elsewhere in the Contract Documents.
- **6.1.2** When separate Contracts are awarded for different portions of the Project or other work on the site, the term Contractor in the Contract Documents in each case shall mean the Contractor who executes each separate State Contractor Agreement.
- **6.1.3** The State will provide for the coordination of the work of its own forces and of each separate contractor with the Work of the Contractor, who shall cooperate therewith as provided in Paragraph 6.2.

## 6.2 MUTUAL RESPONSIBILITY

- **6.2.1** The Contractor shall afford the State and separate contractors reasonable opportunity for the introduction and storage of their materials and equipment and the execution of their work, and shall connect and coordinate his / her Work with the work of the State and separate contractors as required by the Contract Documents.
- **6.2.2** If any part of the Contractor's Work depends for proper execution or results upon the work of the State or any separate contractor, the Contractor shall, prior to proceeding with the Work, promptly report to the Architect any apparent discrepancies or defects in such other work that render it unsuitable for such proper execution and results. Failure of the Contractor so to report shall constitute an acceptance of the State's or separate contractors work as fit and proper to receive

his Work, except as to defects which may subsequently become apparent in such work by others.

- **6.2.3** Any costs caused by defective or ill-timed work shall be borne by the party responsible therefor.
- **6.2.4** Should the Contractor wrongfully cause damage to the work or property of the State, or to other work on the site, the Contractor shall promptly remedy such damage as provided in Subparagraph 10.2.5 of these General Conditions.

#### 6.3 STATE'S RIGHT TO CLEAN UP

**6.3.1** If a dispute arises between the Contractor and separate contractors as to their responsibility for cleaning up as required by Paragraph 4.15 of these General Conditions, the State may clean up and charge the cost thereof to the Contractor.

#### **ARTICLE 7**

## MISCELLANEOUS PROVISIONS

#### 7.1 GOVERNING LAW

**7.1.1** The Contract shall be governed by the laws of the State of Vermont.

### 7.2 SUCCESSORS AND ASSIGNS

**7.2.1** The State and the Contractor each bind themselves, their partners, successors, assigns and legal representatives to the other party hereto and to the partners, successors, assigns and legal representatives of such other party in respect to all covenants, agreements and obligations contained in the Contract Documents. Neither party to the Contract shall assign the Contract or sublet it as a whole without the written consent of the other, nor shall the Contractor assign any moneys due or to become due to him or her hereunder, without the previous written consent of the State.

## 7.3 WRITTEN NOTICE

**7.3.1** Written notice shall be deemed to have been duly served if delivered in person to the individual or member of the firm or entity or to an officer of the corporation for whom it was intended, or if delivered at or sent by registered or certified mail to the last business address known to him who gives the notice.

#### 7.4 CLAIMS FOR DAMAGES

**7.4.1** Should either party to the Contract suffer injury or damage to person or property because of any act or omission of the other party or of any of his / her employees, agents or others for whose acts he / she is legally liable, claim shall be made in writing to such other party within a reasonable time after the first observance of such injury or damage.

## 7.5 BOND REQUIREMENTS

The Contractor hereby agrees to comply with the State's bonding requirements as identified in the Instructions to Bidders which are included in the RFP and are attached and incorporated into the Contract Documents.

#### 7.6 RIGHTS AND REMEDIES

- **7.6.1** The duties and obligations imposed by the Contract Documents and the rights and remedies available there under shall be in addition to and not a limitation of any duties, obligations, rights and remedies otherwise imposed or available by law.
- **7.6.2** No action or failure to act by the State, Architect or Contractor shall constitute a waiver of any right or duty afforded any of them under the Contract, nor shall any such action or failure to act constitute an approval of or acquiescence in any breach thereunder, except as may be specifically agreed in writing.

#### 7.7 TESTS

- **7.7.1** If the Contract Documents, laws, ordinances, rules, regulations or orders of any public authority having jurisdiction require any portion of the Work to be inspected, tested or approved, the Contractor shall give the Architect timely notice of its readiness so the Architect may observe such inspection, testing or approval. The Contractor shall bear all costs of such inspections, tests or approvals conducted by public authorities.
- **7.7.2** If the Architect determines that any Work requires special inspection, testing, or approval which Subparagraph 7.7.1 does not include, he or she will, upon written authorization from the State, instruct the Contractor to order such special inspection, testing or approval, and the Contractor shall give notice as provided in Subparagraph 7.7.1. If such special inspection or testing reveals a failure of the Work to comply with the requirements of the Contract Documents, the Contractor shall bear all costs thereof, including compensation for the Architect's additional services made necessary by such failure; otherwise the State shall bear such costs, and an appropriate Change Order shall be issued.
- **7.7.3** Required certificates of inspection, testing or approval shall be secured by the Contractor and promptly delivered by him to the Architect.
- **7.7.4** If the Architect is to observe inspections, tests or approvals required by the Contract Documents, he will do so promptly.

#### 7.8 INTEREST

Payments due and unpaid under the Contract Documents shall bear interest from the date payment is due at a rate of twelve percent (12 %) per annum. A payment is NOT due and payable unless and until the State has received an invoice filled out correctly and completely with all required supporting documentation.

## 7.9 DISPUTE RESOLUTION PROCESS

Any claim, dispute or other matter in question not resolved by the process identified in Paragraph

2.2.14 of these General Conditions shall be subject to the following dispute resolution process: Contractor may seek review by the Commissioner of Buildings and General Services of the Architect's recommendation. Contractor must seek review by the Commissioner within ten (10) days of the Architect's written recommendation otherwise the recommendation shall be deemed accepted by the Contractor. Failing resolution by the Commissioner, the Contractor then may request that the parties submit the claim to non-binding mediation with a mutually agreed upon mediator utilizing the most current Construction Industry Mediation Rules of the American Arbitration Association.

#### **ARTICLE 8**

#### **TIME**

#### 8.1 **DEFINITIONS**

- **8.1.1** Unless otherwise provided, the Contract Time is the period of time allotted in the Contract Documents for Substantial Completion of the Work as defined in Subparagraph 8.1.3 of these General Conditions, including authorized adjustments thereto.
- **8.1.2** The date of commencement of the Work is the date established in a notice to proceed. If there is no notice to proceed, it shall be the date indicated in State Contractor Agreement. In the absence of a notice to proceed and an express commencement date in the State Contractor Agreement, then the date of execution of the State Contractor Agreement shall be the commencement date.
- **8.1.3** The Date of Substantial Completion of the Work or designated portion thereof is the Date certified by the Architect when construction is sufficiently complete, in accordance with the Contract Documents, so the State can occupy or utilize the Work or designated portion thereof for the use for which it is intended. The Work shall not be considered substantially complete prior to the receipt of approved O & M manuals and record drawings as required by paragraph 4.11 and as required elsewhere in the Contract Documents.
  - **8.1.3.1** Contract completion date shall be defined as: The contract work shall not be considered complete prior to the receipt of approved O & M manuals and record drawings as required by paragraph 4.11 and as required elsewhere in the contract documents.
- **8.1.4** The term day as used in the Contract Documents shall mean calendar day unless otherwise specifically designated.

### 8.2 PROGRESS AND COMPLETION

- **8.2.1** All time limits stated in the Contract Documents are material terms and time is the essence of the Contract. A failure by Contractor to do what is required by the time specified in the Contract Documents is a breach of the contract.
- 8.2.2 The Contractor shall begin the Work on the date of commencement as defined in

Subparagraph 8.1.2 of these General Conditions. He or She shall carry the Work forward expeditiously with adequate forces and shall achieve Substantial Completion within the Contract Time.

8.2.3 If in the sole opinion of the State, the Contractor fails to commence work on the project or to complete the work of said project within the time specified above, or to prosecute the work in such a manner that it appears that the completion date can be assured, the State shall have the right to notify the Contractor by Certified Mail that the terms of the Contract have been violated, and that effective immediately the Contract is terminated and the State has the right to and in fact is taking over and attending to completion of the project without prejudice to the State's remedies for any losses sustained

#### 8.3 DELAYS AND EXTENSIONS OF TIME

- **8.3.1** If the Contractor is delayed at any time in the progress of the Work by any act or failure to act by the State or the Architect, or by any employee of either, or by any separate contractor employed by the State, or by changes ordered in the Work, fire, unusual delay in transportation, adverse weather conditions not reasonably anticipatable, unavoidable casualties, or any causes beyond the Contractor's control, or by delay authorized by the State pending alternative dispute resolution proceedings, or by any other cause which may justify the delay, then the Contract Time shall be extended by Change Order for such reasonable time as recommended by the Architect and approved by the State.
- **8.3.2** Any claim for extension of time shall be made in writing to the Architect not more than twenty days after the commencement of the delay; otherwise it shall be waived. In the case of a continuing delay only one claim is necessary. The Contractor shall provide an estimate of the probable effect of such delay on the progress of the Work.
- **8.3.3** This Paragraph 8.3 does not exclude the recovery of damages for delay by either party under other provisions of the Contract Documents.

#### **ARTICLE 9**

#### PAYMENTS AND COMPLETION

#### 9.1 CONTRACT SUM

**9.1.1** The Contract Sum is stated in the State-Contractor Agreement and, including authorized adjustments thereto, is the total amount payable by the State to the Contractor for the performance of the Work under the Contract Documents.

#### 9.2 SCHEDULE OF VALUES

**9.2.1** Before the first Application for Payment, the Contractor shall submit to the Architect a schedule of values allocated to the various portions of the Work, prepared in such form and supported by such data to substantiate its accuracy as the Architect may require. This schedule, unless objected to by the Architect, shall be used only as a basis for the Contractor's Applications for Payment.

#### 9.3 APPLICATIONS FOR PAYMENT

- **9.3.1** At least ten days before the date for each progress payment established in the State-Contractor Agreement, the Contractor shall submit to the Architect an itemized Application for Payment, notarized if required, supported by such data substantiating the Contractor's right to payment as the State or the Architect may require. The application for payment must, at a minimum, reflect retainage and the required waivers of lien and any other support documentation enumerated elsewhere in the Contract Documents.
- **9.3.2** Unless otherwise provided in the Contract Documents, payments will be made on account of materials or equipment not incorporated in the Work but delivered and suitably stored at the site and, if approved in advance by the State, payments may similarly be made for materials or equipment suitably stored at some other location agreed upon in writing. Payments for materials or equipment stored on or off the site shall be conditioned upon submission by the Contractor of bills of sale or such other procedures satisfactory to the State to establish the State's title to such materials or equipment or otherwise protect the State's interest, including applicable insurance and transportation to the site for those materials and equipment stored off the site.
- **9.3.3** The Contractor warrants that title to all Work, materials and equipment covered by an Application for Payment will pass to the State either by incorporation in the construction or upon the receipt of payment by the Contractor, whichever occurs first, free and clear of all liens, claims, security interests or encumbrances, hereinafter referred to in this Article 9 as "liens"; and that no Work, materials or equipment covered by an Application for Payment will have been acquired by the Contractor, or by any other person performing Work at the site or furnishing materials and equipment for the Project, subject to an agreement under which an interest therein or an encumbrance thereon is retained by the seller or otherwise imposed by the Contractor or such other person.

#### 9.4 CERTIFICATES FOR PAYMENT

- **9.4.1** The Architect will, within seven days after the receipt of the Contractor's completed Application for Payment, either issue a Certificate for Payment to the State, with a copy to the Contractor, for such amount as the Architect determines is properly due, or notify the Contractor in writing his or her reasons for withholding a Certificate as provided in Subparagraph 9.6.1 of these General Conditions. In case of conflict between any time requirements in this paragraph and any time requirements for action by the Architect on Contractor's Application for Payment contained in the State-Contractor Agreement, the time requirements contained in the State-Contractor Agreement shall control.
- **9.4.2** The issuance of a Certificate for Payment will constitute a representation by the Architect to the State, based on his observations at the site as provided in Subparagraph 2.2.3 of these General Conditions and the data comprising the Application for Payment, that the Work has progressed to the point indicated; that, to the best of his or her knowledge, information and belief, the quality of the Work is in accordance with the Contract Documents (subject to an evaluation of the Work for conformance with the Contract Documents upon Substantial Completion, to the results of any subsequent tests required by or performed under the Contract Documents, to minor deviations from the Contract Documents correctable prior to completion, and to any specific

qualifications stated in his Certificate); and that the Contractor is entitled to payment in the amount certified.

#### 9.5 PROGRESS PAYMENTS

- **9.5.1** After the Architect has issued a Certificate for Payment, the State shall make payment in the manner and within the time provided in the Contract Documents.
- **9.5.2** The Contractor shall promptly pay each Subcontractor, upon receipt of payment from the State, out of the amount paid to the Contractor on account of such Subcontractor's Work, the amount to which said Subcontractor is entitled, reflecting the percentage actually retained, if any, from payments to the Contractor on account of such Subcontractor's Work. The Contractor shall, by an appropriate agreement with each Subcontractor, require each Subcontractor to make payments to his or her Sub-subcontractors in similar manner.
- **9.5.3** The Architect may, on request and at his discretion, furnish to any Subcontractor, if practicable, information regarding the percentages of completion or the amounts applied for by the Contractor and the action taken thereon by the Architect on account of Work done by such Subcontractor.
- **9.5.4** Neither the State nor the Architect shall have any obligation to pay or to see to the payment of any moneys to any Subcontractor.
- **9.5.5** No Certificate for a progress payment, nor any progress payment, nor any partial or entire use or occupancy of the Project by the State, shall constitute an acceptance of any Work not in accordance with the Contract Documents.

#### 9.6 PAYMENTS WITHHELD

- **9.6.1** The Architect may decline to certify payment and may withhold his or her Certificate in whole or in part, to the extent necessary reasonably to protect the State, if in the Architect's opinion he or she is unable to make representations to the State as provided in Subparagraph 9.4.2 of these General Conditions. If the Architect is unable to make representations to the State as provided in the aforementioned Subparagraph 9.4.2 and to certify payment in the amount of the Application, he or she will notify the Contractor as provided in Subparagraph 9.4.1 of these General Conditions. If the Contractor and the Architect cannot agree on a revised amount, the Architect will promptly issue a Certificate for Payment for the amount for which he or she is able to make such representations to the State. The Architect may also decline to certify payment or, because of subsequently discovered evidence or subsequent observations, he or she may nullify the whole or any part of any Certificate for Payment previously issued, to such extent as may be necessary in his or her opinion to protect the State from loss because of:
  - (a) Defective Work not remedied,
  - (b) Third party claims filed or reasonable evidence indicating probable filing of such claims,
  - (c) Failure of the Contractor to make payments properly to Subcontractors or for labor, materials or equipment,
  - (d) Reasonable evidence that the Work cannot be completed for the unpaid balance of the Contract Sum,

- (e) Damage to the State or another contractor,
- (f) Reasonable evidence that the Work will not be completed within the Contract Time, or
- (g) Persistent failure to carry out the Work in accordance with the Contract Documents.
- **9.6.2** When the above grounds in Subparagraph 9.6.1 are removed, payment shall be made for amounts withheld because of them.

#### 9.7 FAILURE OF PAYMENT

**9.7.1** If the Architect does not issue a Certificate for Payment, through no fault of the Contractor, within seven days after receipt of the Contractor's Application for Payment, or if the State does not pay the Contractor within seven days after the date established in the Contract Documents any amount certified by the Architect or awarded by alternative dispute resolution proceedings, then the Contractor may, upon seven additional days written notice to the State and the Architect, stop the Work until payment of the amount owing has been received. The Contract Sum shall be increased by the amount of the Contractor's reasonable costs of shut-down, delay and start-up, which shall be effected by appropriate Change Order in accordance with Paragraph 12.3.

#### 9.8 SUBSTANTIAL COMPLETION

- **9.8.1** When the Contractor considers that the Work, or a designated portion thereof which has been accepted in writing to by the State, is substantially complete as defined in Subparagraph 8.1.3 of these General Conditions, the Contractor shall prepare for submission to the Architect a list of items to be completed or corrected. The failure to include any items on such list does not alter the responsibility of the Contractor to complete all Work in accordance with the Contract Documents. When the Architect on the basis of an inspection determines that the Work or designated portion thereof is substantially complete, he will then prepare a Certificate of Substantial Completion which shall establish the Date of Substantial Completion, shall state the responsibilities of the State and the Contractor for security, maintenance, heat, utilities, damage to the Work, and insurance, and shall fix the time, which shall not exceed thirty (30) days, within which the Contractor shall complete the items listed therein. Warranties required by the Contract Documents shall commence on the Date of Substantial Completion of the Work or designated portion thereof unless otherwise provided in the Certificate of Substantial Completion. The Certificate of Substantial Completion shall be submitted to the State and the Contractor for their written acceptance of the responsibilities assigned to them in such Certificate. The Work shall not be considered substantially complete prior to the receipt of approved O & M manuals and record drawings as required by paragraph 4.11 and as required elsewhere in the Contract Documents.
- **9.8.2** Upon Substantial Completion of the Work or designated portion thereof and upon application by the Contractor and certification by the Architect, the State shall make payment, reflecting adjustment in retainage, if any, for such Work or portion thereof, as provided in the Contract Documents.

#### 9.9 FINAL COMPLETION AND FINAL PAYMENT

**9.9.1** Upon receipt of written notice that the Work is ready for final inspection and acceptance and upon receipt of a final Application for Payment, the Architect will promptly make such

inspection and, when he or she finds the Work acceptable under the Contract Documents and the Contract fully performed, he or she will promptly issue a final Certificate for Payment stating that to the best of his / her knowledge, information and belief, and on the basis of his or her observations and inspections, the Work has been completed in accordance with the terms and conditions of the Contract Documents and that the entire balance found to be due the Contractor, and noted in said final Certificate, is due and payable. The Architect's final Certificate for Payment will constitute a further representation that the Contractor has fulfilled the conditions entitling him or her to final payment as set forth in Subparagraph 9.9.2 of these General Conditions.

- **9.9.2** Neither the final payment nor the remaining retained percentage shall become due until the Contractor submits to the Architect (1) an affidavit that all payrolls, bills for materials and equipment, and other indebtedness connected with the Work for which the State or its property might in any way be responsible, have been paid or otherwise satisfied; (2) consent of surety, if any, to final payment; (3) Other data establishing payment or satisfaction of all such obligations, including, but not limited to, receipts, releases, or other supporting documentation. and (4) final waivers of liens arising out of the Contract, to the extent and in such form as may be designated by the State. If any Subcontractor refuses to furnish a release or waiver required by the State, the Contractor may furnish a bond satisfactory to the State to indemnify him against any such lien. If any such lien remains unsatisfied after all payments are made, the Contractor shall refund to the State all moneys that the latter may be compelled to pay in discharging such lien, including all costs and reasonable attorneys' fees.
- **9.9.3** If, after Substantial Completion of the Work, final completion thereof is materially delayed through no fault of the Contractor or by the issuance of Change Orders affecting final completion, and the Architect so confirms, the State shall, upon application by the Contractor and certification by the Architect, and without terminating the Contract, make payment of the balance due for that portion of the Work fully completed and accepted. If the remaining balance for Work not fully completed or corrected is less than the retainage stipulated in the Contract Documents, and if bonds have been furnished as required by the Contract Documents, the written consent of the surety to the payment of the balance due for that portion of the Work fully completed and accepted shall be submitted by the Contractor to the Architect prior to certification of such payment. Such payment shall be made under the terms and conditions governing final payment, except that it shall not constitute a waiver of claims.
- **9.9.4** The making of final payment shall constitute a waiver of all claims by the State except those arising from:
  - (a) unsettled liens,
  - (b) faulty or defective Work appearing after Substantial Completion,
  - (c) failure of the Work to comply with the requirements of the Contract Documents, or
  - (d) terms of any special warranties required by the Contract Documents.
- **9.9.5** The acceptance of final payment shall constitute a waiver of all claims by the Contractor except those previously made in writing and identified by the Contractor as unsettled at the time of the final Application for Payment.

#### PROTECTION OF PERSONS AND PROPERTY

#### 10.1 SAFETY PRECAUTIONS AND PROGRAMS

The Contractor shall be responsible for initiating, maintaining and supervising all safety precautions and programs in connection with the Work.

#### 10.2 SAFETY OF PERSONS AND PROPERTY

- **10.2.1** The Contractor shall take all reasonable precautions for the safety of, and shall provide all reasonable protection to prevent damage, injury or loss to:
  - (a) all employees on the Work and all other persons who may be affected thereby;
  - (b) all the Work and all materials and equipment to be incorporated therein, whether in storage on or off-the site, under the care; custody or control of the Contractor or any of his Subcontractors or Sub-subcontractors; and
  - (c) other property at the site or adjacent thereto, including trees, shrubs; lawns, walks, pavements, roadways, structures and utilities not designated for removal, relocation or replacement in the course of construction.
- **10.2.2** The Contractor shall give all notices and comply with all applicable laws, ordinances, rules, regulations and lawful orders of any public authority bearing on the safety of persons or property or their protection from damage, injury or loss.
- 10.2.3 The Contractor shall erect and maintain, as required by existing conditions and progress of the Work, all reasonable safeguards for safety and protection, including posting danger signs and other warnings against hazards, promulgating safety regulations and notifying owners and users of adjacent utilities.
- **10.2.4** When the use or storage of explosives or other hazardous materials or equipment is necessary for the execution of the Work, the Contractor shall: (1) exercise the utmost care and shall carry on such activities under the supervision of properly qualified personnel; and (2) give the State notice, in writing, seven (7) days in advance of the planned activity of the Contractor's intent to store and/ or use explosives or other hazardous materials or equipment on the project site.
- **10.2.5** When the project involves the placement or replacement of roofing systems of all types on State- owned buildings, including flat, low-pitched and steep roofs, Contractor must comply with all requirements of the latest edition of VOSHA Safety and Health Standards for Construction (29 CFR 1926) including, but not limited to, Subpart M Fall Protection. Prior to execution of a contract by BGS, contractors engaged in placement or replacement of roofing systems of all types must submit a signed certification statement attesting to their intention to comply with VOSHA Fall Protection Regulations. Such certification shall be submitted on a State-provided form along with proof of insurance.
- **10.2.6** The Contractor shall promptly remedy all damage or loss (consistent with Contractor's obligations described in Attachment C) to any property referred to in Clauses 10.2.1(b) and 10.2.1(c) of these General Conditions caused in whole or in part by the Contractor, any

Subcontractor, any Sub-subcontractor, or anyone directly or indirectly employed by any of them, or by anyone for whose acts any of them may be liable and for which the Contractor is responsible under Clauses 10.2.1(b) and 10.2.1(c), except damage or loss attributable to the acts or omissions of the State or Architect or anyone directly or indirectly employed by either of them, or by anyone for whose acts either of them may be liable, and not attributable to the fault or negligence of the Contractor. The foregoing obligations of the Contractor are in addition to his obligations referenced under Paragraph 4.18.

- **10.2.7** The Contractor shall designate a responsible member of his organization at the site whose duty shall be the prevention of accidents. This person shall be the Contractor's superintendent unless otherwise designated by the Contractor in writing to the State and the Architect.
- **10.2.8** The Contractor shall not load or permit any part of the Work to be loaded so as to endanger its safety.

#### 10.3 EMERGENCIES

**10.3.1** In any emergency affecting the safety of persons or property, the Contractor shall act to prevent threatened damage, injury or loss. Any additional compensation or extension of time claimed by the Contractor on account of emergency work shall be determined as provided in Article 12 for Changes in the Work.

#### **ARTICLE 11**

#### **INSURANCE**

#### Insurance obligations of the parties to this agreement are fully set forth below:

Attachment "C", numbered paragraph 8 has been deleted in its entirety and the following provisions are applicable:

#### 11 INSURANCE – GENERAL REQUIREMENTS

- 11.1 Insurance obtained by the Contractor to cover the below-listed requirements shall be procured from an insurance company registered and licensed to do business in the State of Vermont. Before the Contract is executed, the Contractor shall file with the State a certificate of insurance, executed by an insurance company or its licensed agent(s), on form(s) satisfactory to the State, stating that with respect to the Contract awarded, the Contractor carries insurance in accordance with the following requirements:
  - (a) Workers' Compensation Insurance: With respect to all operations performed, the Party shall carry workers' compensation insurance in accordance with the laws of the State of Vermont. Vermont will accept an out-of-state employer's workers' compensation coverage while operating in Vermont provided that the insurance carrier is licensed to write insurance in Vermont and an amendatory endorsement is added to the policy adding Vermont for coverage purposes. Otherwise, the party shall secure a Vermont workers' compensation policy, if necessary to comply with Vermont law.

(b) Commercial General Liability (CGL) Insurance: With respect to all operations performed by the Contractor and any Subcontractors, the Contractor shall carry Commercial General Liability (CGL) Insurance affording all major divisions of coverage including, but not limited to:

Premises Operations
Independent Contractors' Protective Products and Completed Operations
Personal Injury Liability
Contractual Liability
Broad Form Property Damage
Explosion, Collapse and Underground (XCU) Coverage
Fire Legal Liability

State may require additional specific liability coverage(s) when applicable.

Limits of Coverage shall be not less than:

\$1,000,000 Each Occurrence \$2,000,000 General Aggregate Applying, In Total To This Project Only \$1,000,000 Products/Completed Operations Aggregate \$1,000,000 Personal Injury and Advertising Liability \$50,000 Fire Legal Liability \$5,000 Medical Expense

Aggregate Limits of Insurance (Per Project) Form must be included (or as excepted in f below) with the Commercial General Liability Policy.

(c) Automobile Liability Insurance: Contractor shall carry Automobile Liability Insurance covering all motor vehicles, including owned, hired, borrowed and non-owned vehicles, used in connection with the project. Limits of Coverage shall be not less than:

\$1,000,000 Combined Single Limit

- (d) The Contractor shall require each Subcontractor, if any, to maintain the same Workers' Compensation, Commercial General Liability, and Automobile Liability coverage as specified in paragraphs (a), (b), and (c) above.
- (e) Scope of Insurance & Special Hazards: The insurance required under subparagraphs (b), (c), and (d) above shall provide protection for the Contractor and the Subcontractors, if any, against damage claims which may arise from work being performed. Special hazards, may be encountered by the insured or by any person directly or indirectly employed by the Contractor or a Subcontractor, and may be required.
- (f) If Contractor is unable to provide Aggregate Limits of Insurance (Per Project) as required in (b) above, Contractor shall purchase an Owner's Protective Liability Insurance Policy in the name of the State of Vermont to cover all exposures, including bodily injury and death, arising out of and in the course of this Contract. Limits of liability shall be the same as those required of the Contractor under paragraph (b) above.

Additional Insured. The General Liability and Property Damage coverages required for performance of this Agreement shall include the State of Vermont and its agencies, departments, officers and employees as Additional Insureds. If performance of this Agreement involves construction, or the transport of persons or hazardous materials, then the required Automotive Liability coverage shall include the State of Vermont and its agencies, departments, officers and employees as Additional Insureds. Coverage shall be primary and non-contributory with any other insurance and self-insurance.

Notice of Cancellation or Change. There shall be no cancellation, change, potential exhaustion of aggregate limits or non-renewal of insurance coverage(s) without thirty (30) days written prior written notice to the State of Vermont, Office of Purchasing & Contracting.

**11.2 Insurance Requirements for Builder's Risk Insurance.** The State of Vermont does not require Contractor to obtain Builder's Risk insurance for every construction project. The State of Vermont will evaluate each project prior to accepting bids to determine if the State will require the Contractor to obtain and maintain Builder's Risk insurance for the project.

Any project that involves NEW CONSTRUCTION will require the Contractor to obtain and maintain Builder's Risk insurance for the project.

At the discretion of the State, Contractor may not be required to obtain and maintain Builder's Risk insurance on CONSTRUCTION RENOVATION projects. Whether or not the State is going to require Contractor to obtain and maintain Builder's Risk insurance for a particular CONSTRUCTION RENOVATION project will be determined by the State prior to bidding the project and the requirement will be included in the Instructions to Bidders.

- 11.2.1 BUILDER'S RISK FOR NEW CONSTRUCTION PROJECTS. If the project is NEW CONSTRUCTION, then the following provisions apply:
  - (1) The State will require Contractor to purchase and maintain Builder's Risk Insurance. The Contractor shall name the Contractor and the State of Vermont as their interest may occur. Other parties shall be insured as the State may reasonably require.
  - (2) Contractor shall effect and maintain insurance on portions of the work stored off-site, on site and in transit. Boiler & Machinery Insurance may be used in conjunction with this coverage if it is required to meet the testing requirement.
  - (3) Any deductible applicable to loss covered by insurance provided hereunder shall be borne by the Contractor.
  - (4) Except as provided for in (1) above the State and Contractor waive all rights against each other and the Subcontractor, Sub-subcontractors, agents and employees of the other.
- 11.2. 2 BUILDER'S RISK FOR CONSTRUCTION RENOVATION PROJECTS. If the project is CONSTRUCTION RENOVATION of an existing structure, and the State has already determined that Contractor will not be obligated to obtain and maintain Builder's Risk insurance, then the following provisions apply:

- (1) The State will maintain property insurance upon the construction site and will not require Contractor to purchase and maintain Builder's Risk Insurance upon the entire work at the site.
- (2) The decision of the State to waive the requirement that the Contractor maintain builder's risk coverage does not waive Contractor's liability for damage to the State's real and personal property. Contractor's liability for loss to the State's real and personal property will be limited to the first \$100,000 of each and every property loss at the work site provided such loss is covered under the State's property insurance coverage. If the Contractor elects to meet this obligation by purchase of commercial insurance, this insurance shall name the Contractor and the State of Vermont as Named Insureds and shall include the interests of the Contractor and Subcontractors. Other parties shall be insured as the State may reasonably require.
- (3) Contractor shall effect and maintain insurance on portions of the work stored off-site, on site and in transit. Boiler & Machinery Insurance may be used in conjunction with this coverage if it is required to meet the testing requirement. The cost of any deductible applicable to loss covered by insurance provided hereunder shall be borne by the Contractor.
- (4) Except as provided for in 11.2.2 (1)-(3) above the State and Contractor waive all rights against each other and the Subcontractor, Sub-subcontractors, agents and employees of the other.

#### 11.2.3 Builder's Risk – additional provisions

The insurance specified shall be maintained in force until final acceptance of the project by the State.

- (5) Umbrella Excess Liability Policies may be used in conjunction with primary policies to comply with any of the limit requirements specified above.
- (6) "Claims-made" coverage forms are not acceptable without the express written prior consent of the State. Each policy furnished shall contain a rider or non-cancellation clause reading in substance as follows:
  - Anything herein to the contrary notwithstanding, notice of any cancellation, termination or alteration to the insurance contracts must be delivered by registered mail to the Commissioner, Department of Buildings and General Services, State of Vermont, Montpelier, Vermont, at least 60 days before effective cancellation, termination or alteration date unless all work required to be performed under the terms of the Contract is satisfactorily completed as evidenced by the formal acceptance by the State of Vermont.
- (7) No warranty is made that the coverages and limits listed herein are adequate to cover and protect the interests of the Contractor for the Contractor's operations. These are solely minimums that have been set to protect the interests of the State.
- **11.3** The State shall have power to adjust and settle any loss with the insurers.

#### **CHANGES IN THE WORK**

#### 12.1 CHANGE ORDERS

12.1.1 A Change Order is a written order to the Contractor signed by the State, the Architect, and the Contractor issued after execution of the Contract, authorizing a change in the Work or an adjustment in the Contract Sum or the Contract Time. The Contract Sum and the Contract Time may be changed only by Change Order. A Change Order signed by the Contractor indicates his agreement therewith, including the adjustment in the Contract Sum or the Contract Time. A Proposed Change Order (PCO) is a written request submitted to the Architect by either the Contractor or the State requesting a change to the contract price and / or schedule. A Construction Change Directive (CCD) is a written document prepared by the State and / or Architect directing the Contractor to make changes in the Work where the State and the Contractor have not reached an agreement on proposed changes in the contract sum or contract time. A Construction Change Directive becomes part of a PCO.

Payment for changes in the Work will only be made after a Change Order is signed by the Contractor and Architect and executed by the State.

**12.1.2** The State, without invalidating the Contract, may order changes in the Work within the general scope of the Contract consisting of additions, deletions or other revisions, the Contract Sum and the Contract Time being adjusted accordingly. All such changes in the Work REQUIRE AGREEMENT IN WRITING BY THE ACHITECT, STATE AND CONTRACTOR AND shall be authorized by A PROPOSED Change Order (PCO), and shall be performed under the applicable conditions of the Contract Documents. The State will prescribe a Proposed Change Order form for use by the Contractor. The Contractor shall submit all requests for Proposed Change Orders on the prescribed form.

The process, by which changes in the Work within the general scope of the Contract consisting of additions, deletions or other revisions, is as follows:

- (1) A Proposed Change Order (PCO) is submitted to the Architect for review. Said PCO is to be submitted on the State approved form.
- (2) If the PCO originated with the Contractor, the Architect will review the PCO and if the Architect approves the PCO, it is then presented to the State for approval.
- (3) The PCO is signed by the Project Manager.
- (4) Project Manager compiles all PCO's every thirty (30) days and submits them to the Architect to be converted / compiled by the Architect into one (1) Change Order.
- (5) The Change Order is then submitted to the Project Manager who immediately processes the CO with OPC as required by Bulletin 3.5 and BGS' Contracting Plan.
- **12.1.3** The cost or credit to the State resulting from a change in the Work shall be determined in one or more of the following ways:
  - (a) by mutual acceptance of a lump sum properly itemized and supported by sufficient substantiating data to permit evaluation;
  - (b) by unit prices stated in the Contract Documents or subsequently agreed upon;
  - (c) by cost to be determined in a manner agreed upon by the parties and a mutually

acceptable fixed or percentage fee; or **(d)** by the method provided in Subparagraph 12.1.4.

12.1.4 If none of the methods set forth in Clauses 12.1.3(a) –(d) is agreed upon, or cannot be agreed upon at the time due to the nature of the change, emergent conditions, or lack of required information, the Contractor, provided he receives a Construction Change Directive (CCD)) signed by the State, shall promptly proceed with the Work involved. Upon receipt of the CCD, the Contractor shall proceed with implementation of the CCD. The cost of such Work shall then be recommended by the Architect on the basis of the reasonable expenditures and savings of those performing the Work attributable to the change, including, in the case of an increase in the Contract Sum, a reasonable allowance for overhead and profit. In such case, and also under Clauses 12.1.3(c) and 12.1.3(d) above, the Contractor shall keep and present, in such form as the Architect may prescribe, an itemized accounting together with appropriate supporting data for inclusion in the CCD. Unless otherwise provided in the Contract Documents, cost shall be limited to the following: cost of materials, cost of delivery; cost of labor, including social security, medicare and unemployment insurance, and fringe benefits required by agreement or custom; workers' or workmen's compensation insurance; bond premiums; rental value of equipment and machinery; and the additional costs of supervision and field office personnel directly attributable to the change. Pending final determination of cost to the State, payments on account shall be made on the State's Certificate for Payment. The amount of credit to be allowed by the Contractor to the State for any deletion or change which results in a net decrease in the Contract Sum will be the amount of the actual net cost as confirmed by the Architect. When both additions and credits covering related Work or substitutions are involved in any one change, the allowance for Overhead and profit shall be figured on the basis of the net increase, if any, with respect to that change. It is the Project Manager's responsibility to ensure that All CCD's are converted / incorporated into a PCO within Thirty (30) days of completion of the changes to the Work that was the subject of the CCD. The parties agree to utilize the CCD form provided or approved by the State.

**12.1.5** If unit prices are stated in the Contract Documents or subsequently agreed upon, and if the quantities originally contemplated are so changed in a Proposed Change Order that application of the agreed unit prices to the quantities of Work proposed will cause substantial in- equity to the State or the Contractor, the applicable unit prices shall be equitably adjusted.

#### 12.2 CONCEALED CONDITIONS

12.2.1 Should concealed conditions encountered in the performance of the Work below the surface of the ground, or should concealed or unknown conditions in an existing structure be at variance with the conditions indicated by the Contract Documents or should unknown physical conditions below the surface of the ground or should concealed or unknown conditions in an existing structure of an unusual nature, differing materially from those ordinarily encountered and generally recognized as inherent in work of the character provided for in this Contract, be encountered, the Contract Sum shall be equitably adjusted by Change Order upon claim by either

party made within twenty days after the first observance of the conditions.

#### 12.3 CLAIMS FOR ADDITIONAL COST

- 12.3.1 If the Contractor wishes to make a claim for an increase in the Contract Sum, he shall give the Architect written notice thereof within twenty days after the occurrence of the event giving rise to such claim. This notice shall be given by the Contractor before proceeding to execute the Work, except in an emergency endangering life or property in which case the Contractor shall proceed in accordance with Paragraph 10.3 of these General Conditions. No such claim shall be valid unless so made. If the State and the Contractor cannot agree on the amount of the adjustment in the Contract Sum, it shall be determined by the Architect. Any change in the Contract Sum resulting from such claim shall be authorized by Change Order.
- **12.3.2** If the Contractor claims that additional cost is involved because of, but not limited to, (1) any order by the State to stop the Work pursuant to Paragraph 3.3 of these General Conditions where the Contractor was not at fault, or (2) failure of payment by the State pursuant to Paragraph 9.7(of these General Conditions), the Contractor shall make such claim as provided in Subparagraph 12.3.1 of this agreement.
- **12.3.3** The Architect will have authority to order minor changes in the Work not involving an adjustment in the Contract Sum or an extension of the Contract Time and not inconsistent with the intent of the Contract Documents. Such changes shall be effected by written order and shall be binding on the State and Contractor within 10 days unless the Contractor or the State objects to the change in writing.

#### **ARTICLE 13**

#### UNCOVERING AND CORRECTION OF WORK

#### 13.1 UNCOVERING OF WORK

- **13.1.1** If any portion of the Work should be covered contrary to the request of the Architect or to requirements specifically expressed in the Contract Documents, it must, if required in writing by the Architect, be uncovered for his or her observation and shall be replaced at the Contractor's expense.
- 13.1.2 If any other portion of the Work has been covered which the Architect has not specifically requested to observe prior to being covered, the Architect may request to see such Work and it shall be uncovered by the Contractor. If such Work is found in accordance with the Contract Documents, the cost of uncovering and replacement shall, by appropriate Change Order, be charged to the State. If such Work is found not to be in accordance with the Contract Documents, the Contractor shall pay such costs unless it is found that this condition was caused by the State or a separate contractor as provided in Article 6 of these General Conditions, in which event the State shall be responsible for the payment of such costs.

#### 13.2 CORRECTION OF WORK

**13.2.1** The Contractor shall be responsible for correcting all Work which the Architect has found to be defective or which fails to conform to the Contract Documents whether observed before or after Substantial Completion and whether or not fabricated, installed or completed. The Contractor shall bear all costs of correcting such rejected Work, including compensation for the Architect's and the State's additional services made necessary thereby.

The Architect, upon a finding of defect or failure to conform, shall immediately notify the State and Contractor, in writing, of the defect. The Contractor shall begin correcting the defective or non-conforming Work within ten (10) days unless the State agrees to a Change Order which reflects the reduction in Contract Sum due to the value of diminishment of the defective or nonconforming Work.

- 13.2.2 If, within one year after the Date of Substantial Completion of the Work or designated portion thereof or within one year after acceptance by the State of designated equipment or within such longer period of time as may be prescribed by law or by the terms of any applicable special warranty required by the Contract Documents, any of the Work is found to be defective or not in accordance with the Contract Documents, the Contractor shall correct it promptly after receipt of a written notice from the State to do so unless the State has previously given the Contractor a written acceptance of such condition. This obligation shall survive termination of the Contract. The State shall give such notice promptly after discovery of the condition.
- **13.2.3** The Contractor shall remove from the site all portions of the Work which are defective or non-conforming and which have not been corrected under Subparagraphs 4.5.1, 13.2.1 and 13.2.2 of these General Conditions, unless removal is waived by the State.
- **13.2.4** If the Contractor fails to correct defective or nonconforming Work as provided in Subparagraphs 4.5.1, 13.2.1 and 13.2.2, the State may correct it in accordance with Paragraph 3.4 of these General Conditions.
- 13.2.5 If the Contractor does not proceed with the correction of such defective or non- conforming Work within a reasonable time fixed by written notice from the Architect, the State may remove it and may store the materials or equipment at the expense of the Contractor. If the Contractor does not pay the cost of such removal and storage within ten days thereafter, the State may upon ten additional days written notice sell such Work at auction or at private sale and shall account for the net proceeds thereof, after deducting all the costs that should have been borne by the Contractor, including compensation for the Architect's and the State's additional services and expenses made necessary thereby. If such proceeds of sale do not cover all costs which the Contractor should have borne, the difference shall be charged to the Contractor and an appropriate Change Order shall be issued. If the payments then or thereafter due the Contractor are not sufficient to cover such amount, the Contractor shall pay the difference to the State.
- **13.2.6** The Contractor shall bear the cost of making good all work of the State or separate contractors destroyed or damaged by such correction or removal.
- 13.2.7 Nothing contained in this Paragraph 13.2 shall be construed to establish a period of limitation with respect to any other obligation which the Contractor might have under the

Contract Documents, including Paragraph 4.5 of these General Conditions. The establishment of the time period of one year after the Date of Substantial Completion or such longer period of time as may be prescribed by law or by the terms of any warranty required by the Contract Documents relates only to the specific obligation of the Contractor to correct the Work, and has no relationship to the time within which his obligation to comply with the Contract Documents may be sought to be enforced, nor to the time within which proceedings may be commenced to establish the Contractor's liability with respect to his obligations other than specifically to correct the Work.

#### 13.3 ACCEPTANCE OF DEFECTIVE OR NON-CONFORMING WORK

13.3.1 The State may accept defective or nonconforming Work pursuant to Paragraph 13.2.1 of this agreement. If the State elects to accept the defective or nonconforming Work, a Change Order will be issued to reflect a reduction in the Contract Sum. The Architect will recommend to the State the value of diminishment of the defective or nonconforming Work. Such adjustment shall be effected whether or not final payment has been made.

#### **ARTICLE 14**

#### TERMINATION AND SUSPENSION OF THE CONTRACT

#### 14.1 TERMINATION BY THE CONTRACTOR

If the Work is stopped for a period of thirty days under an order of any court or other public authority having jurisdiction, or as a result of an act of government, such as a declaration of a national emergency making materials unavailable, through no act or fault of the Contractor or a Subcontractor or their agents or employees or any other persons performing any of the Work under a contract with the Contractor, or if the Work should be stopped for a period of thirty days by the Contractor because the Architect has not issued a Certificate for Payment as provided in Paragraph 9.7 of these General Conditions or because the State has not made payment thereon as provided in Paragraph 9.7, then the Contractor may, upon seven additional days written notice to the State and the Architect, terminate the Contract and recover from the State payment for all Work executed and for any proven loss sustained upon any materials, equipment, tools, construction equipment and machinery, including reasonable profit and damages.

#### 14.2 TERMINATION BY THE STATE FOR CAUSE

14.2.1 If the Contractor is adjudged a bankrupt, or if he or she makes a general assignment for the benefit of his/ her creditors, or if a receiver is appointed on account of his/ her insolvency, or if he or she persistently or repeatedly refuses or fails, except in cases for which extension of time is provided, to supply enough properly skilled workmen or proper materials, or if he or she fails to make prompt payment to Subcontractors or for materials or labor, or persistently disregards laws, ordinances, rules, regulations or orders of any public authority having jurisdiction, or otherwise is guilty of a substantial violation of a provision of the Contract Documents, then the State, upon certification by the Architect that sufficient cause exists to justify such action, may, without prejudice to any right or remedy and after giving the Contractor and his surety, if any, seven days written notice, terminate the employment of the Contractor and take possession of the

site and of all materials, equipment, tools, construction equipment and machinery thereon owned by the Contractor and may finish the Work by whatever method he may deem expedient. In such case the Contractor shall not be entitled to receive any further payment until the Work is finished.

**14.2.2** If the costs of finishing the Work, including compensation for the Architect's and State's additional services and expenses made necessary thereby, exceed the unpaid balance of the Contract Sum, the Contractor shall pay the difference to the State.

If the unpaid balance of the Contract Sum exceeds the costs of finishing the Work, including compensation for the Architect's and the State's additional services and expenses made necessary thereby, such excess shall be paid to the Contractor.

The amount owed by Contractor to the State shall be certified by the Architect, upon application, in the manner provided in section 9.4 of these General Conditions, and this obligation for payment shall survive the termination of the Contract.

#### 14.3 TERMINATION BY STATE FOR CONVENIENCE

- 14.3.1 The State may, at any time, terminate this agreement for the State's convenience and without cause.
- 14.3.2 Upon receipt of written notice from the State of such termination for the State's convenience, the Contractor shall:
  - (a) cease operations as directed by the State in the notice;
  - (b) take actions necessary, or that the State may direct, for the protection and preservation of the Work; and
  - (c) except for Work directed to be performed prior to the effective date of termination stated in the notice, terminate all existing subcontracts and purchase orders and enter into no further subcontracts and purchase orders.
- 14.3.3 In case of such termination for the State's convenience, the Contractor shall be entitled to receive payment for Work executed, and costs incurred by reason of such termination, along with reasonable overhead and profit on the Work not executed.

#### 14.4 SUSPENSION BY STATE FOR CONVEVIENCE

- 14.4.1 The State may, without cause, order the Contractor in writing to suspend, delay or interrupt the Work in whole or in part for such period of time as the State may determine.
- 14.4.2 The Contract Sum and Contract Time shall be adjusted for increases in the cost and time caused by suspension, delay or interruption as described in section 14.4.1. Adjustment of the Contract Sum shall include profit. No adjustment shall be made to the extent:
  - (a) that performance is, was or would have been so suspended, delayed or interrupted by another cause for which the Contractor is responsible; or
  - (b) that an equitable adjustment is made or denied under another provision of this Contract.

## STATE OF VERMONT - STATEMENT OF WORK (SOW) Request for Proposal (RFP)

(NOTE: Wording in Italics is instructive. Text that appears in orange letters and yellow highlighting must be updated or deleted. Utilize or delete paragraphs as appropriate to the RFP. Wording in regular script should be included in the RFP as is.)

**Contract Service Category:** 

**Request for Proposal** 

Month DD, YYYY

#### ADMINISTRATIVE INFORMATION

## AGENCY/DEPT RESPONSIBILITY FOR STATEMENT OF WORK (SOW) RFP AND PURCHASE ORDER (PO)

Name the person/s and Agency/Dept.

This SOW RFP is being issued in accordance with the Master Agreement between the Contractor and the State of Vermont, Department of Buildings and General Services, Office of Purchasing and Contracting. After an evaluation of Contractor's response to this SOW RFP, the Contracting Agency may elect to enter into a specific Purchase Order (PO) which will outline SOW requirements.

#### SOW PROPOSAL SUBMISSIONS

All SOW Proposals are due no later than: (Date) and Time

Proposals must be submitted by email, fax, mail or in person to: (State Agency Contact e-mail)

#### STATEMENT OF RIGHTS

The State of Vermont reserves the right to obtain clarification or additional information necessary to properly evaluate a proposal. Contractors may be asked to give a verbal presentation of their proposal after submission. Failure of contractor to respond to a request for additional information or clarification could result in rejection of that contractor's proposal. To secure a project that is deemed to be in the best interest of the State, the State reserves the right to accept or reject any and all bids, in whole or in part, with or without cause, and to waive technicalities in submissions. The State also reserves the right to make purchases outside of the awarded contracts where it is deemed in the best interest of the State.

#### METHOD OF AWARD AND PROCEDURE FOR AWARDING A SOW AGREEMENT

EVALUATION CRITERIA (Verify evaluation criteria below is appropriate for your project)

The responses will be evaluated based on the following:

- Cost
- Prior Experience with this type of work
- Timeline for completion of work to be performed

Contractor selection, or the determination to terminate the SOW RFP without award shall be done in the best interest of the State.

#### SCOPE OF WORK/PROJECT REQUIREMENTS:

Describe required project requirements in detail. Under no circumstance should a SOW be developed or an SOW RFP be released where the deliverables are not quantified or the criteria for acceptance are not defined. Be clear and concise. The deliverables identified here should be directly tied to payment provisions.

#### **CHANGE ORDERS**

We do not anticipate the need to make change orders, however, if it becomes necessary, such work must be authorized by the State in writing before such work can proceed and may require an amendment to the SOW Agreement.

#### **SUBCONTRACTORS**

Identify all proposed subcontractors and their full roles that may be involved completing the Scope of Work. No work shall be subcontracted without knowledge of and approval by the State.

#### REQUIRED PRICE PROPOSAL RESPONSE

All pricing must be fixed cost, inclusive of all expenses and fees if this Statement of Work proposal is for a fixed price agreement. (Remove if Time and Materials agreement)

For Time and Materials the pricing proposal must include estimated effort hours, hourly rate for proposed personnel, projected timeline, including timing expectations for the State functional and technical resources and be submitted as a separate document from the rest of the proposal. (Remove if not Time and Materials)

#### **INVOICING AND PAYMENT**

All work performed by the Contractor must be approved in advance by the State. Once work has been completed, delivered and accepted by the State, invoicing can occur. The State's payment terms are net 30 days.

## EXAMPLE of PRICE PROPOSAL FORM

SOW-RFP PROJECT NAME:
PRICE PROPOSAL: \$
COMPLETION DATE:
ACKNOWLEDGEMENT OF ADDENDUMS (IF APPLICABLE)
1.
2. 3.
COMMENCEMENT OF WORK UNDER A SOW AGREEMENT
Commencement of work as a result of the SOW-RFP process shall be initiated only upon issuance of a fully executed Purchase Order.
COMPANY NAME:
NAME:
SIGNATURE:
DATE:

#### ATTACHMENT E: SECURITY REQUIREMENTS

#### **SECTION 01900**

#### SECURITY REQUIREMENTS – CORRECTIONAL FACILITIES

#### **PART 1 - GENERAL REQUIREMENTS**

#### 1.1 Related Documents

- A. All sections of Division 1.
- B. Examine all drawings and all other Sections of the Specifications for requirements therein affecting the work of this Section. Work shall be coordinated with other trades prior to installation to prevent interference and relocations.

#### 1.2 Security Clearances

- A. The General Contractor is to supply the Department of Corrections with full names, dates of birth, and social security numbers of all employees who will be on site. This information must be submitted a <u>minimum of two (2) weeks prior to access to the site</u>. The Department of Corrections has final word on who will be permitted access to the site. <u>A form for this purpose is attached for your use</u>.
- B. Exclusionary Criteria: Only convictions within the last five (5) years will be considered.
  - a. Drug Conviction
  - b. Domestic Assault
  - c. Aggravated Assault
  - d. Escape from Custody

These need to be new convictions. Furlough violations, technical parole violations or probation violations do not factor in.

#### 1.3 Contractor and Employee Security Precautions

- A. The security aspects of working at the Correctional Facility are critical. The following security precautions are part of the site conditions and are a part of this Contract. All persons coming on the site in any way connected with this Work shall be made aware of them, and it is the (General) Contractor's responsibility to check and enforce them.
  - 1. Parked vehicles must have their keys removed and doors locked. Under no circumstances will firearms, drugs, or alcohol be kept in vehicles.
  - 2. No firearms, bows and arrows, etc., are allowed on any persons or in any vehicles. Any weapons found will be confiscated and will not be returned.
  - 3. No alcohol or drugs are allowed on facility grounds (medication exempted by permission only). If drugs or alcohol are found on persons in vehicles, persons involved will be barred from working at this facility or any other State correctional facility for a period of up to one year. Violators will be prosecuted for illegal substances.
  - 4. No tobacco products or cell phones are allowed within the secured perimeter.

- 5. When necessary for employees to enter secure portions of the Facility, they may be subject to search and inventory of tools.
- 6. Workers will not engage in conversation or be in proximity with inmates, nor do or agree to do any personal requests or favors for any inmate.
- 7. All tools must be either in use, locked up, or removed from the site. Any tools found unattended will became the property of the Department of Corrections.
- 8. If any tools, or portions of tools such as hacksaw blades, are lost or misplaced, the disappearance shall be brought to the attention of the Director of Security at the facility immediately.
- 9. Any trailers left on site must be locked each night.
- 10. Regular working hours are from 7:00 a.m. to 4:30 p.m., Monday thru Friday. However, if agreed upon with Correctional Facility staff, the Contractor's working hours may be changed from those referenced above. A security guard will be available to assist contractor's personnel, equipment and vehicles through the trap.
- 11. If an accident occurs on State property, a written report must be submitted to the Chief of Security.

**END OF SECTION 01900** 



# VERMONT DEPARTMENT OF CORRECTIONS WAIVER FOR CRIMINAL RECORD AND BACKGROUND VERIFICATION

Project:   Facility:   BGS Project Manager:   e-mail:		BGS Project Manager is to	to fill-in this section before giving the form to the contractor
Company Name:  Applicant Name:  Alias/Maiden/Other Names:  Social Security Number:  Place of Birth:    City/Town	Project:		Facility:
Applicant Name:  Alias/Maiden/Other Names:  Social Security Number:  Place of Birth:  Cary/Town  Current Residence:	<b>BGS Project Mana</b>	iger:	e-mail:
Applicant Name:  Alias/Maiden/Other Names:  Social Security Number:  Place of Birth:  Current Residence:  # and Street  # and St			
Alias/Maiden/Other Names:  Social Security Number:  Place of Birth:  City/Town State  Current Residence:  # and Street  # and St	<b>Company Name:</b>		
Social Security Number:  Place of Birth:  City/Town State State  Current Residence:  # and Street State/Country  Previous Residence:  # and Street Wgt: Race/Sex  RELEX SE  I,	<b>Applicant Name:</b>		
Place of Birth:    Current Residence:	Alias/Maiden/Othe	er Names:	
Current Residence:  # and Street  # and Stre	Social Security Nu	mber:	Date of Birth:
Current Residence:    # and Street	Place of Birth:		
Previous Residence:  # and Street  # State/Country  # State/Country  # and Street  # Wgt: Race/Sex  # RLEx VE  # Country  # and Street  # and		City/Town	State
Previous Residence:  # and Street  # and Street    Telephone Number:   Telephone Number:   Wgt:   Race/Sex	Current Residence	:	
Telephone Number:  R. LE. SE  I,		# and Street	C'Cown State/Country
Telephone Number:  RELECTE  I,	Previous Residence	e:	2
RELECTE  I,		# and Street	To State/Country
I,	Telephone Number	r:	Vgt: Wgt: Race/Sex
	Department of C maintained by the of that check will suitability for we have the right to Information Cen 05671-1300.	e Vermont Craninal and I be a 1's available to a prince within a Department of Publisher, Department of Publisher	hereby acknowledge and authorize the Vermont a chesof any criminal record of convictions which may be nation Center and the FBI. I understand that the results the Department of Corrections for use in reviewing my ment of Corrections facility. I further understand that I he criminal record check to the Vermont Criminal blic Safety, 45 State Drive, Waterbury, Vermont
Identity Verified by Construction Company Date			
	Identity Verified by Co	onstruction Company	Date

e-mail completed form to:

Gary Dillon

Department of Corrections NOB 2 South

NOB 2 South 280 State Drive

Waterbury, VT 05671-2000 gary.dillon@vermont.gov

#### **SECTION 01901**

#### **SECURITY REQUIREMENTS - COURTS**

#### **PART 1 - GENERAL REQUIREMENTS**

#### 1.1 Related Documents

- A. All sections of Division 1
- B. Examine all drawings and all other Sections of the Specifications for requirements therein affecting the work of this Section. Work shall be coordinated with other trades prior to installation to prevent interference and relocations.

#### 1.2 Security Clearances

A. The General Contractor is to supply the Court with names, dates of birth, and social security numbers of all employees who will be on site. This information must be submitted one (1) week prior to access to the site. The Court has final word on who will or who will not be allowed on the project premises.

#### 1.3 Contractor and Employee Security Precautions

- A. As the project site is an operational Court, the Contractors must take the following security precautions.
- B. Parked vehicles must have their keys removed and doors locked. Under no circumstances will firearms, drugs, or alcohol be kept in vehicles.
- C. No firearms, illegal drugs, or alcohol will be allowed on premise. Violators will be prosecuted.
- D. When necessary for employees to enter secure portions of the Facility, they may be subject to search and inventory of tools.
- E. Employees are to avoid fraternizing with the public while in the facility.
- F. All tools must be either in use, locked up, or removed from the site. Any tools found unattended will became the property of the Courts.
- G. If any tools, or portions of tools such as hacksaw blades, are lost or misplaced, the disappearance shall be brought to the attention of the Director of Security at the facility immediately.

END OF SECTION 01901

BY SIGNING BELOW I UNDERSTAND THAT A ROUTINE CRIMINAL
RECORD CHECK WILL BE PERFORMED IN ACCORDANCE WITH THE BGS
SECURITY DEPARTMENT REGULATIONS.

THE VERMONT JUDICIAL BRANCH REQUIRES BACKGROUND CLEARANCE FOR WORKING WITHIN COURTHOUSES AND JUDICIAL FACILITIES.

PRINT NAME:		<b>)</b>
AUTHORIZING		
SIGNATURE:		
DATE OF BIRTH:		_
SSN:	_XXX-XX	_(LAST 4 DIGITS)
CONTRACTOR:		
PROJECT:		
BGS PROJECT MGI	R	

PLEASE SENI COMPLETE, FORM TO:

Robert Schell

Security and Safety Program Manager

Office of the Court Administrator

Vermont Supreme Court

109 State Street

Montpelier, VT. 05609-0701

#### **DEPARTMENT OF PUBLIC SAFETY**

Once the contractor has been selected any members they have on site and that includes sub contractors must complete the attached form. The Vermont Criminal Information Center (VCIC) section of Public Safety MUST have these forms 2 weeks prior to the contractor starting work or appearing on site to start work. Any employee or sub-contractor that has not had this completed on them will not be authorized to work there or enter the site/building.

David Tifft
Fleet Administrator
Vermont Dept of Public Safety
394 Hegeman Avenue
Colchester, VT 05446
802-655-0748
David.Tifft@state.vt.us



Vermont Department of Public Safety
Building Access Request/Security Clearance/Signature Form
Please note: Applicant completes yellow sections. Supervisor completes pink section. Leave green section blank.
PERSONAL REQUIRED INFORMATION – Please fill o

	Last Name	First Nam	First Name		MI	the state of the s	and the other parties of the same of the s	DOB (yyyy-mm-dd)	
					Division:		CLocation:	☐ Civilian ☐ Sworn	
	Date of Hire (yyyy-m	m-dd) Other							
	EMPLOYMENT STATUS:  Permanent Part-Time			CARD TYPE: New Last 4 Digits of SS Replacement Cancellation XXX-XX-					
						111:0			
IRED	Nace	Height	ight Weight				olor	Gender: Male Female	
EQU	A faved or photocopy	of this consent for				RF			
R	By my signature below, I am asserting that the information above is accurate.  Furthermore, by my signature below, I consent to the process of a critical history and the vehicle records check by the Vermont Department of Public Safety. I understand that continued employing the or work authorization may be denied as a result of such record checks.  With regard to criminal history information, I understand that I have been to to appeal the soults of the criminal record check to the Vermont Criminal Information Center, Department of Public Safety, 103 St. h Main trans, Waterbury, Vermont, 05671-2101.								
	Applicant's signature (Please use felt pen in signature block to the right.)							Date:	
	All VSP Building						orn person	nel)	
		y (Default or o							
	Additional Acce	35 Request	ist bek	w (Require	es superv	isory appr	ovai).		
UIRE									
	(Supervise								
			K CONS	ENT FORM	- CONFIL	s printed nam DENTIALIT	e) Y STATEM	(Date)	
	Criminal record inform	mation from the \	/ermont C	riminal Infor	mation Cer	nter is being	released to t	he undersigned with	
	to any person. Unaut	thorized dissemir	nation or u	sed solely lo	al records is	se intended s punishable	by law and v by a fine of	up to \$5,000 (20	
the understanding that the information will be used solely for the purpose intended by I to any person. Unauthorized dissemination or use of criminal records is punishable by V.S.A. §2056c(g)).  (Reviewer's signature)  (Reviewer's printed name)  Purpose of Check:    Card Type(s):									
19	(Reviewer's signature)			(Reviewer's printed name				(Date)	
	Purpose of Check:	Card Type(s):		Proc User received	Manager and the second		New ID Card	Number(s):	
		☐ BGS Card		T OBEL LECEIAGO	a caru(s) f		RF <sub>1</sub>		
	Employment Contractor			Cub	DCC	1 1			
	☐ Employment ☐ Contractor ☐ Building Security	Westinghous	e   _	Submitted to			RF <sub>2</sub>		
	☐ Contractor		e   L	Submitted to BGS Activate User returned	d Card on _		RF <sub>2</sub> Old ID Card		
	REQUIRED REQUIRED	Last Name	Employee # Job Title  Date of Hire (yyyy-mm-dd)	Employee # Job Title Depart  Date of Hire (yyyy-mm-dd) Other states/could leave the fight Depart  EMPLOYMENT STATUS: Permanent Part-Time Retired ID Only (no door access)  Race Height Weight  APP  A faxed or photocopy of this consent form serves a By my signature below, I am asserting that the informat Furthermore, by my signature below, I consent to the Vermont Department of Public Safety. I understand that record checks.  With regard to criminal history information, I understand Vermont Criminal Information Center, Department of Public Safety. I was signature (Please use felt pen in signature block to the right.)  All VSP Buildings/All Rea Except Mork Place Only (Default or chart of the right.)  Additional Access Request Notes and Safety Safety. I st Ben Safety	Last Name  Employee # Job Title  Date of Hire (yyyy-mm-dd)  Date of Hire (yyyy-mm-dd)  CARD TYP  Permanent Part-Time Replacen  Retired ID Only (no door access)  Race Height Weight Eye  APPLICANT'S  A faxed or photocopy of this consent form serves as an original of Vermont Department of Public Safety. I understand that continued emprecord checks.  With regard to criminal history information, I understand that I have vermont Criminal Information Center, Department of Public Safety. 1  Applicant's signature Please use felt pen in signature (Please use felt pen in signature)  CRIMINAL RECORD CHECK CONSENT FORM  Criminal record information from the Vermont Criminal Information any person. Unauthorized dissemination or use of criminal (Please use of criminal person). Unauthorized dissemination or use of criminal person.	Last Name	Last Name   First Name   MI   Maiden/Ali	Employee # Job Title   Department: Division: Work Location:  Date of Hire (yyyy-mm-dd)   Other states/countries in which you have lived:	

### **ATTACHMENT F**

